# Popularity Bias in Recommendation: A Multi-stakeholder Perspective

by

## Himan Abdollahpouri

B.S., Bu Ali Sina University, 2009

M.S., Iran University of Science and Technology, 2012

A thesis submitted to the Faculty of the Graduate School of the University of Colorado in partial fulfillment of the requirements for the degree of Doctor of Philosophy Department of Information Science

2020

Committee Members:

Prof. Robin Burke

Prof. Christine Lv

Prof. Edward Malthouse

Prof. Bamshad Mobasher

Prof. Stephen Voida

Abdollahpouri, Himan (Ph.D., Information Science)

Popularity Bias in Recommendation:

A Multi-stakeholder Perspective

Thesis directed by Prof. Prof. Robin Burke

Recommendation and ranking systems are ubiquitous. They are used in a variety of different domains where they can influence what news people read, what products they buy, who they date and marry, and what cultural artifacts they consume. Traditionally, especially in academic research in recommender systems, the focus has been solely on the satisfaction of the end-user. While user satisfaction has, indeed, been associated with the success of the business, it is not the only factor. In many recommendation domains, there are other stakeholders whose needs and interests should be taken into account in the recommendation generation and evaluation. In this dissertation I describe the notion of multi-stakeholder recommendation and discuss its main building blocks, challenges, and connections to prior research. In particular, I study one of the most important challenges in recommendation research, popularity bias, from a multi-stakeholder perspective since, as I show later in this dissertation, it impacts different stakeholders in a recommender system. Popularity bias is a well-known phenomenon in recommender systems where popular items are recommended even more frequently than their popularity would warrant, amplifying long-tail effects already present in many recommendation domains. Prior research has examined various approaches for mitigating popularity bias and enhancing the recommendation of long-tail items overall. The effectiveness of these approaches, however, has not been assessed in multi-stakeholder environments. In this dissertation, I study the impact of popularity bias in recommender systems from a multi-stakeholder perspective. In particular, I investigate how this bias impacts users and suppliers of the recommendations. That is, how this bias causes the recommendations to deviate from the users' preferences towards popular items and how it leads to unfair exposure for different supplier groups. Finally, I propose several algorithms each approaching the popularity bias mitigation from a different angle and compare their performances using several metrics with some other state-of-the-art approaches in the literature. I show that, often, the standard evaluation measures of popularity bias mitigation in the literature do not reflect the real picture of an algorithm's performance when it is evaluated from a multi-stakeholder point of view.

## Dedication

This dissertation is a small gift to my parents. To my hard working father who started working when he was a teenager after his father passed away. He never went to school so he could support his mom and siblings. A construction worker who spent his entire life supporting his family. Every evening when he was coming back from work, the sound of his motorcycle was the most pleasant thing we could hear coming from the distance and slowly approaching us. He has been always my hero and the symbol of honesty, authenticity, and integrity. Also to my mother: the selfless woman who never asked for anything but always was the provider to her husband and children. She can't read and use smartphones so she used to go to the neighbors and relatives asking them if she could call her son in the U.S. I still remember the excitement and brightness in their eyes when I first told them I was accepted as a PhD student. It has been five years since the last time I saw them and I hope I can see them soon. Every time I talk to them over the phone they keep saying "We are sorry that we can't financially support you living in a foreign country and all we can offer you is our prayers" little do they know this is the best thing I could ever ask for. Thanks for everything you have giving me till today.

## Acknowledgements

First and foremost, I would like to express my sincere gratitude to my supervisor Prof. Robin Burke for all the effort and patience he has put in mentoring my PhD and guiding my research. I thoroughly enjoyed our brainstorming sessions and learned a lot during this five years both personally and professionally.

Also, special thanks to my PhD committee members Prof. Christine Lv, Prof. Edward Malthouse, Prof. Bamshad Mobasher, and Prof. Stephen Voida for their great and invaluable comments both on my proposal and dissertation defense. I am also thankful to the RecSys community for supporting me and my research. Finally, I would also like to thank the National Science Foundation for supporting my research during my PhD.

# Contents

# Chapter

1	Intro	oductio	on	1
	1.1	Resea	rch Goals and Contributions	1
	1.2	Struct	cure of the Thesis	3
	1.3	Defini	itions and Notations	4
2	Bacl	kgroun	d	7
	2.1	Recon	nmender Systems	7
		2.1.1	Types of Recommendation Algorithms	11
	2.2	Evalu	ation Methodologies	17
		2.2.1	Offline Evaluation	17
		2.2.2	Online Evaluation	18
		2.2.3	User Studies	19
	2.3	Evalu	ation Metrics	19
		2.3.1	Rating Prediction	20
		2.3.2	Ranking	21
		2.3.3	Non-user Evaluation	23
	2.4	Challe	enges in Recommendation	24
		2.4.1	Diversity, Novelty and Long-tail	26

3	Mul	ti-stake	cholder Recommendation	27
	3.1	Introd	luction	27
	3.2	Recon	nmendation Stakeholders and Multi-stakeholder Paradigm	29
		3.2.1	Multi-stakeholder Evaluation	32
		3.2.2	Multi-stakeholder Algorithms	33
		3.2.3	Multi-stakeholder Design	34
		3.2.4	The Landscape of Multi-stakeholder Recommendation	34
	3.3	Relate	ed Research	37
		3.3.1	Economic foundations	37
		3.3.2	Multi-objective recommendation	38
		3.3.3	Personalization for matching	39
	3.4	Evalua	ation	40
		3.4.1	Simulation	40
		3.4.2	Models of utility	41
		3.4.3	Off-line experiment design	42
		3.4.4	User studies	42
		3.4.5	Metrics	43
4	Data	a and M	ſethodology	46
	4.1	Datase	ets	46
	4.2	Recon	nmendation Algorithms	47
	4.3	Stakeł	nolder Grouping	48
	4.4	Evalua	ation Metrics	53
5	Pop	ularity	Bias in Recommendation	59
	5.1	Bias ir	n Data	61
	5.2	Algor	ithmic Bias	62
	5.3	Multi-	-stakeholder Impact of Popularity Bias	63

vii

		5.3.1	Impact on Item Groups	66
		5.3.2	Impact on User Groups	67
		5.3.3	Impact on Supplier Groups	68
6	Tack	kling Po	opularity Bias	75
	6.1	Mode	l-based solutions	76
		6.1.1	Proposed Model-based Technique Using Regularization	78
	6.2	Re-rai	nking Approach	85
		6.2.1	Re-ranking Technique 1: Fair Long-tail Promotion:	88
		6.2.2	Re-ranking Technique 2: Calibrated Popularity	94
7	Con	clusion	and future work	117
7	Con 7.1	clusion Summ	and future work	117 117
7	Con 7.1	Sumn 7.1.1	and future work hary of Contributions	117 117 117
7	Con 7.1	Summ 7.1.1 7.1.2	and future work hary of Contributions	117 117 117 117 118
7	Con 7.1	Summ 7.1.1 7.1.2 7.1.3	and future work hary of Contributions	117 117 117 118 119
7	Con 7.1 7.2	Summ 7.1.1 7.1.2 7.1.3 Future	and future work hary of Contributions	117 117 117 118 119 120
7	Con 7.1 7.2	Summ 7.1.1 7.1.2 7.1.3 Future 7.2.1	and future work hary of Contributions	117 117 117 118 119 120 120
7	Con 7.1 7.2	Summ 7.1.1 7.1.2 7.1.3 Future 7.2.1 7.2.2	and future work hary of Contributions	117 117 117 118 119 120 120 121
7	Con 7.1 7.2	Summ 7.1.1 7.1.2 7.1.3 Future 7.2.1 7.2.2 7.2.3	and future work hary of Contributions	117 117 117 118 119 120 120 121 121

# Bibliography

viii

# Tables

# Table

1.1	List of Publications that are the result of this dissertation and their corresponding
	chapter
4.1	Number of users $ U $ , number of items $ I $ , number of ratings $ R $ , and number of
	suppliers $ \mathcal{P} $ in two datasets MovieLens and Last.fm $\ldots \ldots \ldots \ldots \ldots \ldots 47$
4.2	The summary of the evaluation metrics used in this dissertation, their abbreviations
	and their purpose
6.1	Results of different algorithms on two datasets. The setting that yields the same
	precision for all three reranking methods XQ, DM and CP is chosen and the results
	is reported. Bold values are statistically significant compared to the second best
	value in each column with significance level of 0.05 ( $\alpha$ =0.05)
6.2	Results of different algorithms on two datasets. The setting that yields the same
	precision for all three reranking methods XQ, DM and CP is chosen and the results
	is reported. Bold values are statistically significant compared to the second best
	value in each column with significance level of 0.05 ( $\alpha$ =0.05)

# Figures

# Figure

2.1	A general schema for recommender systems	8
2.2	Contingency Table	21
3.1	Traditional schema for recommendation versus multi-stakeholder schema	31
4.1	Profile size of users in different user groups. Numbers within the parentheses show	
	the average profile size of the users in each group	50
4.2	Number of songs / movies for artists / directors in each supplier group. The first	
	number in the parentheses shows the size of each group and the second number	
	shows the average number of songs / movies for the suppliers in each group	51
4.3	List of two Recommendations given to three users. Dark circles are popular items	
	and light ones are long-tail items. There are ten items overall in the catalog 6 of	
	which are long-tail and the other 4 are popular. $Agg-Div=\frac{4}{10}=0.4$ , $APL=\frac{1}{2}=0.5$	
	and $LC = \frac{1}{6} = 0.166.$	54
5.1	The Long-tail Distribution of Rating data	61
5.2	Item popularity versus recommendation popularity (MovieLens)	64
5.3	Item popularity versus recommendation popularity (Last.fm)	65
5.4	Popularity bias amplification (Item Groups)	69
5.5	Users' Propensity towards item popularity	70
5.6	User Groups' Propensity towards item popularity	71

5.7	Users' centric view of popularity bias	72
5.8	Users' centric view of popularity bias (Groups)	73
5.9	Suppliers' centric view of popularity bias	74
6.1	The Schema for the model-based popularity mitigation	76
6.2	Precision, NDCG, Gini, APL, Aggregate Diversity and Long-tail coverage, Equality	
	of attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popular-	
	ity Deviation (UPD) and Supplier Popularity Deviation (SPD) of our model-based	
	technique on MovieLens data	83
6.3	Precision, NDCG, Gini, APL, Aggregate Diversity and Long-tail coverage, Equality	
	of attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popular-	
	ity Deviation (UPD) and Supplier Popularity Deviation (SPD) of our model-based	
	technique on Last.fm data	84
6.4	The schema for the re-ranking popularity bias mitigation	86
6.5	Precision, NDCG, Gini, APL, Aggregate Diversity and Long-tail coverage, Equality	
	of attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popular-	
	ity Deviation (UPD) and Supplier Popularity Deviation (SPD) of our fairness-aware	
	re-ranking on MovieLens data	92
6.6	Precision, NDCG, Gini, APL, Aggregate Diversity and Long-tail coverage, Equality	
	of attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popular-	
	ity Deviation (UPD) and Supplier Popularity Deviation (SPD) of our fairness-aware	
	re-ranking on Last.fm data	93
6.7	Popularity calibration example	94
6.8	An illustration of the statistical distance between the distribution of item popularity	
	in the user's profile and in the recommendations.	96

6.9	Precision, NDCG, Gini, APL, Aggregate Diversity and Long-tail coverage, Equality
	of attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popu-
	larity Deviation (UPD) and Supplier Popularity Deviation (SPD) of our Popularity
	Calibration Re-ranking approach on MovieLens data
6.10	Precision, NDCG, Gini, APL, Aggregate Diversity and Long-tail coverage, Equality
	of attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popu-
	larity Deviation (UPD) and Supplier Popularity Deviation (SPD) of our Popularity
	Calibration Re-ranking approach on Last.fm data
6.11	Precision, Gini, Gini, APL, Aggregate Diversity and Long-tail coverage, Equality of
	attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popularity
	Deviation (UPD) and Supplier Popularity Deviation (SPD) for all three popularity
	mitigating approaches on MovieLens data
6.12	Precision, Gini, Gini, APL, Aggregate Diversity and Long-tail coverage, Equality of
	attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popularity
	Deviation (UPD) and Supplier Popularity Deviation (SPD) for all three popularity
	mitigating approaches on Last.fm data
6.13	Exposure rate vs popularity across different re-ranking algorithms (MovieLens) 106
6.14	Cumulative frequency of the ratings for different items at a given rank and their
	frequency of appearance in the recommendations
6.15	Popularity deviation metrics for different stakeholder groups
6.16	The ratio of different item groups in the recommendations of different users using
	the popularity bias mitigation techniques
6.17	The ratio of different item groups in the recommendations of different user groups
	using the popularity bias mitigation techniques
6.18	Cumulative Frequency of recommendations from different supplier groups versus
	their popularity in rating data

# Chapter 1

## Introduction

# 1.1 Research Goals and Contributions

This research work focuses on popularity bias problem in recommender systems. Generally speaking, popularity bias refers to the problem where the recommendation algorithm favors a few popular items while not giving deserved attention to the majority of other items. In particular, the goal of this thesis is to evaluate the impact of popularity bias in recommendation on different stakeholders on a given recommender system platform such as users and item suppliers. In addition, several algorithms for mitigating this bias are proposed. With that being said, the following are the research goals and contributions of this thesis:

• **RG-1**: Proposing a comprehensive paradigm called multi-stakeholder recommendation for evaluation and development of recommender systems. Especially in academia, recommender systems are typically evaluated based on the satisfaction of one stakeholder: the end user.

**Contributions:** In this dissertation, I introduce the notion of multi-stakeholder recommendation where the needs and interests of multiple stakeholders are taken into account in the recommendation generation and evaluation, in addition to that of users. In many real-world applications of recommender systems users are not the only stakeholder so the multi-stakeholder paradigm is crucial in such systems. Take a music streaming service such as Spotify as an example. On the one hand, there are listeners who use the service

to find and listen to the music they would enjoy. On the other hand, the creators of those songs (i.e. the artists) also benefit from the platform by their songs being played for / by different users and hence gaining utility. The popularity bias in recommendation can negatively impact both sides of the platform.

• **RG-2**: Investigating the impact of popularity bias in recommendation on different users. Popularity bias plays an important role in the unfairness of the recommendations towards different stakeholders. There has been research on the popularity bias and long-tail recommendation which focuses on the consequences of such bias from the system's perspective such as item coverage, aggregate diversity etc or how this bias can negatively impact the accuracy of the recommendations.

**Contributions:** In this thesis, I investigate how the popularity bias problem impacts different user groups with a varying degree of interest in item popularity. For instance, how users with niche tastes are impacted by this bias versus those with a more inclination towards mainstream products.

• **RG-3**: Investigating the impact of popularity bias in recommendation on different item providers (aka suppliers). As mentioned earlier, suppliers of the items on a recommendation platform are also stakeholders whose needs and preferences may need to be addressed. Typically, in many recommendation domains, the platform may not necessarily own any of the recommended items itself but rather acts as a bridge that connects the suppliers of the items to the users. For instance, in music recommendation, each song or album is produced by an artist; on Airbnb, each listing is provided by a host; on eBay, each product is provided by a seller. Therefore the dynamics of how recommendations impact these item suppliers is important.

**Contributions:** In this thesis, I investigate how popularity bias in recommendation impacts different item suppliers with a varying degree of popularity. For instance, I investigate whether this bias causes over-promotion of items from certain item suppliers while

under-recommending items from some others.

• **RG-4**: Proposing algorithms to mitigate popularity bias in recommendation and evaluating their performance from a multi-stakeholder perspective.

**Contributions:** Most of the existing work on long-tail in recommendation concentrates on either the accuracy of the recommendation in the presence of popularity bias and how to address that or they are mainly focusing on the holistic system-level picture of this bias and its impact on overall recommendation coverage. In this thesis, I propose several algorithms for controlling popularity bias and see how each of these perform from the perspective of different stakeholders. By "control" I mean reducing the over-concentration of the recommendations on few popular items and to give more chance to other items to be seen by the users. In particular, the algorithm based on calibrating recommendations to the users' historical interaction with respect to item popularity shows benefits both to the users and the suppliers of the recommendations.

# **1.2** Structure of the Thesis

This dissertation is structured as follows:

- Chapter 2 gives a brief summary of the field of recommender systems. In particular, different types of recommendation methods are explained and the general evaluation methodologies for evaluating the recommendations are described. Then, I move on to give a short background of several non-accuracy concepts in recommender systems such as diversity, novelty, and long-tail recommendation.
- Chapter 3 gives a thorough description of the characteristics of multi-stakeholder recommendation and particular considerations involved in creating and evaluating multistakeholder recommender systems. The connection between prior work in non-accuracy measures and the proposed multi-stakeholder paradigm is also studied.

- Chapter 4 describes the methodology and data used for experiments in this dissertation. Details regarding the datasets, the methodology for running the experiments and also the evaluation metrics that are used in other chapters of this dissertation are all fully described.
- Chapter 5 describes the popularity bias problem and how it can impact different stakeholders in recommender systems. Two sources of bias including data and algorithms are explained using two datasets.
- Chapter 6 gives the details of the proposed algorithms in this dissertation to mitigate the popularity bias problem. All algorithms are evaluated using different evaluation metrics described in Chapter 4 and two state-of-the-art baselines for popularity bias mitigation are also included for comparison.
- Chapter 7 gives a short summary of the findings in this dissertation and proposes several interesting open questions and possible future work.

Table 1.1 contains the list of main publications that are the result of these dissertations and are appeared in peer-reviewed journals, conferences, and workshops.

# **1.3** Definitions and Notations

I summarise here the main definitions and notations that I use throughout this dissertation. If needed, additional notation will be introduced in chapters where it applies.

A recommendation problem is defined as suggesting items from a collection of items I to a group of users U. I also show an individual user by u and an individual item by i. I show the interaction data as a matrix R where it contains the ratings that users have given to particular items. In certain cases, instead of ratings, we have the information about whether or not a user has interacted with a certain item (liked, clicked, watched, played etc). The rating a user u has given to an item i is denoted as  $r_{ui}$  and the predicted rating for that user and item pair returned **Table 1.1:** List of Publications that are the result of this dissertation and their corresponding chapter.

Publiction	Chapter
[1] Abdollahpouri, Himan, Gediminas Adomavicius, Robin Burke,	
Ido Guy, Dietmar Jannach, Toshihiro Kamishima, Jan	
Krasnodebski, and Luiz Pizzato."Multistakeholder recommendation:	3
Survey and research directions." User Modeling	
and User-Adapted Interaction 30, no. 1 (2020): 127-158.	
[2] Abdollahpouri, Himan, and Robin Burke. "Multi-stakeholder	
recommendation and its connection to multi-sided fairness.	2
" Workshop on Recommendation in Multi-stakeholder Environments	5
RMSE in Conjunction with RecSys (2019).	
[3] Burke, Robin D., Himan Abdollahpouri, Bamshad Mobasher,	
and Trinadh Gupta. "Towards Multi-Stakeholder Utility	2
Evaluation of Recommender Systems." In UMAP	5
(Extended Proceedings). 2016.	
[4] Abdollahpouri, Himan, Robin Burke, and Bamshad	
Mobasher. "Recommender systems as multistakeholder environments.	2
" In Proceedings of the 25th Conference on User Modeling, Adaptation	5
and Personalization, pp. 347-348. 2017.	
[5] Burke, Robin, and Himan Abdollahpouri. "Educational	
recommendation with multiple stakeholders." In 2016	2
IEEE/WIC/ACM International Conference	5
on Web Intelligence Workshops (WIW), pp. 62-63. IEEE, 2016.	
[6] Burke, Robin, and Himan Abdollahpouri. "Patterns of	
Multistakeholder Recommendation." Workshop on	2
Value-aware and Multi-stakeholder Recommendation	5
VAMS in Conjunction with RecSys (2017).	
[7] Abdollahpouri, Himan, Masoud Mansoury, Robin Burke,	
and Bamshad Mobasher. "The unfairness of popularity	5
bias in recommendation." arXiv preprint arXiv:1907.13286 (2019).	
[8] Abdollahpouri, Himan, and Masoud Mansoury.	
"Multi-sided Exposure Bias in Recommendation."	5
Workshop on Industrial Recommendation Systems (IRS),	5
in Conjunction with KDD 2020.	
[9] Abdollahpouri, Himan, Robin Burke, and Bamshad	
Mobasher. "Controlling popularity bias in learning-to-rank	6
recommendation." In Proceedings of the Eleventh ACM Conference	0
on Recommender Systems, pp. 42-46. 2017.	
[10] Abdollahpouri, Himan, Robin Burke, and Bamshad	
Mobasher. "Managing popularity bias in recommender	6
systems with personalized re-ranking."	0
In The Thirty-Second International Flairs Conference. 2019.	
[11] Abdollahpouri, Himan, Masoud Mansoury, Robin Burke,	
and Bamshad Mobasher. "Addressing the Multistakeholder Impact	6
of Popularity Bias in Recommendation	
Through Calibration." arXiv preprint arXiv:2007.12230 (2020).	

by the recommendation algorithm is denoted by  $\hat{r}_{uj}$ . For any user u, we typically recommend a list of items denoted by  $\ell_u$ . The user for whom we generate the recommendations is often referred to as the *target user*. The list of all the recommendation lists given to all users is denoted by L. Additionally I show the profile (i.e. the interaction history) of user u by  $\rho_u$  which is  $\{i \in I, (u, i) \in R\}$ . Each item i is provided by a supplier  $s_i$ . S denotes a set of supplier groups where each supplier group j is shown by  $S_j$ . The set of all individual suppliers in a given dataset is shown by  $\mathcal{P}$ .

# Chapter 2

# Background

#### 2.1 Recommender Systems

It is always good to have a choice. However with the massive amount of information, data, products that are created everyday, it is constantly becoming more difficult to handle this overwhelming number of choices and make a final decision. Information retrieval systems such as search engines can help a user seeking information by filtering from a large collection of documents (or items or products), those that are most relevant to the user's needs. Users express their needs in terms the type of information they are looking for as a query, which for example could be a list of keywords, or some other explicit description of the requirement. In recommender systems, however, often there is no explicit query. Instead, the system uses its knowledge about the user to reduce the cognitive load of choice by predicting which items are most relevant to her while still leaving the final decision making for the user.

Recommender systems are intelligent tools and algorithms that help users find relevant items according to their preferences [134]. Especially in areas where the number of possible choices is overwhelming, recommender systems been successfully used in finding the ones that are most relevant to the user. By offering personalized recommendations, these systems lower the decision effort for the users, making the item selection process much easier. Recommender systems can be seen as personal shoppers who preselect the best items based on the knowledge they have about the customer, such that a she has to only decide which to pick from a small set of items.



Figure 2.1: A general schema for recommender systems

The general schema for a given recommender system is shown in Figure 2.1. The algorithm uses the information about the users and items stored in the rating data to give a list of recommendations  $\ell$  of size n to a user. We can think of this schema as an abstract view that often can be modified to serve different purposes as we will see in Chapter 6.

Recommender systems have become popular in the last decades, however the recommendation process is much older than that. Even in our daily life, people often make decisions based on the suggestions made by others. For example, we may get recommendations from our friends about movies, music, restaurants or even a job. We trust these recommendations if we know that the person who gave it has some knowledge of the topic, has similar taste to ours or knows our taste, or that the past shows we can trust that person's suggestions; this is known as "word of mouth" [24]. What recommender systems do is to mimic the above behaviour on a larger scale, by utilizing viewpoints of many users that do not have to know each other and have similar tastes. This is known as the collaborative filtering.

Recommender systems are not solely tools that were developed to help users in the item selection process. In fact, there are a number of other reasons why different service providers adapt recommender systems in their services [134]. From the business' perspective, probably the most important is the ability to increase sales through recommendations [146]. Another example of business-oriented goals is to diversify sales, such that they do not center around a small set of very popular items. It might be hard to achieve this goal without recommender systems as there is a risk and cost associated with every item promotion. Promoting less popular items is more risky, however recommender systems can help to recommend less popular items to the right users. It is shown that a well-designed recommender system can also improve users' satisfaction with the service [117], and their loyalty to the service [146]. A better understanding of what the users really like can be utilized not only by a recommendation engine, but for other operational or marketing purposes.

Not only are there different reasons we deploy recommender systems, but also different scenarios which may depend on the domain that we are working in. Historically, the most commonly considered scenario is the recommendation task of movies or music (based on datasets such as MovieLens [72], Netflix [32], or Yahoo! Music [56]), since these are the type of data that are publicly available. However recommender systems can be used in more complex settings and in different domains such as e-commerce [?], job recommendation [122], and even in online dating platforms [?].

In recommender systems, the users interaction with the system is often referred to as a transaction. For example, a transaction could be a user listening to a song, or rating a recently booked accommodation. Commonly, with a transaction we may have an associated feedback, in this case the act of playing a song, or rating the accommodation, represent, respectively, implicit and explicit feedback [85]. The main difference between these two is that for the explicit feedback, user consciously expresses her opinion or preference, where in case of the implicit feedback the preference is inferred from other actions recorded by the system [120] — if a user plays a song and skips it after a few seconds, this could represent a dislike rather than a like. In the case of the implicit feedback, it is the recommender system's job to figure out the strength of the signals and what each action could indicate for different users. On the other hand, in the case of explicit feedback, the type of the feedback may vary. Feedback on contents like movies and books have

been commonly expressed through numerical ratings (like 1-5 stars, or 1-10 points), written reviews, or simple like/dislike indicators. In modern systems, often multiple sources of feedback are available all of which can impact the design of a recommender system.

Another factor affecting the design is the consumption pattern of items, how often we interact with an item, and how long the item is 'fresh'. Depending on the domain, a transaction may reoccur, e.g. an item is repurchased by the user [48], and there are other scenarios where interactions are rather unique, or the reoccurrence is unlikely to happen. For example, in a music recommendation system it is natural that the same songs and artists are recommended to users multiple times. However, recommending the same movie to watch a second time may make sense to some but not to all. Similarly, while it may be desirable to recommend coffee beans every few weeks, recommending another coffee machine does not sound reasonable. In the news domain, recommending the same news article to a user is rather useless, but this is not the only problem. News are very dynamic and often the most recent news are more important to the users than old news. The lifetime of news stories is typically very short. All these factors have to be taken into account when designing systems like a news recommendation system [39,65,106].

Lastly, an important aspect that has to be considered is the type of recommendations that the system should produce, and the use cases in which these are used. In some cases, we might be interested in accurately predicting how the user will rate an item when presented to the user. On the other hand, we might want to generate a list of best items to the user without focusing on the accuracy of the predicted ratings per se. These represent the two main types of problems that have been distinguished in the literature [155]: the rating prediction and the ranking problem. Historically, the rating prediction was mostly considered and it was the main evaluation metric used in Netflix Challenge in 2006 [32]. However, more recently the ranking task has been found to be more relevant to the users and business since, in the end, it is how the users would like the recommendations in a list that matters not how close our predicted ratings are to the original ratings given by the user. The choice of the recommendation task significantly affects the evaluation methodology of recommendations. For all these different reasons and scenarios, a number of different algorithms have been proposed that generate recommendations to the users, which I discuss next.

## 2.1.1 Types of Recommendation Algorithms

Recommender system algorithms can be classified according to a number of criteria. Burke [38] distinguished six different classes of recommender systems: content-based, collaborative filtering, demographic, utility-based, knowledge-based, and hybrid recommender systems. I focus my review on the three, most common of them: collaborative filtering, content-based, and hybrid.

#### Content-based Recommender Systems

In the content-based recommender systems [108, 125], the main task of the system is to analyze items with which users interacted and build users' interests profiles which shows the overall taste of that user in terms of different types of items. This is typically done based on item features that are used to describe those items. In the recommendation process, the system matches the features of the user's profile against the features of items to find the best items for the user. For example, if a user has shown an interest in drama movies, this will be indicated in his or her interests profile, and the system will recommend movies that are also related to drama movies.

In Lops et al. [108], a number of advantages and disadvantages of content-based recommendation have been identified, especially if compared with the collaborative filtering approach (which I will discuss next). Content-based systems depend solely on the item features and items that the current user interacted with—we do not need any information about other users. In cases of new users without an established interest profiles, users may not receive useful recommendations. This does not happen when a new item is introduced, as long as the item is properly described in terms of its associated features, and there are other items sharing those features. The process of describing items requires domain knowledge, and may be tedious and costly. However, there are real-life examples where that has been done (like the Music Genome Project <sup>1</sup>) and is leveraged for generating the recommendations. Assuming the set of features associated with

<sup>&</sup>lt;sup>1</sup> https://www.pandora.com/about/mgp

different items are well-defined, it is easier to explain recommendations to the users, which positively impacts the users' experience with the system. On the other hand, by design, this approach favours items that are highly similar to those that the user has already seen, making the recommendations of low novelty (and diversity as well), or of little surprise and engagement to the user, which we consider as an undesirable situation we would like to avoid.

#### Collaborative Filtering Recommender Systems

Collaborative filtering methods, by contrast, make recommendations to the current user based on the similarity of rating behavior between the user and other users, on the assumption that if two users have similar taste, they may be interested in what the other user likes [68]. The similarity in taste is based on the similarity in users' feedback on different items, e.g. how similarly the users have rated different items. While, depending on the type of interaction data (explicit vs implicit) different approaches may be needed, collaborative filtering methods are often known to be domain-independent, which is their advantage compared with content-based methods where they have high dependency on the contents of the items. Also, in many cases collaborative filtering approaches have been more effective.

Most of the collaborative filtering approaches fall into two main categories: model-based and memory-based. Model-based recommender systems are similar to parametric models [130] in machine learning. In a model-based algorithm, a recommendation model, depending on a number of parameters is proposed. These parameters are then learned using the user-item interactions in a given dataset. Thereafter, recommendations can be made, without further recourse to the data. However, unlike typical classification methods in machine learning, the model-based recommendation approaches are personalized meaning the result for each prediction varies depending on the user for whom we make prediction. In a memory-based algorithm, unlike model-based techniques, the dataset is explicitly queried for each new recommendation.

Memory-based methods (also called neighborhood-based) have been originally tasked with the rating prediction for items that are unseen to users. This is typically done in two ways, known as user-based [75,96,119] or item-based recommendations [52,144]. User-based methods predict the interest of the current user for a given item using the interaction of other users, called neighbors, who also were interested in that item. Typically we do not consider all users interested in the item, but the *k* most similar neighbors to the current user. The similarity of different users is calculated using some metrics such as cosine similarity, adjusted cosine similarity, or Pearson correlation. We can then, with the use of the similarity function, calculate the score reflecting the predicted relevance of an unseen item for a given user. In the simplest form, the score can be expressed as a sum over feedback of the *k* closest neighbors to the target user on the target item, weighted by the similarity of those neighbors to the target user:

$$\hat{r}_{ui} = Z \sum_{v \in N^k(u)} sim(u, v) r_{vi}$$
(2.1)

where *u* is the target user, sim(u, v) is the similarity between users *u* and *v*, and  $N^k(u)$  denotes the neighborhood of user *u* of size *k*, and *Z* is a normalization constant. This normalization constant can be used to customize the equation to fit different recommendation tasks. Originally, constant  $Z = \frac{1}{\sum_{v \in N^k(u)} |sim(u,v)|}$  has been used for the rating prediction task [10] — it is required to scale the rating prediction to the correct range of values, same as the original ratings — however it has been pointed that for the ranking task the constant of *Z* = 1 is generally more effective [18,45,51].

Several variations of neighborhood-based methods have been considered. Here we presented a neighborhood of a fixed size but another option is to select neighbors by thresholding the similarity value [31]. In other words, considering a user to be a neighbor for another user only if the similarity between them is above a certain threshold. An interesting approach has been proposed by Said et al. [140], where instead of taking the *k* closest neighbors, the *k* furthest neighbors are chosen instead. This is done based on the assumption that items disliked by users that are dissimilar to the target user may be of interest to that user. This approach also improves diversity of the recommendations. Some probabilistic versions of the method have been also proposed. Adamopoulos and Tuzhilin [8] proposed a probabilistic neighborhood selection in the context of over-specialisation and concentration bias, where Cañamares and Castells [45] provided a formal probabilistic formulation and analysed the problem of popularity bias in recommendations. In the item-based methods [145], instead of similarity between users, the similarity between items is used to generate the recommendations. The basic intuition is that items similar to those that the user already liked, are good choices to be recommended to the user. In the simplest form, the predicted score can be calculated as a sum over item similarities between the target item and previously interacted items weighted by user's preference, given the target item belongs to the k closes neighbors of each item in the user's profile:

$$\hat{r}_{ui} = Z \sum_{j \in \rho_u} \mathbb{1}_{N^k(j)}(i) sim(i,j) r_{uj}$$
(2.2)

where u is the target and the indicator function  $\mathbb{1}_{N^k(j)}(i)$  filters out items if they do not belong to the neighborhood  $N^k(j)$ , sim(i, j) is the similarity between items i and j,  $\rho_u$  is the collection of items that user u interacted with (i.e. the profile of user u),  $r_{uj}$  denotes the preference of user u for item j. Similarly as with the user-based methods, a number of extensions have been proposed many of them were actually proposed for both techniques as they are similar in principle.

One of the advantages of the neighborhood-based methods [119] is that they happen to be simple, requiring little parameterization. They are also intuitive in the way they work which makes it easier to justify the recommendations to users. From the computational point of view, since they require no training, they can efficiently generate recommendations. They may require pre-computation of the neighborhoods, however this is generally much cheaper than training a model. On the other hand, neighborhood-based methods are known to suffer from limited item coverage. In other words, they often concentrate on few popular items. They are also sensitive to sparseness of data (datasets generally contain only a fraction of all possible user-item interactions) and may not perform well on sparse datasets.

In the model-based methods, generally there are two phases that can be distinguished: a learning phase and a prediction phase. In the learning phase a model is learned from the interaction data, and in the prediction phase the model is utilized to generate recommendations. This approach is similar to many machine learning methods, such as linear models, (deep) neural networks, clustering, latent factor models and others. In the field of recommender systems, latent factor models have attracted the most attention among model-based methods, especially to perform dimensionality reduction of a ratings matrix. These models aim to uncover latent features that would explain user preferences hidden in observed ratings. Examples include Probabilistic Latent Semantic Analysis (PLSA) [76], Latent Dirichlet Allocation [34], or matrix factorization [99].

In the context of collaborative filtering, the most commonly used model-based method is matrix factorization. Its goal is to learn two low-rank matrices,  $P \in R|U| \times k$  and  $Q \in R|I| \times k$ , which represent, respectively, all users and items in k-dimensional latent space. Typically, k is much smaller than the number of users or items, and it is a model parameter that needs to be tuned. P and Q matrices are learned through the optimization of an objective function L(P,Q). There are different optimization techniques that can be used, such as alternating least squares [126,161], stochastic gradient descent [78,131,151], or maximum margin matrix factorization [132,170]. After learning P and Q, recommendations can be made based on the score

$$\hat{r}_{ui} = p_u^T q_i \tag{2.3}$$

where  $p_u$  is the row vector of P for the user u, and  $q_i$  is the row vector of Q corresponding to the item i. T indicates the transpose of a given matrix.

#### Hybrid Recommender Systems

In the hybrid methods [10, 37], we combine two or more recommendation algorithms to generate recommendations that overcome limitations of the individual methods. This is similar to fusion strategies in information retrieval which are usually adopted to improve performance of individual retrieval methods [171], or in machine learning where ensemble methods have been used to combine several machine learning models [54] for improving the performance. In Burke [38], seven classes of hybrid methods have been proposed: weighted, switching, mixed, feature combination, cascade, feature augmentation, and meta-level.

First three techniques — weighted, switching, and mixed — assume that several recommendation algorithms simultaneously generate recommendations and a hybridization strategy is applied to combine these recommendations. The simplest strategy is the mixed strategy, in which recommendations from all recommender systems are presented, either in a single list or in multiple recommendation lists. In the weighted strategy, each recommender system returns recommendations along with scores for each recommendation. These recommendations from each system are weighted by the weight assigned to that system. Such weighting systems generally favors items that are considered relevant by many systems (the maajority vote), and recommendations coming from trustful systems — typically a high weight is assigned to systems that have proven to be successful in producing useful recommendations in the past. In the switching strategy, instead of combining recommendations, at a given time, one algorithm is selected based on some conditions to provide recommendations. Both weighted and switching strategies require some meta-information on how to select/combine systems. For example, these can be selected arbitrarily, or learned in similar fashion as stacking in machine learning [54], or using a reinforcement learning technique [159].

The feature combination methods can be seen as a data engineering approach (and not necessarily and algorithm) where multiple different data sources are combined together, and the enhanced final dataset is used to train a single recommendation algorithm. In other words, the hybridization is done on data and not on the algorithms.

The last group of strategies—cascade, meta-level— require staged processing, and feature augmentation, where results of the first stage is used by the second stage recommender system. The methods differ in what is produced in the first stage, and how it is used in the next stage. For example, in cascade method, recommendations produced by a recommender system are refined by the second stage system. In meta-level recommender system, model learned by one technique is used as input to another. Finally, in the feature augmentation strategy, output coming from one technique is added to the feature sets used by the next stage algorithm.

#### 2.2 Evaluation Methodologies

So far I have discussed a number of different recommendation algorithms. The question is how to assess the quality of the recommendations generated by any of these algorithms. In order to select one that performs the best for a given scenario or which parameters to use for each algorithm, typically a set of experiments is run where performance of the algorithms are compared. Following Shani and Gunawardana [149], three main types of evaluation are typically considered — offline experiments, user studies, and online experiments.

## 2.2.1 Offline Evaluation

Offline evaluation is the most common approach used in the literature to evaluate the effectiveness of recommender systems. It is performed by using a pre-collected dataset of users and items rated/chosen by them. The idea behind offline evaluation is based on the assumption that user characteristics at the time the data was collected will be similar to when a system trained on that data is used. The main advantage of this type of evaluation is its efficiency in testing many algorithms at a low cost. However this type of evaluation cannot answer many questions related to the quality of the recommendations and the success of the recommender system in general.

Datasets used in offline evaluation should match as closely as possible the data in the online system. In other words, a great amount of care must be taken in order to make sure no additional biases are introduced, as bias correction is often difficult. These biased may be introduced in the filtering and splitting process that some recommender systems require — favouring algorithms that work better with e.g. denser data — but also at the data collection stage. For example, in the case of explicit feedback, people tend to rate items that they like which means that data is not missed at random.

In order to evaluate the system, offline evaluation requires a way to simulate users interacting with the system. A common approach — popular in machine learning — is to set aside a subset of the recorded interactions and treat it as the ground truth. This splits the dataset into the training set—made of 'known' interactions — and the test set — made of 'hidden' interactions. There are different possibilities for performing the split. One simple approach is to randomly split all the recorded past interactions. This, however, assumes that the time aspect of the system is unimportant and users preferences does not change over time. Other techniques include a time-aware split of the dataset, or time-aware split performed separately for each user, or to select a fixed number of known items per user, or a fixed number of hidden items per user. As there are many options, it is necessary to understand which technique matches the target recommendation scenario better.

#### 2.2.2 Online Evaluation

In online evaluation, users are interacting with a live recommender system and react to the recommendations generated by that system. In many real-world recommender systems, online testing mechanisms are implemented through which experiments are run and algorithms are compared. A common approach is to select a subgroup of users and redirect them to the recommendation algorithm that is being tested and record its performance. It is important to select a subgroup randomly so there is no bias involved, and fair comparisons can be made. Although, in some scenarios users should be systematically chosen if the algorithm is particularly needs to be tested on a certain group of users. Also, conditions of the experiments should be controlled — this means if one aspect of the system is under experimentation, then other components ideally should be fixed.

Performing an online evaluation of a new recommender system comes with risk. For example the algorithm may not work as expected and may cause harm to user experience and hence harming the business. Therefore, in practice, before conducting an online experiment, an offline evaluation is performed and if the results are stable and reasonable, online testing can be performed next.

#### 2.2.3 User Studies

If access to a live system with real users is not possible or too risky, user studies can be conducted as a mean to gather real feedback from users of the system. Generally a number of test subjects (i.e. users) is asked to interact with the system, or a simulation of such interaction is shown and subjects are asked to judge their recommendations. User studies can go even beyond evaluating users' interaction with a recommender system. For example, they can be used to survey users to directly gather more insights on the system and learn users' opinion on certain aspects of the system that could not be collected during offline or online evaluation. Running user studies is however expensive as it requires finding subjects that would be willing to spare their time and to participate in the study. Also the process of selecting subjects is complex as to perform a fair user study we should have a representative sample of the population of users in the system. In addition, in user studies users often have no fear of the costs of their choice. In reality, when a user buys an item, rents an apartment, or even watches a recommended movie, s/he should accept some kind of cost either financial or simply the time s/he spends on using that item. This is usually not the case in user studies. Finally, we should be aware of different biases, coming from different sources. For example, users might act differently than normally if they know that they are being surveyed or observed; others might be more interested in or aware of the topic of the survey, which may have an impact on the results. All these aspects (and many more) can directly affect the results.

#### 2.3 Evaluation Metrics

Besides the type of experiment used to perform the evaluation, another important aspect is the type of the problem that the recommender system is used for. It affects the choice of algorithms, form of output recommendations, and methods used to evaluate the performance. In Steck [155], two types of problems have been identified: the rating prediction task, and the ranking task. In the realy days of research on recommender systems, rating prediction was more common. However, later on researchers realized that ranking is more closely related to real recommendation scenarios ([51]).

#### 2.3.1 Rating Prediction

In rating prediction, the goal of the system is to accurately predict how a user u would rate an item i in the test set. Also it is worth noting that explicit ratings are required in order to perform rating prediction. With respect to evaluation, rating prediction task is commonly evaluated through one of the error metrics such as the Mean-Absolute Error (MAE), or the Root-Mean-Square Error (RMSE):

$$MAE = \frac{1}{|\tau|} \sum_{(u,i)\in\tau} |r_{ui} - \hat{r}_{ui}|$$
(2.4)

$$RMSE = \sqrt{\frac{1}{|\tau|} \sum_{(u,i)\in\tau} (r_{ui} - \hat{r}_{ui})^2}$$
(2.5)

where  $\tau$  represents the test set,  $r_{ui}$  the original rating that user u has given to item i, and  $\hat{r}_{ui}$  is the predicted rating. Values closer to 0 are considered better, however obtaining low error values does not necessary indicate that users will be satisfied with the results of recommendations. This is due to the fact that ratings are not missing at random which is commonly assumed by systems. Missing at random in the context of recommender systems means the probability of a rating to be missing does not depend on its value. If data are missing at random, using only the observed data for statistical analysis yields 'correct' results (e.g., unbiased maximum-likelihood parameter estimates, and predictions) [105, 139]. In contrast, if the data are missing not at random (MNAR), the missing data mechanism cannot be ignored in general, and has to be modeled precisely as to obtain correct results [105, 139]. In other words, if the missing data mechanism is ignored, the resulting recommendation accuracy may be degraded significantly. Also, in many systems, explicit ratings are not available.



Figure 2.2: Contingency Table

#### 2.3.2 Ranking

In the ranking task, instead of predicting how a user would rate an item, the recommender system is asked to select items that it thinks are relevant to the user and recommend them to the user. Commonly the selection of top-N relevant recommendations is considered. Formally the task can be expressed as follows: given a set of items, the system is tasked to create a ranked list  $\ell_u$  for each user u. The ranking is determined by scores  $\hat{r}_{ui}$  calculated by a recommendation system. Then scores are sorted in a descending order, and the top-N items are kept.

Figure 2.2 shows a typical contingency table for a recommendation task. Given a list of recommendation  $\ell$  given to the user, each element in the contingency table are interpreted as follows:

- True Positive (TP): When a recommended item is consumed by the user.
- False Positive (FP): When a recommended item is not consumed by the user.
- **True Negative (TN)**: When an item is not recommended to the user and it is not in user's profile either.
- False Negative (FN): When an item is not recommended to the user but it is in the user's profile.

Common metrics used to evaluate the ranking task are precision, recall, and Normalised Discount Cumulative Gain (NDCG) [84] of recommendations, which can be defined as:

$$Precision = \frac{TP}{TP + FP}$$
(2.6)

Which can be calculated as:

$$Precision(\ell_u) = \frac{\sum_{i \in \ell_u} \mathbb{1}(i \in \rho_u)}{|\ell_u|}$$
(2.7)

where  $\mathbb{1}(.)$  is an indicator function which returns 1 if its argument is true and 0 otherwise. Similarly,

$$Recall = \frac{TP}{TP + FN}$$
(2.8)

Which can be calculated as:

$$Recall(\ell_u) = \frac{\sum_{i \in \ell_u} \mathbb{1}(i \in \rho_u)}{|\rho_u|}$$
(2.9)

If the order of the correct recommendations matters in the list (in contrast to precision and recall where the order is not evaluated), then we can use NDCG which is defined as follows:

$$NDCG(\ell_u) = \frac{1}{iDCG} \sum_{i \in \ell_u} \frac{2^{\mathbb{I}(i \in \rho_u)} - 1}{\log_2(K_i + 1)}$$
(2.10)

where  $k_i$  is the position of the item *i* in the list and *iDCG* is the normalisation factor set as the maximum possible value of *NDCG* for an ideal ranking and it can be measured as:

$$iDCG(\ell_u) = \sum_{k=1}^{\min(|\ell_u|, |\rho_u|)} \frac{1}{\log_2(k+1)}$$
(2.11)

Typically the mean of these values over all test users is reported as the overall system performance.

Although a ranked list of recommendations can be generated via predicting a score for each item and pick the ones with the highest scores, it has been found that better ranking performance can be obtained with algorithms that address the ranking task directly. Such methods are known as learning-to-rank techniques [91], which originated from the machine learning and information retrieval fields, as an alternative to other tasks such as regression and classification that have

been commonly used to create rankings. Learning-to-rank methods can be split into three main categories, depending on the way the ranking model is learned. These are: pointwise [97], pairwise [79, 161], and listwise [152]. The pointwise ranking models are simple models where item ranks are independent of other items and the item's feedback—like ratings—indicates the position in the ranked list—e.g. items rated as 5 should be higher in the list than items rated as 4. In contrast, pairwise and listwise methods consider relationships between items to learn the ranking model. In the pairwise ranking, pairs of items are compared, and the relationship between them defines their position. For example, if most of the times, item *A* has higher utility than item *B*, then item *A* should be recommended higher in the list. Finally, the listwise ranking model tries to learn relationships between the entire set of items for which a user expressed her or his preferences.

## 2.3.3 Non-user Evaluation

User-oriented evaluation of recommender systems is the main type that is commonly considered. While it is important to measure user satisfaction, it is not the only perspective on a system evaluation. For example, businesses do not only deploy recommender systems to improve users' satisfaction, but also to optimise sales, e.g. by promoting less popular items, or items on which the business profits more. These are all important system performance metrics that should also be considered while evaluating a recommender system.

One of the common metrics for measuring how the recommender system performs with respect to covering different items from the catalog is *Aggregate Diversity* (aka catalog coverage) [14]. Aggregate diversity measures the ratio of unique items out of all the available items that the recommender system was able to recommend to different users. This is an important metric as many recommendation algorithms often are biased towards popular items [47] and therefore they have low coverage. Businesses want to make sure the recommendation algorithm can recommend different items to their desired audience. Aggregate diversity of a recommender system is defined as follows:

$$Agg-Div = \frac{\left|\bigcup_{u \in U} \ell_u\right|}{|I|} \tag{2.12}$$

Aggregate diversity counts a recommended item even if it is only recommended once. Therefore it is possible for an algorithm to have a high aggregate diversity while some of the items are recommended only few times. Another metric that gives a better sense of the distributional properties of the recommendations is *Gini Index* which measures how evenly the recommendations are distributed across different items. In an ideal situation where every items are recommended equally, the Gini index is zero. In contrast, an algorithm with extreme inequality of recommendation frequency for different items will have a Gini index of 1. More formally, the Gini index is defined as follows:

$$Gini(L) = 1 - \frac{1}{|I| - 1} \sum_{j=1}^{|I|} (2j - |I| - 1)p(i_j|L)$$
(2.13)

where p(i|L) is the ratio of occurrence of item *i* is *L*.

In addition to the metrics I described in this section, several additional metrics for evaluating other aspects of recommendation related to multi-stakeholder paradigm will be presented and fully discussed in Chapter 4.

# 2.4 Challenges in Recommendation

While many proposed recommendation solutions have been proven to be successful on the task of recommending items to users, there are still many open issues that worsen the experience to some users, or limits the usefulness of such systems. I am going to depict and discuss a few of them now.

In any machine learning system it is generally true that the more data we have, the more accurate models we can learn. The same applies to recommender systems —if we have more useritem interactions, the more we know about what users like and who likes certain items, the more accurate recommendations we can make. However when we have a new user joining the platform,
or a new item is added, we know very little about them which makes the recommendation task much harder or technically impossible and of higher risk. This type of a problem is commonly known as the cold start problem [95]. Generally, content-based models are less prone to suffer from the cold start problem since the only information needed is some content information about the new item or some features about the new user. However, collaborative filtering methods cannot generate personalized recommendations for a completely new user or recommend a new item to the users in a personalized way since no previous interaction data exist for those type of users or items. Nevertheless, they can still generate recommendations that are non-personalized such as randomly recommending items or recommending the most popular items.

In addition, certain types of recommender systems — mostly content-based methods, are known to be affected by the filter bubble, or over-specialisation, where recommendations are similar to what the user already knows and this causes the user to be stuck in a bubble. While this might be seen as a good indicator of the ability of the recommender system to identify user interests and provide personalized recommendations, such recommendations may not be engaging and would not help the user discover new content. Similarly, recommending lists of similar items, even if relevant, may not provide the value that the users are looking for, and more diverse recommendations are rather expected [177, 180].

Finally, many online systems have been known to suffer from the so-called the long-tail effect, or the popularity bias [23]. Most of the interactions that are recorded by a system happen to refer to a relatively small set of items, commonly known as the short-head, and the rest is called the long-tail. Due to such bias in the interactions towards the head, many collaborative filtering techniques are prone to also recommend these popular items, making them even more popular [178]. This situation is not desirable from both user and business perspectives—popular items are likely to be already known by the users but also of little utility as they are less surprising or engaging; where for businesses, recommendations focusing on most popular items lead to poor utilization of the whole catalogue of items.

#### 2.4.1 Diversity, Novelty and Long-tail

Traditionally, the focus of recommender systems has been put on recommending as many relevant items as possible and maximising the overall accuracy of a system. While high accuracy is desirable, Herlocker et al. [74] stated that high accuracy on its own may not be enough to provide useful and engaging recommendations, and that other properties should be simultaneously considered. In work of Kaminskas and Bridge [88] and McNee et al. [110], number of properties have been identified, such as novelty, coverage, and diversity.

Starting from novelty, looking at the Merriam Webster dictionary, it is defined as 'the quality or state of being new, different, and interesting'<sup>2</sup>. In the context of recommendations, novelty typically refers to two aspects — an item being unknown to the user, and an item being different from what the user has seen before [88]. The former of the above two reflects the global, system utility of recommendations, and the latter user-level novelty. Simply put, the global aspect mainly describes whether a recommended item is known to the user or not. Typically, popular items are already known to the users and therefore recommending them would result in low novelty.

Diversity refers to the fact that the items in a given recommendation list is better to be different from each other so the user has enough choice [88,163]. For instance, in a movie recommendation, a list containing Scream1, Scream2 and Scream3 is not a good recommendation even if the user is interested in horror movies since there is not enough choice in the list for the user and it would have been better if the list contained a diverse set of horror movies.

The main property that I focus on in this dissertation is the novelty of the recommendations. Popular items have low novelty to the users while less popular, long-tail items are considered to be more novel as the users are less likely to know about them. In Chapter 5 I will describe the long-tail and popularity bias in detail.

<sup>&</sup>lt;sup>2</sup> http://www.merriam-webster.com/dictionary/novelty

# Chapter 3

# Multi-stakeholder Recommendation

### 3.1 Introduction

Recommender systems provide personalized information access, supporting e-commerce, social media, news, and other applications where the volume of content would otherwise be overwhelming. They have become indispensable features of the Internet age, found in systems of many kinds. One of the defining characteristics of recommender systems is personalization. In research contexts, recommender systems are typically evaluated on their ability to provide items that satisfy the needs and interests of the end user. Such focus is entirely appropriate. Users would not make use of recommender systems if they believed such systems were not providing items that matched their interests. Still, it is also clear that, in many recommendation domains, the user for whom recommendations are generated is not the only stakeholder in the recommendation outcome. Other users, the providers of products, and even the system's own objectives may need to be considered when these perspectives differ from those of end users.

In many practical settings, such as e-commerce, recommendation is viewed as a form of marketing and, as such, the economic considerations of the retailer will also enter into the recommendation function [101,124]. A business may wish to highlight products that are more profitable or that are currently on sale, for example. More recently, system-level objectives such as fairness and balance have been considered by researchers, and these social-welfare-oriented goals may at times run counter to individual preferences. Sole focus on the end user hampers researchers' ability to incorporate such objectives into recommendation algorithms and system designs.

I believe that, far from being special "edge cases", these examples illustrate a more general point about recommendation, namely, that recommender systems often serve multiple goals and that the purely user-centered approach found in most academic research does not allow all such goals to enter into their design and evaluation. What is needed is an inclusive approach that expands outward from the user to include the perspectives and utilities of multiple stakeholders.

It is relevant to note a shift that occurred in microeconomics in the early part of the 21st century with the development of the theory of multi-sided platforms [61, 135]. Prior to that time, the traditional business model focused on a firm's ability to produce products and deliver them to customers at a price that could ensure profitability. By contrast, multi-sided platforms create value by bringing buyers and sellers together, reducing search and transaction costs. Many online systems are exactly such multi-sided platforms [62].

As noted above, when it comes to the study of personalized information access in the form of recommender systems, academic research has, with few exceptions, examined only a single side of these interactions. The stage was set historically by the first recommender systems implementations, which either operated on objects with no associated price (newsgroup posts [96]) or were external to any commerce associated with their recommendations (such as non-commercial music, movie, and restaurant recommenders [35,40,150]). These systems brought users and products together, but they were not themselves party to any transactions. While academic research has largely concentrated on the user, commercial systems have regularly taken a broader view of recommendation objectives [116,136]. There is, therefore, a gap between the complexity of real-world applications of recommender systems and those on which academic research has focused.

The integration of the perspectives of multiple parties into the design of recommender systems is the goal underlying the sub-field of *multi-stakeholder recommendation* [1, 3, 42, 116] which I first introduced in [42] and I describe in detail in this dissertation. This chapter is intended to describe the current state of the art in multi-stakeholder recommendation research and to outline research questions that should be addressed to support the demands of recommendation applications in environments where the perspectives of multiple parties are important.

## 3.2 Recommendation Stakeholders and Multi-stakeholder Paradigm

The concept of a stakeholder appears in business management literature as a way to discuss the complexities of corporate governance. According to [69], the term 'stakeholder' appears to have been invented in the early '60s as a deliberate play on the word 'stockholder' to signify that there are other parties having a 'stake' in the decision-making of the modern, publicly-held corporation in addition to those holding equity positions.

In his classic work, *Strategic Management: A Stakeholder Perspective*, Freeman extends older definitions that emphasize a "stake" as a kind of investment, and instead defines stakeholders as "any groups or individuals that can affect, or are affected by, the firm's objectives" [64], pg. 25. I adopt this definition for our aims, focusing specifically on recommender systems.

**Definition 3.2.1** A *recommendation stakeholder* is any group or individual that can affect, or is affected by, the delivery of recommendations to users.

As recommender systems are elements of an organization's operations, they will necessarily inherit the large and wide-ranging set of stakeholders considered in the management literature. However, only some of these stakeholders will be particularly salient in the generation of recommendations. In this dissertation, I will consider three key groups of stakeholders who are particularly close to the recommendation interaction:

- **Consumers (aka users):** The consumers are the end-users who receive / consume recommendations. They are the individuals whose choice or search problems bring them to the platform, and who expect recommendations to satisfy those needs.
- **Providers (aka suppliers):** The (item) suppliers are those entities that supply or otherwise stand behind the recommended objects.
- **System:** The final category is the organization itself, which has created a platform and associated recommender system in order to match consumers with suppliers. The platform may be a retailer, e-commerce site, broker, or other venue where consumers seek recommendations.

Of course, none of these are stakeholder groups are unitary entities — not even the system, which stands in for various internal groups within an organization who may have different, possibly competing demands on a recommender system. Differentiation among stakeholder groups may be necessary, depending on the application. This taxonomy does, however, highlight an important consequence of a multi-stakeholder perspective, namely the foregrounding of the multi-sided nature of recommendation, which has been slow to emerge in the research literature.

Figure 3.1 contrasts the traditional view of recommendations versus the multi-stakeholder one. Traditionally, the system is assumed to generate a set of recommendations to the users and its performance is measured from the users' perspective with an average value across all users. In other words the differences between users and groups of users are often omitted from the evaluation. However, even if we only focus on users as the sole stakeholders, they are not a single entity but rather different individuals with certain characteristics who might belong to different groups often protected by law (e.g. gender, race etc. ). In addition, the multi-stakeholder schema considers the suppliers of the items as another important party in the recommendation process whose needs and satisfaction should be taken into account. The recommender system might even give insights to the suppliers based on how the users interact with their items and the feedback they receive from the users. On top of that, some side stakeholders may also impose certain constraints and limitations on the recommendations. For example, on a music recommendation system, the society or law regulations may also be a considered as stakeholders to enforce certain human values in the recommendations such as recommending local versus international music in a fair way or making sure female artists are fairly represented in the recommendations given to different users. Of course, the term fair could be interpreted in many different ways [115] but often there are certain law regulations that businesses need to follow [70].

The multi-stakeholder perspective on recommendation can manifest itself in various aspects of recommender systems research and development. We may adopt a multi-stakeholder view of evaluation: asking the question of how different groups of stakeholders are impacted by the recommendations a system computes. A developer may employ the multi-stakeholder perspective in



(a) Traditional View of Recommendation



(b) Multi-stakeholder View of Recommendation



designing and optimizing an algorithm, incorporating the objectives of multiple parties and balancing among them. Finally, the creation of the recommender itself may be a multi-stakeholder endeavor, in which different stakeholders participate in developing a design.

# 3.2.1 Multi-stakeholder Evaluation

In research settings, recommender systems traditionally are evaluated from the users' perspective. Metrics such as precision, recall, NDCG, diversity, novelty, etc. all capture different aspects of recommendation quality as experienced by the end user. In on-line testing, click-through rate, dwell time, and other interaction metrics capture similar types of outcomes. These metrics are typically averaged over all users to produce a single outcome. This methodology in entirely reasonable and logical as, in the end, users are one of the most important stakeholders of any recommender system. We will take it as a given that any recommender system evaluation will include an outcome of this type, the *user summary* evaluation.

A multi-stakeholder perspective on evaluation, however, brings to light additional aspects of system performance that may be quite important. As mentioned above, multi-sided platforms such as eBay, Etsy, or AirBnB, have a key business requirement of attracting and satisfying the needs of providers as well as users. A system of this type will need to evaluate its performance from the provider perspective.

As mentioned above, even when only a single group of stakeholders is under consideration, a methodology that relies on a single point estimate of system performance under some metric may miss differences among stakeholder groups, even among users, who are typically the target of evaluation. Stakeholder theory recognizes that subgroups within the stakeholder categories may experience a range of different impacts from a firm's decisions [64]. In recommendation, recent work has shown that, depending on the algorithm, male and female users may experience different outcomes in movie recommendations [59]. Therefore, a multi-stakeholder evaluation is one that includes different stakeholder groups in its assessment, in addition to the user summary evaluation. **Definition 3.2.2** A *multi-stakeholder evaluation* is one in which the quality of recommendations is assessed across multiple groups of stakeholders, in addition to a point estimate over the full user population.

A multi-stakeholder evaluation may entail the use of different kinds of metrics and different evaluation methodologies than typically encountered in recommender systems research. For example, typical cross-validation methodologies that distribute user profiles between test and training sets may not yield reliable results when assessing outcome for other stakeholders, especially providers. We will take up the issue of provider-side metrics in Section 3.4.5.1.

# 3.2.2 Multi-stakeholder Algorithms

In implementing a given recommender system, a developer may choose to use the metrics associated with multi-stakeholder evaluation in algorithm design, implementation, and optimization. In general, this will entail balancing the objectives of different stakeholders, as it is unlikely that the optimal solution for one will be the best for all. Some solutions may combine all such objectives into a single optimization, a challenging approach given the methodological complexities noted above; others use a multi-stage approach incorporating different stakeholder concerns throughout a pipeline. Both approaches are discussed in the examples in Chapter 6.

Multi-stakeholder algorithms are particularly differentiated from typical recommendation approaches when the stakeholder concerns lie on different sides of the recommendation platform. For example, it is not difficult to change the loss function associated with a factorization algorithm to prefer balanced outcomes over multiple subgroups of users, rather than a simple overall mean. However, system or provider objectives may be orthogonal to user concerns and form a separate optimization problem that cannot be simply combined with the users'. I therefore define a multistakeholder recommendation algorithm with special attention to this subclass.

**Definition 3.2.3** A *multi-stakeholder recommendation algorithm* takes into account the preferences of multiple parties when generating recommendations, especially when these parties are on different

sides of the recommendation interaction.

#### 3.2.3 Multi-stakeholder Design

Beyond implementing metrics and tuning algorithms, any fielded recommender system will also go through phases of design in which the system's role within a particular platform and its specific requirements are formulated. System designers may choose to engage in a design process such as participatory design [93] in which external stakeholders are incorporated into decision making. Although these techniques are well-established in the HCI community, they have not seen much discussion in recommender systems research. Anecdotal information suggests that commercial platforms with multiple stakeholders do engage in stakeholder consultation with item providers in particular [148].

**Definition 3.2.4** A *multi-stakeholder design process* is one in which different recommendation stakeholder groups are identified and consulted in the process of system design.

### 3.2.4 The Landscape of Multi-stakeholder Recommendation

The above list of the key recommendation stakeholders provides an outline for understanding and surveying the different types of multi-stakeholder recommendation. We can conceptualize a recommender system as designed to maximize some combined utility associated with the different stakeholders, and consider how different types of applications yield different stakeholder concerns.

# 3.2.4.1 Consumer-side issues

If we concentrate only on the individuals consuming recommendations, multi-stakeholder issues arise when there are tradeoffs or disparities between groups of users in the provision of recommendations. For example, [59] explored the performance of recommender system algorithms on users belonging to different demographic groups (gender, age) and observed that some algorithms perform better for certain groups than others. Other researchers have found that users with niche or unusual tastes can be poorly served by particular recommendation algorithms, lapses that are not detectable from point estimates of accuracy measures [5,7,67].

In some settings, such differences in system performance for different users may be considered examples of unfair treatment. For example, in Chapter 6 we will see that the recommendation algorithms perform different for users with different levels of tendency towards popular items. In particular, we will see that the recommendations for users who have lesser tendency towards popular items have a higher deviation from what they expected to receive from the recommender system.

#### 3.2.4.2 Supplier-side issues

As noted above, multi-sided platforms need to satisfy both the consumers of recommendations and the suppliers of items that are being recommended. The health of such a platform depends both on a user community and a catalog of items of interest. Suppliers whose items are not recommended may experience poor engagement from users and lose interest in participating in a given platform. Platforms that facilitate peer-to-peer interactions, such as the online craft marketplace Etsy, may be particularly sensitive to the need to attract and retain sellers.

Depending on the application, suppliers may have particular audiences in mind as appropriate targets for their items. A well-known example is computational advertising in which advertisers seek particular target markets and ad platforms try to personalize ad presentation to match user interests [175]. In this application, market forces, expressed through auction mechanisms, serve to mediate between supplier interests. In other cases, such as online dating, preferences may be expressed on both sides of the interaction but it is up to the recommender system to perform appropriate allocation.

#### 3.2.4.3 System/platform issues

In many real-world contexts, the system may gain some utility when recommending items, and is therefore a stakeholder in its own right. Presumably, an organization creates and operates a recommender system in order to fulfill some business function and generally that is to enhance user experience, lower search costs, increase convenience, etc. This would suggest a consumer stakeholder point of view is sufficient.

However, there are cases in which additional system considerations are relevant, as noted above, and internal stakeholders have an impact on how a recommender system is designed. For example, in an e-commerce platform, the profit of each recommended item may be a factor in ordering and presenting recommendation results. This marketing function of recommender systems was apparent from the start in commercial applications, but rarely included as an element of research systems.

Alternatively, the system may seek to tailor outcomes specifically to achieve particular objectives that are separate from either provider or consumer concerns. For example, an educational site may view the recommendation of learning activities as a curricular decision and seek to have its recommendations fit a model of student growth and development. Its utility may, therefore, be more complex than a simple aggregation of those of the other stakeholders.

# 3.2.4.4 Other stakeholders

Complex online ecosystems may involve a number of stakeholders directly impacted by recommendation delivery beyond item suppliers and recommendation consumers. For example, an online food delivery service, such as UberEats, depends on delivery drivers to transport meals from restaurants to diners. Drivers are affected by recommendations delivered to users as the efficiency of routing and the distribution of order load will be a function of which restaurants receive such orders.

# 3.3 Related Research

Multi-stakeholder recommendation brings together research in a number of areas within the recommender systems community and beyond: (1) in economics, the areas of multi-sided platforms and fair division; (2) the growing interest in multiple objectives for recommender systems, including such concerns as fairness, diversity, and novelty; and, (3) the application of personalization to matching problems.

# 3.3.1 Economic foundations

The study of the multi-sided business model was crystallized in the work of [135] on what they termed "two-sided markets." Economists now recognize that such contexts are often multi-sided, rather than two-sided, and that "multi-sidedness" is a property of particular business platforms, rather than a market as a whole [62].

Many of today's recommender systems are embedded in multi-sided platforms and hence require a multi-sided approach. The business models of multi-sided platforms are quite diverse, which means it is difficult to generalize about multi-stakeholder recommendation as well. A key element of the success of a multi-sided platform is the ability to attract and retain participants from all sides of the business, and therefore developers of such platforms must model and evaluate the utility of the system for all such stakeholders.

The theory of just division of resources has a long intellectual tradition going back to Aristotle's well-known dictum that "Equals should be treated equally." Economists have invested significant effort into understanding and operationalizing this concept and other related ideas. See [114] for a survey. In recommendation and personalization, we find ourselves on the other side of Aristotle's formulation: all users are assumed unequal and unequal treatment is the goal, but we expect this treatment to be consistent with diverse individual preferences. Some aspects of this problem have been studied under the subject of social choice theory [27]. However, there is not a straightforward adaptation of these classical economic ideas to recommendation applications as the preferences of users may interact only indirectly and in subtle ways. For example, if a music player recommends a hit song to user A, this will not in any way impact its desirability or availability to user B. On the other hand, if a job recommender system recommends an appealing job to user A, it may well have an impact on the utility of the same recommendation to user B who could potentially face an increased competitive environment if she seeks the same position.

### 3.3.2 Multi-objective recommendation

Multi-stakeholder recommendation is an extension of recent efforts to expand the considerations involved in recommender system design and evaluation beyond simple measurements of accuracy. There is a large body of recent work on incorporating diversity, novelty, long tail promotion and other considerations as additional objectives for recommendation generation and evaluation. See, for example, [2,4,81,153,163,181]. There is also a growing body of work on combining multiple objectives using constraint optimization techniques, including linear programming. See, for example, [15,16,80,87,137,160]. These techniques provide a way to limit the expected loss on one metric (typically accuracy) while optimizing for another, such as diversity. The complexity of these approaches increases exponentially as more constraints are considered, making them a poor fit for the general multi-stakeholder case. Also, for the most part, multi-objective recommendation research concentrates on maximizing multiple objectives for a single stakeholder, the end user.

Another area of recommendation that explicitly takes a multi-objective perspective is the area of health and lifestyle recommendation. Multiple objectives arise in this area because users' short-term preferences and their long-term well-being may be in conflict [104, 128]. In such systems, it is important not to recommend items that are too distant from the user's preferences – even if they would maximize health. The goal to be persuasive requires that the user's immediate context and preferences be honored.

Fairness is an example of a system consideration that lies outside the strict optimization of an individual user's personalized results. Therefore, recent research efforts on fairness in recommendation are also relevant to this work [41,89,90,102,111,174]. The multi-stakeholder framework provides a natural "home" for such system-level considerations, which are otherwise difficult to integrate into recommendation. See [1] for a more in-depth discussion.

## 3.3.3 Personalization for matching

The concept of multiple stakeholders in recommender systems is suggested in a number of prior research works that combine personalization and matching. The earliest work on two-sided matching problems [138] assumes two sets of individuals, each of which has a preference ordering over possible matches with the other set. The task to make a stable assignment has been shown to have an  $O(n^2)$  solution. This formulation has some similarities to reciprocal recommendation. However, it assumes that all assignments are made at the same time, and that all matchings are exclusive. These conditions are rarely met in recommendation contexts, although extensions to this problem formulation have been developed that relax some of these assumptions in online advertising contexts [30].

Researchers on reciprocal recommendation have looked at bi-lateral considerations to ensure that a recommendation is acceptable to both parties in the transaction. A classical example is on-line dating in which both parties must be interested in order for a match to be successful [127, 172]. Other reciprocal recommendation domains include job seeking [137], peer-to-peer "sharing economy" recommendation (such as AirBnB, Uber and others), on-line advertising [77], and scientific collaboration [107, 162]. See [1] for a detailed discussion.

The field of computational advertising has given considerable attention to balancing personalization with multi-stakeholder concerns. Auctions, a well-established technique for balancing the concerns of multiple agents in a competitive environment, are widely used both for search and display advertising [112, 175]. However, the real-time nature of these applications and the huge potential user base makes recommender-style personalization computationally intractable in most cases.

# 3.4 Evaluation

At this point in the development of multi-stakeholder recommendation research, there is a diversity of methodological approaches and little agreement on basic questions of evaluation. In part, this is a reflection of the diversity of problems that fall under the multi-stakeholder umbrella. Multi-stakeholder evaluation and algorithm development do not always use the same methodologies.

A key difficulty is the limited availability of real-world data with multi-stakeholder characteristics. The reason is these type of data are highly business-critical and often including private information such as margins associated with each supplier and the commissions negotiated by the platform. Close collaboration is required to obtain such sensitive proprietary data. Some researchers have built such collaborations for multi-stakeholder research, but progress in the field requires replicable experiments that proprietary data does not support. Areas of multi-stakeholder research that involve public, rather than private, benefit may offer advantages in terms of the availability of data: see, for example, the data sets available from the crowd-funded educational charity site DonorsChoose.org<sup>1</sup>.

## 3.4.1 Simulation

In the absence of real-world data with associated valuations, researchers have typically turned to simulations. Simulated or inferred supplier data is useful for transforming publiclyavailable recommendation data sets in standard user, item, rating format into ones that can be used for multi-stakeholder experimentation. The experiments in [158] provide an example of this methodology: each item in the MovieLens 1M data set was assigned to a random supplier, and the distribution of utilities calculated. To capture different market conditions, the experimenters use two different probability distributions: normal and power-law. There is no accepted standard for producing such simulations and what are reasonable assumptions regarding the distribution

<sup>&</sup>lt;sup>1</sup> https://data.donorschoose.org/explore-our-impact/

of provider utilities or the formulation of system utilities, except in special cases.

Researchers have also used objective aspects of data sets to infer proxy attributes for multistakeholder evaluation. In [42], the first organization listed in the production credits for each movie was treated as the provider – a significant over-simplification of what is a very complex system of revenue distribution in the movie industry. In other work, global metrics such as network centrality [19] have been used to represent system utility for the purposes of multi-stakeholder evaluation. [43] demonstrated an alternate approach to generate synthetic attribute data based on behavioral characteristics that can be used to evaluate system-level fairness properties.

More sophisticated treatments of profitability and recommendation are to be found in the management and e-commerce literature, some using public data as seen in [9, 49, 121], but these techniques and associated data sets have not yet achieved wide usage in the recommender system community.

## 3.4.2 Models of utility

A multi-stakeholder framework inherently involves the comparison of outcomes across different groups of individuals that receive different kinds of benefits from the system. In economic terms, this entails utility calculation and comparison. As with data, different researchers have made different assumptions about what types of utilities accrue from a recommender system and how they are to be measured. A standard assumption is that the output of a recommendation algorithm can be treated as an approximation of user utility. Yet, research has confirmed that users prefer diverse recommendation lists [129], a factor in tension with accuracy-based estimates of user utility.

These are, of course, more concentrated on the short-term definitions of utility and things that can be measured from the users feedback in a single run of recommendation generation. More research is therefore required to understand the potential positive and negative long-term effects of profit-aware recommendation and other strategies that are not strictly user-focused. Future models could also consider the price sensitivity and willingness-to-pay of individual consumers in the recommendation process.

#### 3.4.3 Off-line experiment design

A standard off-line experimental design which I described in Chapter 2 makes a bit less sense in a multi-stakeholder context, and this is where the essential asymmetry of the stakeholders comes into play. Suppliers are, in a key sense, passive – they have to wait until users arrive at the system in order to have an opportunity to be recommended. The randomized cross-fold methodology measures what the system can do for each user, given a portion of their profile data, the potential benefit to be realized if the user visits and a recommendation list is generated. Evaluating the supplier side under the same conditions, while a commonly-used methodology, lacks a similar justification.

A more realistic methodology from the suppliers' point of view is a temporal one, that takes into account the history of the system up to a certain time point and examines how supplier utilities are realized in subsequent time periods. See [44] for a comprehensive discussion of time-aware recommender systems evaluation. However, time-aware methods have their own difficulties, forcing the system to cope with cold-start issues possibly outside of the scope of a given project's aims. Therefore, for certain models of utility where the dynamics of suppliers (such as price, profit margin etc) may be of less important, offline evaluation can be still valuable. For example, our evaluation of the popularity bias in recommendation from the perspectives of different stakeholders is done in an offline setting as we will see in Chapter 4 and 5.

#### 3.4.4 User studies

User studies are another instrument available to researchers that has not been extensively applied to multi-stakeholder recommender systems. As usual for such studies, the development of reliable experimental designs is challenging as the participants' decision situation typically remains artificial. Furthermore, as in the study by Azaria et al. [28], familiarity biases might exist – in their study participants were willing to pay more for movies that they already knew – which

have been observed for other types of user studies in the recommender systems domain [82]. Ultimately, more field tests – even though they are typically tied to a specific domain and specific business model – are needed that give us more insights into the effects of multi-stakeholder recommendations in the real world.

#### 3.4.5 Metrics

The building block of multi-stakeholder evaluation is the measurement of the utility each of the stakeholders gets within a recommendation platform. Common evaluation metrics such as RMSE, precision, NDCG, diversity, etc. are all different ways to evaluate the performance of a recommender system from the user's perspective. As noted above, these measures are implicitly a form of system utility measure as well: system designers optimize for such measures under the assumptions that (1) higher accuracy metrics correspond to higher user satisfaction and (2) higher user satisfaction contributes to higher system utility through customer retention, trust in the recommendation provided, etc. However, the formulation of multi-stakeholder recommendation makes it possible to characterize and evaluate system utility explicitly.

Typically, evaluation metrics are averaged over all users to generate a point score indicating the central tendency over all users. However, it is also the case that in a multi-stakeholder environment additional aspects of the utility distribution may be of interest. For example, in an e-commerce context, suppliers who receive low utility may leave the eco-system, suggesting that the distribution of supplier utility may be important as well as the mean. One suggested practice would be to report results for significant user groups within their populations as I will discuss in Chapter 4 and 6.

## 3.4.5.1 Supplier metrics

When evaluating the utility of a recommender system for a particular supplier, we may take several different stances. One views the recommender as a way to garner user attention. In this case, the relevance of an item to a user may be a secondary consideration. Another perspective views the recommender as a source of potential leads. In this view, recommending an item to uninterested users is of little benefit to the supplier. In the first situation, simply counting (with or without a rank-based discount) the number of times a supplier's products appears in recommendation lists would be sufficient. In the second situation, the metric should count only those recommendations that were considered "hits", those that appear positively rated in the corresponding test data.

Another supplier consideration may be the reach of its recommended products across the user population. A metric could count the number of unique users to whom the supplier's items are recommended. Multi-stakeholder applications may differ in their ability to target specific audiences for their items. In a targeted system, it would make sense to consider reach relative to the target population. For example, in an online dating application where the user can specify desired properties in a match, an evaluation metric might be the fraction out of the target audience receiving the recommendation.

Finally, where the consideration is the accuracy of system's predictions, we can create a supplier-specific summary statistic of a measure like RMSE. [60] uses this method to examine differences in error when recommending books by male and female authors. Since the statistic by itself is not that useful for a single provider, a better metric would indicate the provider's position relative to other providers in the overall distribution

#### 3.4.5.2 System metrics

System utility may, in many cases, be a simple aggregation of the utilities of other parties. For example, in a simple commission-oriented arrangement, the profit to the system might be some weighted aggregate of the *Hits* metric, taking item price and commission rate into account. However, other cases arise where the system has its own targeted utility framework.

An important such context is algorithmic fairness [60,173]. In general, we should not expect that suppliers will care if the system is fair to others as long as it provides them with good outcomes. Any fairness considerations and related metrics will therefore be ones defined by system considerations.

As we will see in Chapter 5, popularity bias in recommendation is a multi-stakeholder problem that impacts different sides of the recommendation platform including the users and the suppliers. It also might create challenges for the system itself if, for example, the system designer cares about giving certain amount of exposure to different items with different levels of popularity. In next chapter, I discuss our methodology for measuring the popularity bias from the perspective of different stakeholders: users, suppliers, and the system.

# Chapter 4

## Data and Methodology

In this chapter we describe the details of the datasets we used for the experiments in this dissertation and several evaluation metrics that we introduce for measuring the multi-stakeholder aspect of the popularity bias.

# 4.1 Datasets

In Chapter 3, we highlighted the main stakeholders of a given recommender system: users, suppliers and the system itself. To incorporate our analysis for the suppliers, we needed datasets where the item suppliers could be identified. We found two publicly available datasets for our experiments: the first one is a sample of the Last.fm (LFM-1b) dataset [147] used in [55]. The dataset contains user interactions with songs (and the corresponding albums). We used the same methodology in [55] to turn the interaction data into rating data using the frequency of the interactions with each item (more interactions with an item will result in higher rating). In addition, we used albums as the items to reduce the size and sparsity of the item dimension, therefore the recommendation task is to recommend albums to users. We considered the artists associated with each album as the supplier. We removed users with less than 20 ratings so only consider users for which we have enough data. The resulting dataset contains 274,707 ratings by 2,697 users to 6,006 albums. Total number of artists is 1,998.

The second dataset is the MovieLens 1M dataset [72]<sup>1</sup>. This dataset does not have the

<sup>&</sup>lt;sup>1</sup> Our experiments showed similar results on MovieLens 20M, and so we continue to use MovieLens 1M for efficiency reasons.

**Table 4.1:** Number of users |U|, number of items |I|, number of ratings |R|, and number of suppliers  $|\mathcal{P}|$  in two datasets MovieLens and Last.fm

	U	I	R	$ \mathcal{P} $
MovieLens	6,040	3,043	995,487	831
Last.fm	2,697	6,006	274,707	1,998

information about the suppliers. We considered the director of each movie as the supplier of that movie and we extracted that information from the IMDB API. Total number of ratings in the MovieLens 1M data is 1,000,209 given by 6,040 users to 3,706 movies. Overall, we were able to extract the director information for 3,043 movies reducing the ratings to 995,487. The total number of directors is 831.

For convenience, the characteristics of both datasets is shown in Table 4.1.

# 4.2 **Recommendation Algorithms**

As we mentioned in Chapter 2, there are different types of recommendation algorithms each of which have their own cons and pros. Some are neighborhood-based and some are based on matrix factorization methods. We also mentioned content-based models where the content information of the items (or users) is used to generate the recommendations. In this dissertation, our main focus is on collaborative filtering models where no content information is used in the algorithm. Both neighborhood-based and factorization-based models are considered collaborative filtering techniques since they use the rating (or interaction data) information given by the users to the items. We have used several well-known recommendation algorithms in the following chapters that cover both neighborhood-based models and also factorization-based approaches (either rating prediction-based or ranking-based).

• Alternative Least Square Ranking (RankALS) [161]: This method is a pair-wise learningto-rank technique that directly optimizes the order of the items in the recommendation lists without the need for calculating the predicted ratings for all items and sorting them. The idea is, if one item is ranked higher than another item in rating data, this order should be kept in the recommended list. The algorithm tries to minimize the pairwise ranking discrepancies between every pair of items in rating data.

- **Biased Matrix Factorization (Biased-MF)** [99]: This is another factorization-based technique but, unlike RankALS, it does not optimize for ranking but rather seeks to minimize the distance between the predicted rating and the original rating for any given ;user,item; pair. Therefore, the optimization is done in the rating prediction step and once the predicted ratings for ;user,item; pairs are calculated, they are sorted and the top *n* items will be returned as the recommendations.
- User-based Collaborative Filtering (User-CF) [17]: This algorithm is a neighborhoodbased technique meaning the rating prediction is directly using the similarities between users. For any given user, the algorithm finds top most similar users and uses their rating information for items that the target user has not rated. In order to find the similarity between users, distance metrics such as Pearson Correlation or Cosine similarity are used.
- Item-based Collaborative Filtering (Item-CF) [144]: Similar to *User-CF*, this algorithm is also neighborhood-based but, unlike *User-CF* that leverages similarity between users, it uses the similarity between items that the user has rated before and other items. Again, the similarity calculation is done using distance metrics such as Pearson Correlation or Cosine similarity. The algorithm then recommends most similar items.

# 4.3 Stakeholder Grouping

Throughout this dissertation, for better illustration of certain patterns and behavior of different algorithms, I use grouping (either items, users, or suppliers) so I can discuss how each algorithm performs for items, users, or suppliers in different groups. Many researchers have used demographic information such as age, gender, race etc to define groups of users and to measure demographic parity between those groups [21, 59, 60]. However, these works assume there is no intrinsic differences among individuals in different groups which is not supported by prior work where significant differences have been found [29, 57]. In this dissertation, our grouping strategy is consistent with the problem I am addressing. The problem I am concentrating on is popularity bias and its impact of different stakeholders and hence a natural way of grouping would have the popularity component in its consideration.

For all stakeholder types (items, users, and suppliers), the grouping is based on the popularity of the individuals or their interest in popular items in each stakeholder type. In other words, items are grouped based on their popularity, users are grouped based on their interest towards popular items, and suppliers are grouped according to the average popularity of their items. Before going any further, it is important to define what exactly I mean by popularity.

**Definition 4.3.1 Item Popularity:** The popularity of an item is defined as the ratio of users who have interacted (or rated) that particular item. More formally:

$$Pop(i) = \frac{\sum_{u \in U} \mathbb{1}(i \in \rho_u)}{|U|}$$
(4.1)

where  $\rho_u$  is the list of rated (or interacted with) items by user *u*.

Therefore, the more users rate or interact with a certain item, the higher its popularity will be. In a sense, popularity of an item measures the amount of attention the items receives.

Here is how the grouping is done for each stakeholder type:

• Item Grouping: Generally speaking, in recommendation domains there are a small number of very popular items and many other items have either medium popularity or are extremely non-popular. These few popular items are referred to as the *short-head* in the literature which take up roughly 20% of the ratings according to the Pareto Principle [141]. The rest of the items in a long-tail distribution are usually divided into two other parts [47] as I also do in this dissertation: Tail items (*T*) are the larger number of less popular items which collectively take up roughly 20% of the ratings<sup>2</sup>, and *Mid* items (*M*) include a larger number of items in between that receive around 60% of the ratings, collectively.

<sup>&</sup>lt;sup>2</sup> Tail items may also be new items that have yet to find their audience and will eventually become popular. In this way, popularity bias is related to the well-known item cold-start problem in recommendation.





**Figure 4.1:** Profile size of users in different user groups. Numbers within the parentheses show the average profile size of the users in each group



**Figure 4.2:** Number of songs / movies for artists / directors in each supplier group. The first number in the parentheses shows the size of each group and the second number shows the average number of songs / movies for the suppliers in each group.

• User Grouping: Users have different interest in popular items. Some are mainly focused on popular and blockbuster products and some are more inclined towards less popular, or niche, items. There are also some who fall in between and they have a more diverse taste in terms of the popularity of the items they tend to consume. In this dissertation, I sort the users from the highest interest towards popular items to the lowest based on the ratio of different item groups (*H*, *M*, and *T*) in their profiles. Users are lexicographically sorted according to the ratio of *H*, *M*, and *T* items in their profiles with priority given to *H* followed by *M* and *T*. Sorted users are divided into three equal-sized bins *G* = {*G*<sub>1</sub>, *G*<sub>2</sub>, *G*<sub>3</sub>} from most popularity-focused (*G*<sub>1</sub>) to least (*G*<sub>3</sub>).

Figure 4.1 shows the distribution of profile size of the users within each group for both MovieLens and Last.fm datasets. We can see that, on MovieLens, groups that have higher interests towards popular movies have smaller average profile size compared to the groups with lesser interest in popular movies. In fact, the average profile size for the users in  $G_3$ (group with the least interest in popular movies) is almost five time larger than the highlypopularity-focused group ( $G_1$ ). Interestingly, this trend on Last.fm seems to be the exact opposite where the average profile size of the user groups who have higher focus on popular songs is larger. This is indeed an interesting distinction between the movie and music domain.

• Supplier Grouping: Suppliers also have different levels of popularity. In movies, some directors are making more popular movies and therefore they have a higher popularity according to our definition of popularity which is the average popularity of the items for that supplier. Similar to items, supplier popularity has also an extreme long-tail shape and therefore I need to group them in a way that the resulting groups vary in terms of popularity. Therefore, I, again, use the Pareto Principle for the suppliers to determine the appropriate cut-off points to create the groups. I have defined three supplier groups based on their popularity  $S = \{S_1, S_2, S_3\}$ :  $S_1$  represents few popular suppliers whose items take

up 20% of the ratings,  $S_2$  are larger number of suppliers with medium popularity whose items take up around 60% of the ratings, and  $S_3$  are the less popular suppliers whose items get 20% of the ratings.

Figure 4.2 shows the distribution of the number of items owned by the suppliers in each supplier group and also the number of suppliers in each group. In both datasets, the supplier groups with higher popularity own larger number of items compared to the groups whose suppliers do not have high popularity.

## 4.4 Evaluation Metrics

A multi-stakeholder evaluation entails multiple metrics representing the perspective of different stakeholders on system performance. In this dissertation, in order to evaluate different recommendation algorithms with respect to their performance in terms of handling popularity bias, I use several well-established metrics in the literature. I also introduce several other metrics that are able to better differentiate the performance of different algorithms when they are assessed from the perspective of multiple stakeholders. In particular, I propose metrics that are sensitive to the performance across different groups of stakeholders: suppliers in different popularity categories, items in different popularity categories, and users with different levels of interests in item popularity.

# 4.4.0.1 Overall

To capture overall long-tail performance, a variety of metrics have been used in prior research. I compute the following metrics:

- Aggregate Diversity: The ratio of unique recommended items across all users which I defined in Chapter 2.
- Long-tail Coverage (LC): The concept of long-tail coverage is used in many prior works in popularity bias [4,12] and it measures the ratio of long-tail items (in our case it is *M* ∪ *T*



**Figure 4.3:** List of two Recommendations given to three users. Dark circles are popular items and light ones are long-tail items. There are ten items overall in the catalog 6 of which are long-tail and the other 4 are popular. Agg- $Div = \frac{4}{10} = 0.4$ ,  $APL = \frac{1}{2} = 0.5$  and  $LC = \frac{1}{6} = 0.166$ .

since I have divided the long-tail into two separate groups) that appear in the recommendation lists of different users. Effectively, this is *Agg-Div* applied only to the  $M \cup T$  portion of the catalog.

$$LC = \frac{\left|\bigcup_{u \in U} (\ell_u \cap (M \cup T))\right|}{|M \cup T|}$$
(4.2)

*I* be the set of all items in the catalog and *U* is the set of all users.

• **Gini Index:** Measures the inequality across the frequency distribution of the recommended items which was also defined in Chapter 2.

Another metric for measuring long-tail performance of a given algorithm which I introduce in this dissertation is *Average Percentage of Long-tail Items* (*APL*) which I define as follows:

$$APL = \frac{1}{|U|} \sum_{u \in U} \frac{|\{i, i \in (\ell_u \cap (M \cup T))\}|}{|\ell_u|}$$
(4.3)

This metric tells us what percentage of items in users' recommendation lists belongs to the long-tail of the item distribution  $(M \cup T)$ . This is purely a user-centric metric as it does not look at the overall percentage of long-tail items but rather only in a user-level. With that said, it is possible for an algorithm to have high *APL* meaning every user has received a high percentage of long-tail items but, overall, it may not have covered many items from the long-tail when it is looked at across all users. Therefore, *LC* and *APL*, together, can give a better picture of how an

algorithm is performing in terms of long-tail promotion. Look at Figure 4.3 for example. There are ten items in the catalog  $I_1...I_{10}$ . Popular items are shown by dark circles and less popular (long-tail) one shows by light circles. There are six long-tail items and four popular items in the catalog. Three users in this figure have each received two items as the recommendations. We can see that, overall, four unique items are recommended across all users so the aggregate-diversity will be 0.4. Each user has received 1 long-tail item out of a list with size 2 so the *APL* will be  $\frac{1}{2} = 0.5$ . However, as we can see, all users have received the same long-tail item ( $I_4$ ) so, overall, only one long-tail item is recommended and *LC* will be  $\frac{1}{6} = 0.166$ . The *APL* of 0.5 indicates 50% of the recommendations given to the users are long-tail items. This might create the sense that the recommendation algorithm has done a good job in terms of covering long-tail items. However, looking at *LC* we can see that out of 6 long-tail items only 1 has been recommended to different users. So, depending on the purpose of the recommender system, each of the *APL* and *LC* metrics could show a different side of the recommendation performance and, therefore, both can be measured to capture a better picture of the recommendation performance in terms of long-tail.

#### 4.4.0.2 Supplier Groups

Similar to [111], I operationalize the concept of fairness for the suppliers (i.e. artists and directors in Last.fm and MovieLens datasets, respectively) using their popularity. In [111], the authors grouped the artists in Spotify data into 10 equal-sized bins. As I noted in Section 4.3, I group the suppliers into three different bins based on their position in the popularity spectrum  $S = \{S_1, S_2, S_3\}$  with the  $S_1$  being the most popular suppliers followed by  $S_2$  and  $S_3$  that have lower popularity.

Following [111], I use *Equality of Attention Supplier Fairness* (ESF), which operationalizes the idea that a supplier's chance of being recommended should be independent of the popularity bin to which they belong. The *ESF* of a list of recommendations (L) given to all users from the suppliers' perspective is defined as:

$$ESF = \sum_{i=1}^{|S|} \sqrt{\sum_{j \in L} \mathbb{1}(A(j) \in S_i)}$$
(4.4)

where  $S_i$  is the list of suppliers belonging to the popularity bin *i*, and A(j) is a mapping function that returns the supplier of item *j*. |S| is the number of supplier groups which in this dissertation it is 3.

*ESF* rewards sets that are diverse in terms of the different supplier bins represented, thus providing a fairer representation of different supplier bins. Given the nature of the function, there is more benefit to selecting suppliers from a bin not yet having one of its suppliers already chosen. When a supplier from a particular bin is represented in the recommendations, other suppliers from the same bin will be penalized due to the concavity of the square root function (e.g.  $\sqrt{1}+\sqrt{1} > \sqrt{2}$ ).

The *ESF* measure does not, however, consider the inherent popularity of suppliers in different bins. I introduce an alternative metric that does so: *Supplier Popularity Deviation* (SPD).

For any supplier group s, SPD(s) = q(s) - p(s) where q(s) is the ratio of recommendations that come from items of supplier group s (i.e.  $q(s) = \frac{\sum_{u \in U} \sum_{j \in \ell_u} \mathbb{1}(A(j) \in s)}{n \times |U|}$ ), and p(s) is the ratio of ratings that come from items of supplier group s (i.e.  $p(s) = \frac{\sum_{u \in U} \sum_{j \in \rho_u} \mathbb{1}(A(j) \in s)}{\sum_{u \in U} |\rho_u|}$ ). The average *SPD* across different groups of suppliers can be calculated as:

$$SPD = \frac{\sum_{i=1}^{|S|} |SPD(S_i)|}{|S|}$$
(4.5)

Lower values for *SPD* indicate a better match between the distribution of items from different suppliers in rating data and in the recommendations. The inverse 1 - SPD can be considered a type of proportional fairness metric since it measures how the items from different supplier groups are exposed to different users *proportional* to their popularity in rating data.

#### 4.4.0.3 Item Groups

We can use the same formalization as *SPD* to define a metric *Item Popularity Deviation* (IPD) that looks at groups of items across the popularity distribution. For any item group c, IPD(c) =

q(c) - p(c) where q(c) is the ratio of recommendations that come from item group c (i.e.  $q(c) = \frac{\sum_{u \in U} \sum_{j \in \ell_u} \mathbb{1}(j \in c)}{n \times |U|}$ ). p(c) is the ratio of ratings that come from item group c (i.e.  $p(c) = \frac{\sum_{u \in U} \sum_{j \in \rho_u} \mathbb{1}(j \in c)}{\sum_{u \in U} |\rho_u|}$ ). The average *IPD* across different groups of items can be measured as:

$$IPD = \frac{\sum_{c \in C} |IPD(c)|}{|C|},\tag{4.6}$$

## 4.4.0.4 User Groups

Finally, based on our finding above that users with different interests in popular items may receive results with different degrees of calibration, we can examine the popularity calibration of recommendation results across groups of users with different popularity propensity. This metric is *User Popularity Deviation* (UPD). For any user group *g*,  $UPD(g) = \frac{\sum_{u \in g} \mathfrak{J}(P(\rho_u), Q(\ell_u))}{|g|}$ . The average *UPD* across different groups of users is:

$$UPD = \frac{\sum_{g \in G} UPD(g)}{|G|}$$
(4.7)

*UPD* can be also seen as the average popularity miscalibration of the recommendations from the perspective of users in different groups.

For all three metrics *IPD*, *SPD* and *UPD*, lower values are desirable.

Table 4.2 summarizes the evaluation metrics I described in this chapter. I will use these in the following chapters.

**Table 4.2:** The summary of the evaluation metrics used in this dissertation, their abbreviations and their purpose.

Metric Name	Abbreviation	Purpose	
		The ratio of unique	
Aggregate Diversity	Agg-Div	recommended items compared to all	
		the available items.	
		The average percentage of	
Average Percentage of Long-tail Items	APL	long-tail items in individual	
		recommendation lists.	
		The percentage of long-tail	
Long-tail Coverage	LC	items that are covered by	
		the recommender system.	
		Measures how evenly	
Gini Index	Gini	the recommendations are distributed	
		across all items.	
		Measures the fairness	
Equality of Attention Supplier Fairpose	ESE	of exposure for different supplier	
Equality of Attention Supplier Pairless	LOI	groups based on the equality of	
		appearance for different supplier groups.	
		Measures the deviation of	
Supplier Popularity Deviation	SPD	exposure for items from different	
Supplier ropularity Deviation	51 D	supplier groups compared to their	
		original popularity in data.	
		Measures the average amount	
	UPD	of deviation the algorithms impose on	
Users Popularity Deviation		users in different groups	
		compared to what their	
		original interest in popularity was.	
		Measures the deviation of	
Items Popularity Deviation	IPD	exposure for items from different	
icino i opularity Deviation	пр	item groups compared to their original	
		popularity in data.	

# Chapter 5

## **Popularity Bias in Recommendation**

Skew in wealth distribution is well-known: less than 1% of the human population has more wealth than more than 40% of the world combined <sup>1</sup>; in recommender systems a similar problem exists: a small number of popular items appear frequently in user profiles and a much larger number of niche items appear rarely. In many consumer taste domains, where recommender systems are commonly deployed, there is a highly-skewed distribution of user interest in items. A music catalog might contain artists whose songs have been played millions of times (Beyoncé, Ed Sheeran) and others whose songs have a much smaller audience (Iranian musician Kayhan Kalhor, for example). Recommender systems often recommend popular items at a much higher frequency than even their high popularity would call for, amplifying the bias already present in the data.

Recommending serendipitous items from the long-tail (items with low popularity) are generally considered to be valuable to the users [22, 149], as these are items that users are less likely to know about. Brynjolfsson and his colleagues showed that 30-40% of Amazon book sales are represented by titles that would not normally be found in brick-and-mortar stores at the time of their writing [36]. Access to long-tail items is a strong driver for e-commerce growth: the consumer surplus created by providing access to these less-known book titles is estimated at more than seven to ten times the value consumers receive from access to lower prices online.

Long-tail items are also important for generating a fuller understanding of users' prefer-

<sup>&</sup>lt;sup>1</sup> https://inequality.org/facts/global-inequality/

ences. Systems that use active learning to explore each user's profile will typically need to present more long-tail items because these are the ones that the user is less likely to have rated, and where user's preferences are more likely to be diverse [118,133].

Finally, long-tail recommendation can also be understood as a social good. A market that suffers from popularity bias will lack opportunities to discover more obscure products and will be, by definition, dominated by a few large brands or well-known artists [47]. Such a market will be more homogeneous and offer fewer opportunities for innovation and creativity.

As Anderson discusses in his book [22], in an era without the constraints of physical shelf space and other bottlenecks of distribution, narrowly targeted products and services can be as economically attractive as mainstream fare. When you can dramatically lower the costs of connecting supply and demand, it changes not just the numbers, but the entire nature of the market. This is not just a quantitative change, but a qualitative one, too. Bringing niches within reach reveals latent demand for noncommercial content. Then, as demand shifts towards the niches, the economics of providing them improve further, and so on, creating a positive feedback loop that will transform entire industries – and the culture. "We are turning from a mass market into a niche nation, defined now not by our geography but by our interests" Anderson says in his book.

The constraints of physical shelf space was particularly a barrier to the idea of long-tail and niche market in the past [33] as it was not economically effective for the businesses to store a large number of products some of which had a limited number of customers interested in them. In the era of digitalization, however, the physical storage of the goods is no longer a necessity and hence there is more opportunity for the niche market. For instance, movie and music platforms such as Netflix, Hulu, Spotify, and Pandora are connecting millions of consumers to a large number of movies and songs completely on a digital platform. In such a large marketplace, the popular items are still more attractive and getting more attention but the niche products also can reach a specific range of audience. That being said, the recommendations generated on such platforms are often still biased towards hits and popular items and a mechanism for alleviating this bias is needed to give a fair exposure to different items from different popularity levels in the popularity spectrum.


Figure 5.1: The Long-tail Distribution of Rating data

The popularity bias in recommender systems can originate from different sources: the underlying data for user interactions and the inherent bias of the recommendation algorithms.

# 5.1 Bias in Data

Rating data in many recommendation domains is generally skewed towards more popular items-there are a few popular items which take up the majority of rating interactions while the majority of the items receive small attention from the users. For example, few songs and movies are internationally popular, few people on social media have millions of followers, a small percentage of research papers have thousands of citations, and few tourism destinations attract millions of people.

Figure 5.1 shows the percentage of users rated different items in MovieLens and Last.fm datasets: the popularity of each item. Items are ranked from the most popular to the least with the most popular item being on the far left. The curve has a long-tail shape [23, 47] indicating few popular items are taking up the majority of the ratings while many other items on the far right of the curve have not received much attention. This type of distribution can be found in many other domains such as e-commerce where a few products are best-sellers, online dating where a few profiles attract the majority of the attention, and social networking sites where a few users have millions of followers.

The bias in rating data could be due to two different reasons:

- External Bias: Some items and products are inherently more popular than others even outside of the recommender systems in the real world. For instance, even before music streaming services emerged, there were always a small number of artists that were nationally or internationally popular such as *Shakira, Jennifer Lopez*, or *Enrique Iglesias*. As a result of this external bias (or tendency) towards popular artists, users also often listen to those artists more on streaming services no matter if they are recommended to them or not, and hence they get more user interactions. Another external motive for some items being more popular could be due to social influence [94, 100], word of mouth [63], and other external factors that impact the popularity of an item and the users' intention to buy those products.
- Feedback Loop: As we will see later in this chapter, recommendation algorithms have a higher tendency towards recommending popular items. That means, the popular items have a higher chance to be recommended to the users and hence garnering a larger number of interactions from the users. Therefore, when these interactions are logged and stored in the system as the new data, the popularity of those items increases since they get more and more interactions over time [86, 109].

## 5.2 Algorithmic Bias

Due to this imbalance property of the rating data, often algorithms inherit this bias and, in many cases, intensify it by over-recommending the popular items and, therefore, giving them a higher opportunity of being rated by more users [83, 123, 154]: *the rich get richer and the poor get poorer*.

Figure 5.2 shows the percentage of users who have rated an item on the x-axis and the percentage of users who received that item in their recommendation lists using four different recommendation algorithms *RankALS* [161], *Biased Matrix Factorization (Biased-MF)* [99], *User-based* 

*Collaborative Filtering (User-CF)* [17], and *Item-based Collaborative Filtering (Item-CF)* [144]. In other words, the plot aims to show the correlation between the popularity of an item in the rating data versus how often it is recommended to different users using different algorithms. It is clear that in all four algorithms, many items are either never recommended or just rarely recommended.

Among the four algorithms, *Item-CF* and *User-CF* show the strongest evidence that popular items are recommended much more frequently than the others. In fact, they are recommended to a much greater degree than even what they initial popularity warrants. For instance, the popularity of some items have been amplified from roughly 0.4 to 0.7 indicating a 75% increase. Both *Item-CF* and *User-CF* are over-promoting popular items (items on the right size of the x-axis) while significantly hurting other items by not recommending proportionate to what their popularity in data warrants. In fact, the vast majority of the items on the left are never recommended indicating an extreme bias of the algorithms towards popular items and against less popular ones. On *RankALS* the positive correlation between popularity in rating data and how frequently an item is recommended can still be observed but it is nowhere as extreme as *User-CF* and *Item-CF*. *Biased-MF* does not show a positive correlation between popularity in data and in recommendations although some items are still over-recommended (have much higher popularity in recommendations versus what they had in rating data). However, this over-recommendation is not concentrated on only popular items and some items from lower popularity values are also over-recommended.

Figure 5.3 shows the same information on Last.fm data. The magnitude of the popularity bias of different recommendation algorithms seems to be lower on Last.fm dataset. The reason is the original rating data in Last.fm is less biased towards popular items as we can see in y-axis of Figure 5.1 where the most popular item in MovieLens has a popularity close to 0.45 where in Last.fm this value is less than 0.14.

## 5.3 Multi-stakeholder Impact of Popularity Bias

As I discussed in Chapter 3, recommender systems are multi-stakeholder environments where, in addition to the users, some other stakeholders such as the supplier of the items also



Figure 5.2: Item popularity versus recommendation popularity (MovieLens)



Figure 5.3: Item popularity versus recommendation popularity (Last.fm)

benefit from the recommendation of their items and gaining a larger audience. On top of that, the recommender system designer might also have certain goals such as recommending a particular group of items more frequently in certain times depending on their business goals such as sales promotions, items that need to be sold before a certain time etc.

Algorithmic popularity bias can negatively impact different parts of a recommender system platform. In this section I intend to demonstrate how this bias can impact different groups of items, users, and the suppliers of the items.

## 5.3.1 Impact on Item Groups

Figure 5.4 contrasts item popularity and recommendation popularity for four well-known recommendation algorithms *RankALS*, *Biased Matrix Factorization* (*Biased-MF*), *User-based Collaborative Filtering* (*User-CF*), and *Item-based Collaborative Filtering* (*Item-CF*). The black curve shows the cumulative frequency of ratings for different items at a given rank in the MovieLens 1M [73] data set. The x-axis indicates the rank of each item when sorted from most popular to least popular. The y-axis indicates the cumulative prevalence of that item in the data. As we can see, a few highly ranked items dominate the entire rating history. For instance, only 111 items (less than %3 of the items) take up more than %20 of the ratings. A similar type of distribution can be seen in many other domains. These few popular items are referred to as short-head (*H*) in the literature, followed by the middle-tail (*M*), which are items that are not blockbusters but have significant popularity, and tail items (*T*), which are more obscure and appeal to niche groups of users [47].<sup>2</sup>

The blue curves in each figure show popularity bias at work across the four algorithms. In *Item-CF*, no item beyond rank 111 is recommended. The head of the distribution constituting less than 3% of the items are actually taking up 100% of the recommendations – a "lion's share" indeed. In *User-CF* this number is 99%. The other algorithms are only slightly improved in this regard, where the head items take up more than 64% and 74% of the recommendations in *RankALS* 

<sup>&</sup>lt;sup>2</sup> Tail items may also be new items that have yet to find their audience and will eventually become popular. In this way, popularity bias is related to the well-known item cold-start problem in recommendation.

and Biased-MF, respectively.

#### 5.3.2 Impact on User Groups

We can also take a consumer- / user-centered view of popularity bias. Not every user is equally interested in popular items. In cinema, for instance, some might be interested in movies from Yasujiro Ozu, Abbas Kiarostami, or John Cassavetes, and others may enjoy more mainstream directors such as James Cameron or Steven Spielberg. Figure 5.5 shows the ratio of rated items for three item categories H, M, and T in the profiles of different users in the MovieLens 1M and Last.fm datasets. Users are sorted from the highest interest towards popular items to the least and divided into three equal-sized bins  $G_{1..3}$  from most popularity-focused to least, as I described in Chapter 4. The y-axis shows the proportion of each user's profile devoted to different popularity categories. The narrow blue band shows the proportion of each users profile that consists of popular items, and its smooth decrease reflects the way the users are ranked. Note, however, that all groups have rated many items from the middle (green) and tail (red) parts of the distribution, and this makes sense: there are only a small number of really popular movies and even the most blockbuster-focused viewer will eventually run out of them.

Figure 5.6 shows the same information as Figure 5.5 but all ratios are calculated as an average value across users in each user group. The differences between different user groups in this figure are more visible. For example, looking at the head ratio (blue color), we can clearly see that users in group  $G_1$  has the highest ratio of head items followed by  $G_2$  and  $G_3$ .

The plots in Figure 5.7 are parallel to Figure 5.5, with the users ordered by their popularity interest, but now the y-axis shows the proportion of recommended items using different algorithms from the different item popularity categories. The difference with the original user profiles in rating data especially in the case of Item - CF and User - CF is stark. Where the users' profiles are rich in diverse popularity categories, the generated recommendations are nowhere close that to what the user has shown interest at. In fact, in Item - CF almost 100% of the recommendations are from the head category, even for the users with the most niche-oriented profiles. Tail items do

not appear at all. I demonstrated here that popularity bias in the algorithm is not just a problem from a global, system, perspective. It is also a problem from the user perspective: users are not getting recommendations that reflect the diversity of their profiles and the users with the most niche tastes ( $G_3$ ) are the most poorly served.

Similar to Figure 5.6, Figure 5.8 shows the ratio of different item groups in the profiles of users in different user groups as an average value. Here, we can clearly see the differences between algorithms in term of how the have performed for different user groups. **Item-CF** clearly consists of only head items for all user groups.

#### 5.3.3 Impact on Supplier Groups

As noted above, multi-stakeholder analysis in recommendation also includes providers or as termed here *suppliers*, "those entities that supply or otherwise stand behind the recommended objects" [1]. We can think of many different kinds of contributors standing behind a particular movie. As I mentioned in Chapter 4, I consider the director of each movie and the artist for each song as the suppliers in MovieLens and Last.fm respectively.

Figure 5.9 is similar to Figure 5.4 but shows the rank of different directors by popularity and the corresponding recommendation results from the *Item-CF* algorithm. A similar pattern of popularity bias is evident. The recommendations have amplified the popularity of the popular suppliers (the ones on the extreme left) while suppressing the less popular ones dramatically. Strikingly, movies from just 3 directors (less than 0.4% of the suppliers here) take up 50% of recommendations produced, while the tail of the distribution is seeing essentially zero.

What this analysis demonstrates is that popularity bias is a multi-stakeholder phenomenon, that entities on different sides of the recommender system are impacted by it (sometimes quite severely) and that any evaluation of it must capture the perspective of different stakeholders.

In next chapter, I will propose several recommendation algorithms for mitigating the popularity bias and evaluate their performance using the multi-stakeholder paradigm to see how they perform from the perspective of different stakeholders.



Figure 5.4: Popularity bias amplification (Item Groups)



Figure 5.5: Users' Propensity towards item popularity



Figure 5.6: User Groups' Propensity towards item popularity



Figure 5.7: Users' centric view of popularity bias





Figure 5.8: Users' centric view of popularity bias (Groups)



Figure 5.9: Suppliers' centric view of popularity bias

# Chapter 6

## **Tackling Popularity Bias**

As I mentioned in Chapter 2 and 3, traditionally recommendation algorithms have been developed in a way that could maximize the accuracy of the delivered recommendations to the users. However, this over-emphasis on being accurate could hurt other important factors of a successful recommendation such as diversity, serendipity and novelty [46, 66, 164] and impacts across system stakeholders [1].

One of the problems that is beyond being accurate is the popularity bias phenomena I discussed in Chapter 5. We saw that this bias can negatively impact different stakeholders in a given recommender system and hence the algorithms for mitigating such bias need to be evaluated from the perspective of different stakeholders.

The solutions for tackling popularity bias in the literature can be categorized into two groups. A third option, preprocessing, is generally not useful for popularity bias mitigation because undersampling the popular items greatly increases the sparsity of the data.

- **Model-based:** In this group of solutions, the rating prediction step is modified, so that the popularity of items are taken into account in the rating prediction [157, 166].
- Re-ranking: Most of the solutions for tackling popularity bias fall into this category [12, 13]. A re-ranking algorithm takes an output recommendation list extracts a re-ordered list with improved long-tail properties.



Figure 6.1: The Schema for the model-based popularity mitigation

## 6.1 Model-based solutions

The general schema for the model-based approach to tackle popularity bias is shown in Figure 6.1. This is similar to Figure 2.1 in Chapter 2 with one distinction: an extra component which I call *Item Popularity Consideration* is incorporated in the recommendation algorithm before the final recommendation list is generated to the user.

This consideration of item popularity can be incorporated in different ways such as in the rating prediction step whether it is in neighborhood-based models [53] or in matrix factorization techniques based on rating prediction [99] or in the calculation of the ranking order of preferred items from the user's perspective in learning-to-rank approaches [91,155]. Note that the calculation of the ranking order of preferred items in learning-to-rank models is still within the recommendation algorithm.

Controlling popularity bias can be incorporated into this step before the recommendation list is generated to the user. For example, Vargas et al. in [166] propose a creative way to reduce the popularity bias and to increase the sales diversity by reversing the role of items and users. Generally, as discussed in Chapter 2, the goal of the recommender system is to recommend a list of items to the users. However, due to popularity bias, some items are not recommended to anyone or are not recommended enough. By reversing the role of items and users, the goal would be, for each item, find a set of users who might be interested in that item. This way, less popular items are no longer ignored and they will reach to a set of users which leads to reducing the bias on popular items. In another work, Adamopoulos and Tuzhilin in [8] propose a modification to the standard neighborhood-based recommender systems to handle the over-specialization and concentration bias of the recommendations. The most important step in the neighborhood-based algorithms is the selection of the most similar users to the target user in order to use their rating pattern to find interesting and relevant items to be recommended to the target user. However, selecting the most similar users often lead to dictating the preferences of some users (those that appear in many neighborhood lists of other users) on other users. Therefore, authors in this paper use a probabilistic approach to select the neighbors for each user instead of selecting the most similar neighbors. Authors assign an initial weight to each candidate neighbor according to its distance from the target user and then the candidates are sampled, without replacement, proportionally to their assigned weights. By having a more diverse set of neighbors, as opposed to top most similar ones, the recommendations will be less prone to concentrate only on popular items.

Incorporating the popularity bias control in the rating prediction step in factorization models often is done by modifying the objective function used to minimize the rating prediction error. For instance, Kamishima et al. in [90] introduced the notion of neutrality in recommendation algorithms with respect to a certain feature or characteristic. In particular, the authors explore this concept in correcting the popularity bias in recommendations. Their method is a variant of the probabilistic matrix factorization model that adopts a constraint term for enhancing neutrality. The neutrality of a user towards item popularity (or any other aspect) is calculated using users' profiles based on how tolerant they are with respect to popular items. In another work, [50] Coba et al. propose a regularization-based technique for counteracting popularity bias in a matrix factorization-based technique by adding an extra constraint to the general matrix factorization objective function that minimizes the distance between the popularity of a given item and the average popularity of the items in the users' profile. Therefore we can say this method is a personalized approach for mitigating popularity bias.

The model-based techniques can be also used for improving the diversity of the recommen-

dations. For instance, Wasilewski and Hurley in [167] proposed a modification of a ranking-based recommendation algorithm (*RankALS*) by adding a regularization term to its objective function that minimizes the similarity (or increases the diversity) of the items within each list. Generally speaking, many other types of recommendation properties such as novelty, diversity and even fairness can be incorporated into the underlying recommendation algorithm as a model-based technique [102, 169, 179].

#### 6.1.1 Proposed Model-based Technique Using Regularization

In this section, I present a flexible framework for a model-based popularity mitigation built on the top of a learning-to-rank algorithm. When applied to long-tail items, this approach enables the system designer to tune the application to achieve a particular trade-off between ranking accuracy and the inclusion of long-tail items. However, the framework is sufficiently general that it can be used to control recommendation bias for or against any group of items.

#### *Learning to Rank*

I focus on matrix factorization approaches to recommendation in which the training phase involves learning a low rank  $k \times |U|$  latent user matrix P and a low-rank  $k \times |I|$  latent item matrix Q, such that the estimated rating  $\hat{r}_{ui}$  can be expressed as  $\hat{r}_{ui} = p_u^T q_i$  where  $p_u^T$  is the  $u^{th}$  row of P,  $q_i^T$  is the  $i^{th}$  row of Q, and k is the chosen dimension of the latent space. P and Q are learned through the minimization of an accuracy-based objective [98].

In this method, I focus on the learning-to-rank problem and I add a regularization term to its objective function that adjusts the scoring and gives a higher chance to less popular items to be included in the recommendation lists. Generally, the objective function of any factorization-based technique is based on minimizing a loss function that measures the error of a recommendation algorithm. In rating prediction models, this error is the squared difference between the original rating a user has given to an item and the predicted rating returned by the algorithm  $((r_{ui} - \hat{r}_{uj})^2)$ . In a learning-to-rank method, however, the error is the difference between the original rating and predicted rating between a pair of items. In other words, in learning-to-rank, what matters is the correct order of any pair of items in the user's profile and in the recommendations  $(((\hat{r}_{ui} - \hat{r}_{uj}) - (r_{ui} - r_{uj}))^2)$ . That is, if item *i* is above item *j* in a user's profile based the rating the user has given to these two items, in the recommendations this order should be also the same.

For this work, I use the objective function proposed in [79] for a learning-to-rank recommendation system and further developed in [161]:<sup>1</sup>

$$\sum_{u \in U} \sum_{i \in I} c_{ui} \sum_{j \in I} s_j [(\hat{r}_{ui} - \hat{r}_{uj}) - (r_{ui} - r_{uj})]^2 + \beta (\|Q\|^2 + \|P\|^2)$$
(6.1)

The role of  $c_{ui}$  is to select user-item pairs corresponding to positive feedback from all possible pairs. I consider the implicit feedback case in which  $c_{ui} = 0$  if  $r_{ui} = 0$ , and 1 otherwise.  $s_j$  is an importance weighting for item j. In this work, because I am trying to reduce the influence of popular items, I use uniform importance weighting  $s_j = 1$ , for all j.  $\beta$  is the standard regularisation parameter of norm-based regularization to avoid over-fitting.

#### Fairness-aware Regularization

Regularization is one of the main techniques to control the behavior of an objective function. Generally speaking, it is mainly used to avoid overfitting in machine learning models [20] by adding an additional penalty term in the error function. This additional term controls the excessively fluctuating function such that the coefficients don't take extreme values. For example, the terms  $\beta(||Q||^2 + ||P||^2)$  in equation 6.1 is a form of regularization that controls the magnitude of the entries in *P* and *P* to discourage the complexity of the model and avoid overfitting.

In my work, I use regularization as a tool to add extra constraint to the original objective function such that, in addition to ranking accuracy, the fairness of representation between popular and less popular items is taken into account in the optimization process. This work was inspired by the work of Wasilewski and Hurley [167] in which regularization is used to enhance recommendation diversity in a learning-to-rank setting, specifically using the RankALS algorithm [161]. The authors calculate a pair-wise dissimilarity matrix *D* for all pairs of items, demonstrate that

<sup>&</sup>lt;sup>1</sup> A number of such objectives have been proposed in the literature and the regularization method I propose here could be incorporated with any such pair-wise objective.

intra-list diversity (ILD) can be represented in terms of this matrix, and then experiment with different ways that increased ILD can be made into an optimization objective.

Following a similar approach, I explore the use of regularization to control the popularity bias of a recommender system. I start with an optimization objective of the form:

$$\min_{P,Q} acc(P,Q) + \lambda reg(P,Q)$$
(6.2)

where acc(.) is the accuracy objective similar to Equation 6.1, reg(.) is our regularization term, and  $\lambda$  is a coefficient for controlling the effect of regularizer.

Our goal therefore is to identify a regularization component of the objective that will be minimized when the distribution of recommendations is fair in terms of popular versus less popular (long-tail) items. I define a *fair recommendation list* as one that achieves a 50/50 balance between long-tail ( $M \cup T$ ) and short-head (H) items which is referred to as equality of attention (or opportunity) in the literature [71].

Wasileski and Hurley start from a dissimilarity matrix D, which contains all pair-wise dissimilarities between items. The regularizer is computed from D in such a way that it pushes the optimization solution towards recommendation lists that balance ranking accuracy with intra-list diversity (ILD).

In my case, I do not aim for diversifying the recommendation lists in terms of content (movie genre, music genre etc) but rather based on different item groups according to their popularity. This framework is a binary method meaning it needs two groups of items so it can balances the recommendations between them. I define the first group as popular items (short-head) H. For the second group which represents long-tail items, I combine the M and T groups defined in Chapter 4 and consider them as long-tail items  $\gamma$  ( $\gamma = M \cup T$ ). I also define a co-membership matrix D, over these sets such that, for any pair of items i and j, d(i, j) = 1 if i and j are in the same set and 0 otherwise.

Our equivalent for intra-list distance is a measure of the lack of fairness for the two sets of

items in a given recommendation list to user u,  $\ell_u$ . I define intra-list binary unfairness (ILBU) as the average value of d(i, j) across all pairs of items i, j.

$$ILBU(\ell_u) = \frac{1}{n(n-1)} \sum_{i,j \in \ell_u} d(i,j)$$
(6.3)

where *n* is the number of items in the recommendation list. The fairest list is one that contains equal number of items from each set, which can be easily seen: if we start with a list containing only *H* items and replace an item from *H* with one from  $\gamma$ , the highest number of non-zero pairs occur when there is an equal number of items from *H* and  $\gamma$ . If every item in the set are from the same group (either *H* or  $\gamma$ ), *ILBU* will be high since all d(i,j) pairs are 1 which means higher unfairness.

Note that I do not in my definition of fairness require that each item have an equal chance of being recommended. There are many fewer short-head items and we are, essentially, allocating half of the recommendation list to them.<sup>2</sup> The 50/50 heuristic is a starting point for exploring the effectiveness of our regularization-based technique.

As the ILBU value has the same form as the ILD measure used in [167], I can take advantage of their experience in deriving various regularization terms from it. I chose the LapDQ regularizer, which performed well in Wasilewski and Hurley's experiments. LapDQ is based on the Laplacian  $L_D$  of the D matrix and has the form  $tr(Q^T L_D Q)$ , where tr is the trace of the matrix. This version of the regularizer only influences the item factors and therefore the p step of the RankALS algorithm, which computes and modifies the user factors, is unchanged.

The details of the derivation of the regularizer is not the focus of this dissertation and readers are referred to [167] for further information. One way to understand the function of LapDQis to note that our D matrix is block-structured, and has the same structure as the adjacency matrix for a graph consisting of two separate components of fully interconnected items. For such a graph, the partition is represented by the eigenvector of the Laplacian corresponding to the largest

<sup>&</sup>lt;sup>2</sup> Note that it would be possible to adjust the balance between long-tail and short-head items by creating a smaller penalty for larger numbers of short-head items. These possibilities I leave for future work.

eigenvalue. In our case, we already know what this partition is – it is the division of the nodes into short-head and long-tail items. The trace of  $Q^T L_D Q$  is the product of each latent factor with the Laplacian of the membership matrix. If the item factors most resemble the partitioning eigenvector – that is, they contain elements from both partitions in opposite signs, then this vector product, and therefore the matrix trace, is minimized. The regularizer therefore pushes each item factor in Q to span both partitions as evenly as possible.

#### Evaluation

A multi-stakeholder evaluation entails multiple metrics representing the perspective of different stakeholders on system performance. I examine the behavior of this proposed algorithm relative to different metrics of long-tail performance that I described in Chapter 4. Conventional metrics are those that look at a system's overall recommendation of long-tail items. I show these results, but in addition, I also use metrics that are sensitive to the performance across different groups of stakeholders as I discussed in Chapter 4: suppliers in different popularity categories, items in different popularity categories, and users with different levels of interests in item popularity.

Figure 6.2(a) shows the behavior of our model-based method when we vary the weight of the regularization: the  $\lambda$  parameter. As we can see, the algorithm loses some accuracy (both precision and NDCG) when we increase the  $\lambda$  and give more weight to the fairness criteria. The Gini slightly improves meaning the recommendations are more uniformly distributed compared to the base algorithm with no fairness component.

Other metrics related to long-tail performance and fairness, as can be seen in Figure 6.2(b) have all improved for the largest value of  $\lambda$  but for some metrics such as *LC*, *ESF*, and *SPD* the behavior of the algorithm is not completely monotonic. For instance, up until  $\lambda = 0.4$ , the *LC* actually slightly decreases and then it starts to increase for larger values of  $\lambda$ . However this anomaly is only seen on the MovieLens data. The trend for *ESF* on MovieLens data seems to be unpredictable. However, looking at the scale for the vertical axis, we can see that the changes are not statistically significant(*p*-value > 0.05). *IPD* and *UPD*, on the other hand, show a monotonic



(a) Precision, NDCG, Gini, APL, Aggregate Diversity, and Long-tail coverage for our model-based approach on MovieLens data



(b) Equality of attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popularity Deviation (UPD) and Supplier Popularity Deviation (SPD) of our model-based technique on MovieLens data

**Figure 6.2:** Precision, NDCG, Gini, APL, Aggregate Diversity and Long-tail coverage, Equality of attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popularity Deviation (UPD) and Supplier Popularity Deviation (SPD) of our model-based technique on MovieLens data



(a) Precision, NDCG, Gini, APL, Aggregate Diversity and Long-tail coverage for our model-based approach on Last.fm data



(b) Equality of attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popularity Deviation (UPD) and Supplier Popularity Deviation (SPD) of our model-based technique on Last.fm data

**Figure 6.3:** Precision, NDCG, Gini, APL, Aggregate Diversity and Long-tail coverage, Equality of attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popularity Deviation (UPD) and Supplier Popularity Deviation (SPD) of our model-based technique on Last.fm data

improvement over different values of  $\lambda$ . *SPD* improves up until  $\lambda = 0.4$  and then it starts to go up again for larger values of  $\lambda$ . *APL* on the other hand, which measures a more localized view of long-tail diversity in the individual lists and more in line with what I am optimizing for is improving when we increase the  $\lambda$ . I believe the reason for the behavior of *LC* can be explained by the nature of our objective function. our *ILBU* metric only looks at the individual lists in terms or the ratio of short-head versus long-tail items. Therefore, it is possible for an algorithm to make the individual lists more diverse in terms of item popularity (higher *APL*) but overall, the number of unique long-tail items in the recommendations across all users might decrease (low *LC*) as we also saw in our illustrative example in Figure 4.3 in Chapter 4.

Figure 6.3 shows the results on Last.fm data. On this dataset, unlike MovieLens, the metrics show a more monotonic behavior and we can see, generally, when we increase the  $\lambda$ , the algorithm loses accuracy (both Precision and NDCG) while other metrics all are improving.

What we see in these results is similar to the findings of [167], namely that a regularization term incorporating list-wise properties of interest can be incorporated successfully into a learning-to-rank framework. In Wasilewski and Hurley's work, this was a similarity-based term that enabled them to enhance list-wise diversity; in our case, it is a group-membership-based term that enables us to enhance list-wise coverage of long-tail items.

Note in particular that because this loss function is local to individual recommendation lists, it does not aim to increase catalog coverage in long-tail directly. However, our results show that, for these datasets, the optimization of our local objective also increases overall coverage of less popular items which makes sense since when we increase the long-tail coverage of individual lists, the overall long-tail coverage may also increase.

# 6.2 **Re-ranking Approach**

The second class of approaches for mitigating popularity bias is based on the re-ranking of an original list generated by a given recommendation algorithm. The idea is, for each item in the original list, both the relevance of the item to the user and its popularity are taken into



Figure 6.4: The schema for the re-ranking popularity bias mitigation

account so items with high relevance and also lower popularity could be pushed up to the top of the list [11]. Figure 6.4 shows the general schema for this class of algorithms. As we can see, the recommendation component is not modified but, instead, the recommender system generates a larger list of recommendations  $\ell'$  of size m and the re-ranking algorithm creates a shorter list  $\ell$  of size n (m >> n) that has better long-tail properties.

The advantage of the re-ranking techniques over the model-based ones is that they are not dependent on a particular recommendation algorithm. Unlike the model-based techniques, where popularity mitigation is done within the recommendation algorithm itself, the re-ranking methods do not modify the underlying recommendation algorithm but rather acts as a post-processing technique that could be applied on top of any existing recommendation algorithm.

There are many existing works for mitigating popularity bias that are based on the reranking approach. Miyamoto et al. [113] proposed a post-processing re-ranking technique for reducing frequently recommended items. The algorithm, at first, predicts the ratings a user would give to a list of items. Then, it keeps those items whose predicted ratings is above a certain threshold and, for each of those items, the algorithm calculates how many times the item was recommended to other users. After that, the items are sorted based on this score and the items with the lowest score are put in the final recommendation list and the score for those items increases by one. This process repeats for every user. In another work [13] propose a graph-based approach based on the maximum flow theory for improving the aggregate diversity of the recommendations. In this work, the max-flow value will be equal to the largest possible number of recommendations than can be made from among the available (highly predicted) items, where no user can be recommended more than n items, and no item can be counted more than once, which is precisely the definition of the aggregate diversity. In a similar work which is also based on maximum flow theory [25], Antikacioglu et al. use minimum-cost network flow method to efficiently find recommendation sub-graphs that optimize diversity. Authors in this work define a target distribution of item exposure (i.e. the number of times each item should appear in the recommendations) as a constraint for their objective function. The goal is therefore to minimize the discrepancy of the

recommendation frequency for each item and the target distribution. In addition, Jannach et al. in [83] propose a personalized re-ranking technique that adjusts the relevance scores of the items so it can boost the predicted rating for less popular items to push them to the top of the list. The adjustment of relevance is done by taking into account the propensity of the users towards popular items and also the popularity of the recommended list and tries to minimize the norm-2 of this distance. In this dissertation, I also propose two re-ranking based techniques for controlling popularity bias that have the tolerance of the users towards item popularity incorporated in the algorithm.

#### 6.2.1 Re-ranking Technique 1: Fair Long-tail Promotion:

I present a general and flexible approach for controlling the balance of item exposure in different portions of the item catalog as a post-processing phase for standard recommendation algorithms. Our work is inspired by [142] where authors introduced a novel probabilistic framework called *xQuAD* for Web search result diversification which aims to generate search results that explicitly account for various aspects associated with an under-specified query. One motivating example for the usefulness of such diversification in search results is the ambiguity of certain queries. For instance, if a user types the word *Apple* in the search bar, s/he could be searcher for the fruit, or maybe the company. So, we can say, in this example, that the word apple has two aspects: 1) being a fruit and 2) being a company. Therefore, in order to make sure we are not limiting the choices for the user, the returned result is better to contain web pages related to both of those aspects. By doing so, we give the user the choice in selecting which of the search results s/he intended to find. The algorithm, re-ranks an initial list of search results so it contains a fair proportion of web pages related to each of these two different aspects. It is worth noting that a search query could be related to more than two aspects.

I adapt the xQuAD approach to the popularity bias problem. Here, the concept of diversification is slightly different as I no longer use aspects associated with each item. In my approach, each item is either popular and not. Therefore, I am dealing with a binary categorization and the goal of the re-ranking technique, is to return a list of recommendations that contains a balanced ratio of items belonging to either popular or less popular item categories. By doing that, we are helping the less popular items to have a better representation in the recommendation lists as, initially, they are under-represented.

Given the nature of our proposed approach, it enables the system designer to tune the system to achieve the desired trade-off between accuracy and better coverage of long-tail, less popular items.

## xQuAD

Result diversification has been studied in the context of information retrieval, especially for web search engines, which have a similar goal to find a ranking of documents that together provide a complete coverage of the aspects underlying a query [143]. EXplicit Query Aspect Diversification (xQuAD) [142] explicitly accounts for the various aspects associated with an underspecified query. Items are selected iteratively by estimating how well a given document satisfies an uncovered aspect.

In adapting this approach, I seek to recognize the difference among users in their interest in long-tail items. Uniformly-increasing diversity of items with different popularity levels in the recommendation lists may work poorly for some users. I propose a variant that adds a personalized bonus to the items that belong to the under-represented group (i.e. the long-tail items). The personalization factor is determined based on each user's historical interest in long-tail items.

### Methodology

I build on the *xQuAD* model to control popularity bias in recommendation algorithms. As described in section 6.2, I assume that for a given user *u*, a ranked recommendation list  $\ell'$  with size *m* has already been generated by a base recommendation algorithm. The task of the modified xQuAD method is, therefore, to produce a new re-ranked list  $\ell$  of size *n* that manages popularity bias while still being accurate.

The new list is built iteratively according to the following criterion:

$$P(i|u) + \lambda P(i, u|i \notin \ell) \tag{6.4}$$

where P(i|u) is the likelihood of user  $u \in U$  being interested in item  $i \in I$ , independent of the items on the list so far as, predicted by the base recommender. The second term  $P(i, u|i \notin \ell)$  denotes the likelihood of user u being interested in an item *i* as an item not in the currently generated list  $\ell$ .

Intuitively, the first term incorporates ranking accuracy while the second term promotes diversity between two different categories of items (i.e. short head and long tail). The parameter  $\lambda$  controls how strongly controlling popularity bias is weighted in general. The single item that scores most highly under the equation 6.4 is added to the output list  $\ell$  and the process is repeated until  $\ell$  has achieved the desired length.

To achieve more diverse recommendation containing items from both short-head and longtail items, the marginal likelihood  $P(i, u | i \notin \ell)$  over both item categories long-tail ( $\gamma = M \cup T$ ) and short head (H) is computed by:

$$P(i, u | i \notin \ell) = \sum_{c \in \{H, \gamma\}} P(i, u | c \notin \ell) P(c | u)$$
(6.5)

Following the approach of [142], I assume that the remaining items are independent of the current contents of  $\ell$  and that the items are independent of each other given the short-head and long-tail categories. Under these assumptions, we can compute  $P(i, u | c \notin \ell)$  in Eq.6.5 as

$$P(i, u | c \notin \ell) = P(i | c) P(c \notin \ell) = P(i | c) \prod_{j \in \ell} (1 - P(j | c \in \ell))$$
(6.6)

By substituting equation 6.6 into equation 6.5, we can obtain

$$score = (1 - \lambda)P(i|u) + \lambda \sum_{c \in \{H,\gamma\}} P(c|u)P(i|c) \prod_{j \in \ell} (1 - P(j|c \in \ell))$$

$$(6.7)$$

where P(i|c) is equal to 1 if  $i \in c$  and 0 otherwise.

I measure  $P(j|c \in \ell)$  as the ratio of items  $(j \in \ell)$  that covers category *c*.

The likelihood P(c|u) is the measure of user preference over different item categories. In other words, it measures how much each user is interested in short head items versus long tail items. I calculate this likelihood by the ratio of items in the user profile which belong to category *c*.

In order to select the next item to add to  $\ell$ , I compute a re-ranking score for each item in  $\ell' \setminus \ell$  according to Eq. 6.7. For an item  $i' \in c$ , if  $\ell$  does not cover c, then an additional positive term will be added to the estimated user preference P(i'|u). Therefore, the chance that it will be selected is larger, balancing accuracy and popularity bias.

#### Evaluation

As with other re-ranking techniques, our proposed re-ranking method also needs a base algorithm to generate the initial list of recommendations for post-processing. Since our modelbased algorithm I proposed in section 6.1.1 works on the RankALS algorithm, for comparison purposes, I chose this algorithm as the base and I call it *Base*.

Similar to [92] I set the size of the recommendations generated by the *Base* algorithm to 100 (m = 100), and the size of the final recommendation list is 10 (n = 10).

Figure 6.5(a) shows the behavior of our fairness-aware re-ranking method for different values of  $\lambda$  which controls the weight of the fairness(i.e. long-tail enhancement) compared to accuracy. Similar, to model-based method we saw in previous section, we can see that the re-ranking algorithm also loses accuracy (precision and NDCG) when we increase the  $\lambda$ . The Gini improves when we increase the  $\lambda$  parameter indicating the algorithm achieves a more uniformly distributed recommendations across all users.

The long-tail enhancement and fairness metrics, however, are all improving as we can be seen in Figure 6.5(a) and 6.5(b). Although, almost for all metrics, we see a stepwise behavior meaning the curve goes up / down, stays constant at some points, and then goes up / down again. Looking at the scale of the y-axis, we can see this re-ranking approach performs better than the model-based technique when it comes to long-tail enhancement and fairness. I will compare the performance of all three algorithms proposed in this dissertation later in section 6.2.2.



(a) Precision, NDCG, Gini, APL, Aggregate Diversity and Long-tail coverage for our fairness-aware re-ranking approach on MovieLens data



(b) Equality of attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popularity Deviation (UPD) and Supplier Popularity Deviation (SPD) of our fairness-aware re-ranking approach on MovieLens data

**Figure 6.5:** Precision, NDCG, Gini, APL, Aggregate Diversity and Long-tail coverage, Equality of attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popularity Deviation (UPD) and Supplier Popularity Deviation (SPD) of our fairness-aware re-ranking on MovieLens data



(a) Precision, NDCG, Gini, APL, Aggregate Diversity and Long-tail coverage for our fairness-aware re-ranking approach on Last.fm data



(b) Equality of attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popularity Deviation (UPD) and Supplier Popularity Deviation (SPD) of our fairness-aware re-ranking approach on Last.fm data

**Figure 6.6:** Precision, NDCG, Gini, APL, Aggregate Diversity and Long-tail coverage, Equality of attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popularity Deviation (UPD) and Supplier Popularity Deviation (SPD) of our fairness-aware re-ranking on Last.fm data



Figure 6.7: Popularity calibration example

# 6.2.2 Re-ranking Technique 2: Calibrated Popularity

The second re-ranking method I propose in this dissertation is Calibrated Popularity.

One of the problems of algorithmic popularity bias is the tendency of over-recommending popular items even to those users who do not have much interest in popular items. However as we saw in Chapter 5 there are many users who are interested in less popular items and hence exclusively recommending popular items to these users would hurt their experience in interacting with the recommender system. Therefore, I propose an algorithmic approach based on the novel notion of *popularity calibration* that provides a comprehensive solution to the problem of popularity bias by addressing the needs of multiple stakeholders. A recommendation list is calibrated based on popularity when the range of items it covers matches the user's profile in terms of item popularity. For example, if a user has 20% H items, 40% M items and 40% T items in her profile, we call the recommendations to be calibrated when the ratio of each item category in the recommended list is consistent with the aforementioned ratios. I show that my approach, while yielding more calibration recommendations for users, also results in fairer recommendations from the suppliers' perspective.

#### 6.2.2.1 Motivating Example

Figure 6.7 shows a user whose name is Ross<sup>3</sup>. Out of all the items in a movie recommender's catalog, Ross has interacted with (or liked) four items *I*<sub>2</sub>, *I*<sub>4</sub>, *I*<sub>5</sub>, and *I*<sub>7</sub>. Suppose dark circles are popular items (H) and white ones are less popular items (either M or T). That means his profile consists of an equal ratio of popular versus non-popular items (50% each), and I assume that the contents of his profile represents the scope of his interest in movies. In addition, each item is provided by one of the suppliers A or B. A (red color) who is a popular supplier (based on the average popularity of their items) owns  $I_2$ ,  $I_3$ ,  $I_5$ ,  $I_6$ ,  $I_7$ ,  $I_8$ ,  $I_9$  and  $I_{10}$ . B who is a less popular supplier owns  $I_1$  and  $I_4$ . Note that a popular supplier could also own some less popular items (A owns  $I_8$ ). Since Ross is equally interested in popular and less popular items based on his profile, a well-calibrated recommender should seek to deliver the same ratio in his recommendations, reflecting the diversity of interests that he has shared with the system. Assuming the size of the recommendation set is 2 (for illustration purposes), he should get one popular and one less popular recommendation. A recommendation algorithm influenced by popularity bias would be more likely to generate a recommendation list containing only popular items: "Non-Calibrated" in the figure. With this set of recommendations, the supplier B is completely out of the picture and has received zero exposure even though Ross's profile does show interest in less popular movies. The calibrated recommendations, on the right side, shows preferred situation where the user has received recommendations that match his interests across the item popularity spectrum, while at the same time both suppliers have received exposure. This shows that the popularity calibration for the users would also benefit the suppliers.

#### 6.2.2.2 Calibrated Popularity

Our proposed technique, *Calibrated Popularity* (CP), is a re-ranking method. I build on the work of Steck [156] in using re-ranking to provide results that better match the distributional properties of user profiles. In Steck's case, the distribution of interest was the distribution of

<sup>&</sup>lt;sup>3</sup> Ross is the name of the first author's favorite character in Friends.



**Figure 6.8:** An illustration of the statistical distance between the distribution of item popularity in the user's profile and in the recommendations.

genres across recommended movies. In our case, it is the distribution of the item popularity I seek to control<sup>4</sup>.

Figure 6.8 shows the distribution of three item groups H, M, and T in a given user's profile. The distribution of the popularity item groups for two recommendation lists  $L_1$  and  $L_2$  are also shown. We can see that the ratios of different item groups in  $L_1$  better match that of user's. The ratios of H, M, and T in  $L_2$ , on the other hand, has no similarity with the those ratios in the user's profile. Therefore, we can say that the distribution of item popularity groups in  $L_1$  has smaller distance to that of user's profile and, by definition, is more calibrated.

CP algorithm operates on an initial recommendation list  $\ell'$  of size m generated by a base recommender to produce a final recommendation list  $\ell$  of size n (m >> n). Similar to [156], I measure distributional differences in the categories (groups) to which items belong  $C = \{c_1, c_2, ..., c_k\}$ . For our purposes, these are the three H, M and T item groups described above (i.e.  $C = \{H, M, T\}$ ).

To do this comparison, I need to compute a discrete probability distribution P for each user u, reflecting the popularity of the items found in their profile  $\rho_u$  over each item group  $c \in C$ . I also need a corresponding distribution Q over any given recommendation list  $\ell$ , indicating what item popularity groups are found among the listed items. For measuring the interest of each user towards each item popularity group, I use Vargas et al.'s [165] measure of category propensity.

<sup>&</sup>lt;sup>4</sup> Note that unlike the genre labels in [156] where it is possible for a movie to have multiple genres, each item only belongs to one item group.
Specifically, I calculate the propensity of each user u towards each item group  $c \in C$  in her profile  $\rho_u(p(c|u))$  and the ratio of such item group in her recommendation list  $\ell_u(q(c|u))$  as follows:

$$p(c|u) = \frac{\sum_{i \in \rho_u} r(u, i) \mathbb{1}(i \in c)}{\sum_{c_j \in C} \sum_{i \in \rho_u} r(u, i) \mathbb{1}(i \in c_j)}, \quad q(c|u) = \frac{\sum_{i \in \ell_u} \mathbb{1}(i \in c)}{\sum_{c_j \in C} \sum_{i \in \ell_u} \mathbb{1}(i \in c_j)}$$
(6.8)

 $\mathbb{1}(.)$  is the indicator function returning zero when its argument is False and 1 otherwise.

In order to determine if a recommendation list is calibrated to a given user, I need to measure the distance between the two probability distributions P and Q. There are a number of metrics for measuring the statistical distance between two distributions [103]. Steck [156] used Kullbeck-Liebler (KL) Divergence. In this paper, I am using Jensen–Shannon divergence, which is a modification of KL Divergence that has two useful properties which KL divergence lacks: 1) it is symmetric:  $\mathfrak{J}(P,Q) = \mathfrak{J}(Q,P)$  and 2) it has always a finite value even when there is a zero in Q. For our application, it is particularly important that the function be well-behaved at the zero point since it is possible for items from certain item groups to be missing from the recommendation list.

Given the *KL* as the KL divergence function, the Jensen–Shannon divergence ( $\mathfrak{J}$ ) between two probability distributions *P* and *Q* is defined as follows:

$$\mathfrak{J}(P,Q) = \frac{1}{2}KL(P,M) + \frac{1}{2}KL(Q,M), \quad M = \frac{1}{2}(P+Q)$$
(6.9)

Similar to [156, 168], I use a weighted sum of relevance and calibration for creating our reranked recommendations. In order to determine the optimal set  $\ell^*$  from the *m* recommended items, I use the weighted combination of relevance and miscalibration as follows:

$$\ell^* = \underset{\ell, |\ell|=n}{\operatorname{arg\,max}} (1-\lambda) \cdot \operatorname{Rel}(\ell) - \lambda \mathfrak{J}(P, Q(\ell))$$
(6.10)

where  $\lambda$  is the weight controlling the relevance versus the popularity calibration and  $Rel(\ell)$  is the sum of the predicted scores for items in  $\ell$ . Since smaller values for  $\mathfrak{J}$  are desirable, I used its negative for our score calculation.

As noted in [156], finding the optimal set  $\ell^*$  is an NP-hard problem. However, similar to the first re-ranking method I proposed earlier, a simple greedy optimization approach is computationally effective. The greedy process starts with an empty set and iteratively adds one item from the larger list to the under-construction list until it reaches the desired length. At each step j, both  $Rel(\ell)$  and  $\mathfrak{J}(P, Q(\ell))$  are calculated using the union of the items that are already in the under-construction list and the item i that is a candidate to be added to the list (i.e.  $\ell@j \cup i$ ) and the item that gives the highest score will be added to the list. This greedy solution achieves a (1 - 1/e) optimality to the best possible  $\ell^*$  list with e being Euler's number.

#### 6.2.2.3 Evaluation

Similar to the two other algorithms I discussed in previous sections, I evaluate the performance of our Calibrated Popularity algorithms with respect to different metrics explained in Chapter 4.

Figure 6.9(a) shows the results for the calibrated popularity algorithm on precision, NDCG, Gini, average percentage of long-tail items in recommendation lists, aggregate diversity, and longtail coverage. Similar to previous techniques, there is a loss in precision when we increase the weight for the calibration component ( $\lambda$ ). Gini, aggregate diversity and long-tail coverage (both list-wise (APL) and overall (LC)) are all improving when we increase the values for  $\lambda$  parameter meaning more unique items are recommended, recommendation are more uniformly distributed and also more long-tail items are appeared in the recommendations. It is interesting that aggregate diversity and long-tail coverage significantly improve up until a certain value for  $\lambda$  ( $\lambda = 0.2$ ) and then it loses this sharp momentum for larger values of  $\lambda$ . The reason is, the calibrated popularity algorithm does not directly aim for increasing aggregate diversity or long-tail coverage but rather tries to give more exposure to different items in different item popularity groups. In other words, after a certain point, it does not add any new item to its list of covered items but tries to give these already covered items more exposure as can be seen from the monotonic improvement for Gini and also *IPD* in Figure 6.9(b) which measures how much the exposure for different items varies from their popularity in the original rating data.

The long-tail enhancement and fairness metrics are all improved as we can see from Figure 6.9(b). Similarly, Figure 6.10(a) and 6.10(b) show the same trends for the calibrated popularity on

different metrics on Last.fm dataset.

Figures 6.11 and 6.12 compares the three algorithms I proposed in this dissertation all within the same plot. RG is the model-based technique based on regularization, XQ is the fairness-aware re-ranking technique based on xQuAD, and CP is the popularity calibration approach I proposed in this section.

First of all, the superiority of both re-ranking algorithms (XQ and CP) over the modelbased technique (RG) is quite obvious in many of the metrics. Overall, the precision for all three algorithms, as mentioned earlier, is decreasing over different values for  $\lambda$ . The precision for XQ, overall, has experienced a lower drop compared to other two indicating it keeps its accuracy better. Also, for smaller values of  $\lambda$ , the precision of RG is higher than that of CP but for larger values it experiences a larger loss.

Looking at Gini for three algorithms we can see that the Gini of RG looks almost like a straight line when it is contrasted with the other two re-ranking techniques. That means, RG does not improve the uniformity of the recommendation distribution compared to CP or XQ. CP is clearly outperforming XQ here.

When it comes to aggregate diversity and long-tail coverage, XQ outperforms the other two. *ESF* which also has a strong connection to aggregate diversity is higher for XQ compared to *CP* and *RG* when we increase  $\lambda$ . However, all the metrics for measuring the proportionality of popularity (*IPD*, *UPD*, and *SPD*) indicate that *CP* outperforms the other two by a large margin. Another interesting pattern we can see in the results for MovieLens dataset, is that *RG* performs worse than *XQ* on *UPD* but it outperforms *XQ* on *SPD*. In other words, *XQ* better keeps the ratio of different item popularity groups in the users' profile when generating the recommendations but the popularity of different suppliers is not as well-kept as in *RG*.

I also want to compare all the three proposed methods I proposed in this dissertation with some of the state-of-the-art techniques for mitigating popularity bias. I used two recent state-ofthe-art re-ranking techniques for mitigating popularity bias in the literature. These two methods are as follows:



(a) Precision, NDCG, Gini, APL, Aggregate Diversity and Long-tail coverage for our Popularity Calibration Re-ranking approach on MovieLens data



(b) Equality of attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popularity Deviation (UPD) and Supplier Popularity Deviation (SPD) of our Popularity Calibration Re-ranking approach on MovieLens data

**Figure 6.9:** Precision, NDCG, Gini, APL, Aggregate Diversity and Long-tail coverage, Equality of attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popularity Deviation (UPD) and Supplier Popularity Deviation (SPD) of our Popularity Calibration Re-ranking approach on MovieLens data



(a) Precision, NDCG, Gini, APL, Aggregate Diversity and Long-tail coverage for our Popularity Calibration Re-ranking approach on Last.fm data



(b) Equality of attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popularity Deviation (UPD) and Supplier Popularity Deviation (SPD) of our Popularity Calibration Re-ranking approach on Last.fm data

**Figure 6.10:** Precision, NDCG, Gini, APL, Aggregate Diversity and Long-tail coverage, Equality of attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popularity Deviation (UPD) and Supplier Popularity Deviation (SPD) of our Popularity Calibration Re-ranking approach on Last.fm data



**Figure 6.11:** Precision, Gini, Gini, APL, Aggregate Diversity and Long-tail coverage, Equality of attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popularity Deviation (UPD) and Supplier Popularity Deviation (SPD) for all three popularity mitigating approaches on MovieLens data



**Figure 6.12:** Precision, Gini, Gini, APL, Aggregate Diversity and Long-tail coverage, Equality of attention Supplier Fairness (ESF), Item Popularity Deviation (IPD), User Popularity Deviation (UPD) and Supplier Popularity Deviation (SPD) for all three popularity mitigating approaches on Last.fm data

- Discrepancy Minimization (DM) [26]: In this method, the goal is to improve the total number of unique recommended items, also referred to as aggregate diversity (see Equation 2.12 in Chapter 2) of recommendations using minimum-cost network flow method to efficiently find recommendation sub-graphs that optimize diversity. Authors in this work define a target distribution of item exposure (i.e. the number of times each item should appear in the recommendations) as a constraint for their objective function. The goal is therefore to minimize the discrepancy of the recommendation frequency for each item and the target distribution.
- FA\*IR (FS) [176]: This method was originally used for improving group fairness in job recommendation and was adapted here to improve the fairness of recommendations in terms of head (*H*) versus long-tail ( $M \cup T$ ) items in recommendations. The algorithm creates queues of protected and unprotected items and merges them using normalized scoring such that protected items get more exposure. I define protected and unprotected groups as long-tail and head items, respectively. I performed grid search over the two hyperparameters, proportion of protected candidates in the top *n* items <sup>5</sup> and significance level<sup>6</sup>, using values of {0.25, 0.5, 0.75, 0.95} and {0.05, 0.1, 0.15}, respectively.

The two re-ranking baselines (DM and FS) and the proposed techniques in this dissertation (RG, XQ and CP) have hyperparameters that control the trade-off between relevance and a second criterion: long-tail regulariation in RG, diversification in XQ, fairness in FS, aggregate diversity in DM, and popularity calibration in CP. To establish a point of comparison across the algorithms, I varied these trade-off hyperparameters for each algorithm, and chose, for each, a hyperparameter that yields a recommendation precision that is within a small margin (Difference in precision is less than 0.01 for MovieLens and less than 0.001 for Last.fm) compared to that off other algorithms since getting exactly an equal precision is not possible for certain algorithms.

First I look at the overall performance of these algorithms. The overall results for different

<sup>&</sup>lt;sup>5</sup> Based on suggestion from the released code, the range should be in [0.02, 0.98]

<sup>&</sup>lt;sup>6</sup> Based on suggestion from the released code, the range should be in [0.01, 0.15]

**Table 6.1:** Results of different algorithms on two datasets. The setting that yields the same precision for all three reranking methods XQ, DM and CP is chosen and the results is reported. Bold values are statistically significant compared to the second best value in each column with significance level of 0.05 ( $\alpha$ =0.05)

	Agg-Div↑	LC↑	$Gini \downarrow$	$ESF\uparrow$	$IPD\downarrow$	$UPD\downarrow$	$SPD\downarrow$
Base	0.194	0.164	0.885	2.956	0.295	0.368	0.143
RG	0.196	0.167	0.872	2.960	0.265	0.341	0.127
XQ	0.384	0.3610	0.839	3.055	0.273	0.308	0.127
DM	0.519	0.501	0.623	3.057	0.213	0.302	0.109
FS	0.198	0.168	0.863	2.965	0.170	0.268	0.109
СР	0.281	0.253	0.831	3.056	0.092	0.152	0.090

algorithms on both MovieLens and Last.fm datasets can be seen in Table 6.1 and 6.2. The arrow next to the metric name indicates the direction of preference for a better outcome.

On MovieLens, with respect to aggregate diversity (*Agg-Div*), long-tail coverage (*LC*), and Gini, On MovieLens, the *DM* algorithm has the best performance on these metrics. *CP* outperforms XQ on aggregate diversity and long-tail coverage but they have similar performance with respect to Gini. Even though *RG* and *FS* have both improved these metrics over the *Base* algorithm they still stand behind the other algorithms. Not only is greater coverage for *DM* achieved for the whole catalog, but a better distribution as shown by its low Gini index. The results for Last.fm dataset in Table 6.2 shows that *CP* outperforms the other algorithms with respect to all metrics followed by *FS* and *XQ*. *RG* still performs the worst among all.

A more detailed picture of the way that item popularity is transformed in recommendation frequency is available in Figure 6.13(a) and 6.13(b). These scatter plots have rating frequency and popularity segments (i.e. item groups) on the x-axis with the *Tail* items (T) having low rating frequency on the left, moving to the popular *Head* items (H) on the right. Also, y-axis shows the frequency of the recommendations for each item (i.e. exposure rate). Looking first at part (a) showing the MovieLens results, we see that, in the *Base* algorithm, the T items are rarely recommended, as previously noted. The XQ, and CP boost the T and M items significantly, while a more limited effect is seen for DM, RG and FS. The division of the items into groups has an



Figure 6.13: Exposure rate vs popularity across different re-ranking algorithms (MovieLens)

**Table 6.2:** Results of different algorithms on two datasets. The setting that yields the same precision for all three reranking methods XQ, DM and CP is chosen and the results is reported. Bold values are statistically significant compared to the second best value in each column with significance level of 0.05 ( $\alpha$ =0.05)

	Agg-Div↑	LC↑	$Gini \downarrow$	$ESF\uparrow$	$IPD\downarrow$	$UPD\downarrow$	$SPD\downarrow$
Base	0.137	0.0970	0.922	4.253	0.354	0.466	0.21
RG	0.145	0.105	0.915	4.324	0.318	0.435	0.184
XQ	0.169	0.129	0.908	4.429	0.331	0.424	0.185
DM	0.175	0.134	0.905	4.274	0.339	0.453	0.2
FS	0.176	0.142	0.891	4.440	0.186	0.328	0.129
СР	0.203	0.172	0.879	4.501	0.171	0.246	0.106

obvious effect as the promotion of lower-popularity items is concentrated at the top end of the T and M items, which could be expected as these are the items in this set with the most number of ratings. Note also that CP and FS algorithms reduce the number of items with the maximum rating frequency (contained in more than 12.5% of recommendation lists).

In Figure 6.13(b), we see the results for Last.fm. We can clearly see that the popularity bias of the *Base* recommender is lower compared to that of for MovieLens in its neglect of the lower parts of the popularity distribution here and that gives all of the re-rankers more to work with. CP and XQ boost the frequency of the recommendations for T items somewhat and reduce the extremes of recommendation frequency. Also, in addition to CP, FS improves the exposure for the items in the M category. Overall, the CP algorithm has performed the best with respect to giving items from different item groups the exposure that they deserve.

There is another interesting pattern that helps explain the performance of the *DM* algorithm on the *Agg-Div*, *LC*, and *Gini* metrics on Movielens dataset. On this dataset, the *DM* algorithm promotes some of the *T* and *M* items as can be seen in the small step increase in the plot in these regions. Thus, it increases aggregate and long-tail coverage since in both of these two metrics, the recommendation frequency does not matter meaning recommending an item even once would be counted the same as recommending it to many users. By giving very small but consistent exposure to this range of items, it improves the Gini index as well.

Figure 6.14 shows a similar plot to Figure 5.7 in Chapter 5 but for the re-ranking algorithms. The black curve shows the cumulative frequency of ratings for different items at a given rank in MovieLens and Last.fm datasets. The x-axis indicates the rank of each item when sorted from most popular to least popular. The blue curves in each figure show the cumulative frequency of the recommendations for each item at a given rank. Three item groups H, M, and T are depicted on the top of each plot. The point that the blue curve intersects with each of the dotted vertical lines shows the cumulative percentage of the recommendations that come from items up that point. For example, on MovieLens, in the Base algorithm, approximately 60% of the recommendation come from items in the H which is significantly higher that what they originally had in rating data (Hitems take up 20% of the ratings). The lower this number is for the re-ranking algorithms, the better they have controlled this bias. RG does not seem to perform well as this intersection is still at around 60%. XQ has slightly lowered this bias to around 50%. CP seems to have the best performance across all re-ranking algorithms and it has lowered the frequency of H items to around 30% which is very close to their original popularity. CP has also given the highest exposure to the T items. The same pattern repeats for the Last.fm data where CP performs well on different item groups compared to other algorithms.

Without a multi-stakeholder analysis, we might conclude from the overall metrics shown in Table 6.1 that *DM* is a superior approach for remediating popularity bias, on MovieLens data. However, evaluating the recommendations from the perspective of the different stakeholders reveal a different story. Because *CP* aims to improve popularity calibration, we might expect that *UPD*, which more or less measures the inverse of average calibration, would be improved and this is indeed a strong effect across both datasets. *CP* is applying its long-tail enhancement where it matters for users and therefore is able to have a big impact on calibration. Thus, we might say that *UPD* is measuring *useful long-tail diversity* in recommendation results. Interestingly, this emphasis on users also has a beneficial effect on the metrics related to other recommendation stakeholders as shown in the superior performance on *IPD*, and *SPD*. All algorithms have a comparable *ESF*.

Additional detail on the comparative performance for different stakeholders and sub-groups



**Figure 6.14:** Cumulative frequency of the ratings for different items at a given rank and their frequency of appearance in the recommendations

can be seen in Figure 6.15. These results compare the popularity deviation metrics (*IPD*, *UPD*, *SPD*) for different sub-groups.

#### 6.2.2.4 Item Groups

First, we can see that all algorithms have positive *IPD* values for items in *H* meaning they all over-recommend popular items. This is not surprising since the *Base* algorithm exhibits a stronger bias towards these items and although the re-ranking techniques have reduced this over-concentration they still have not removed it completely. The *CP* algorithm comes the closest to doing so. Also, on Movielens, in all algorithms except for *CP*, items from the *M* group have negative deviation meaning they are under-represented in the recommendations. But again, *CP* performs significantly better in the *M* group giving these items a slight boost, as we also saw in Figure 6.13.

The result for the *T* items show that *CP* and *XQ* have lower deviation for this item group on MovieLens compared to other algorithms, but all algorithms including *CP* have performed poorly. On Last.fm, *CP* and *XQ* performs better than other algorithms on this item group. So overall, no algorithm has outperformed *CP* in any of these item groups.

#### 6.2.2.5 User Groups

Looking at the charts for user groups, again, we see the superiority of CP. Firstly, on Movie-Lens,  $G_1$ , the group with the highest interest towards popular items, has the lowest UPD (i.e. lowest miscalibration) using all algorithms and  $G_3$  has the highest miscalibration. That shows users with high interest towards popular items are served much better as expected. Regardless, all user groups have experienced the lowest miscalibration using CP. FS also performs well here with lower UPD compared to RG, DM, and XQ. What is interesting about CP and FS with regard to UPD is that they both have comparable UPD for  $G_1$  but the UPD of CP is significantly lower for  $G_2$  and  $G_3$  indicating CP also performs well for users with lesser interest in popular items as it calibrates its recommendations better to different users' profiles. The same for DM and FS where



Figure 6.15: Popularity deviation metrics for different stakeholder groups

FS works better for users in  $G_1$  (lower *UPD*) while DM is superior for  $G_3$ . These are distinctions that we would not have observed without a multi-stakeholder approach to evaluation.

Another view of the performance of different algorithm for different user groups is shown in Figure 6.16 which is a similar plot to what we saw in Figure 5.5 in Chapter 5. Users are ranked based on their interests towards popular items from left to right and the three user groups are shown in each plot. We can see that different recommendation algorithms produced different density of recommended items from different item groups. Some still have recommended many popular items even to the users who have lesser interest items in these items and some have done a better job in giving items from different popularity bins to different users according to their interest. RG still does not look good from the users' perspective as it almost has the shape as the original RankALS. DM seems to the most aggressive one in recommending T items. CP shows a better overall performance as the ratio of different recommended item groups seem to better match the original ratio of these items in the users' profiles.

The difference of performance for different algorithms in terms of recommending the right ratio of different item groups for users in different user groups can be seen in Figure 6.17. The ratio of tail items recommended to all three user groups using RG and FS is essentially zero. DM, XQ, and CP have all recommended tail items but the proportion of different item categories recommended using CP is closer to the historical users' interaction with these different item categories as we saw in Figure 5.6 in Chapter 5.

### 6.2.2.6 Supplier Groups

Finally, the chart for supplier groups similarly confirms the superiority of CP for  $S_1$  and  $S_2$ in this multi-stakeholder analysis. On both datasets, CP has the lowest SPD for the  $S_1$  (group with high popularity). On Movielens, this metric for  $S_2$  has a positive value meaning CP has actually given a boost to suppliers from this group in terms of exposure.

One other interesting finding that can be seen in this chart for MovieLens dataset is the relationship between item popularity and supplier popularity. One might expect that when an algo-



**Figure 6.16:** The ratio of different item groups in the recommendations of different users using the popularity bias mitigation techniques



**Figure 6.17:** The ratio of different item groups in the recommendations of different user groups using the popularity bias mitigation techniques

rithm performs better than another on items with lower popularity (e.g. T) it should also perform better on the suppliers with lowest popularity as these two seem to be correlated. However, as we can see on MovieLens dataset, CP and DM have comparable performance on T items (both have roughly equal IPD) but DM outperforms CP on  $S_3$  (lower SPD). The reason is, we observed not every item from a popular / non-popular supplier is necessarily popular / non-popular and therefore it is possible for an algorithm to perform differently on item groups and supplier groups as it is the case on MovieLens dataset for CP and DM.

Figure 6.18 shows the cumulative Frequency of the recommendations from different supplier groups versus their popularity in rating data using different popularity bias mitigation techniques. All algorithms (even *RG* to some extent) have reduced the over-concentration of the recommendations on the items from popular supplier groups. Especially, *CP* and *FS* seem to have narrowed down the gap between the blue and black curve which shows their superior performance in giving proportionate exposure to different suppliers with a varying degree of popularity.

Overall, we learned that, among all the popularity bias mitigation approaches we studied in this chapter, *CP* can be considered the superior approach since it performs better than the others from the perspective of different stakeholders. This is an interesting finding which means matching the recommendations with the users' range of interests in terms of item popularity not only does it calibrate the recommendations to the users' profiles, it also gives a fairer exposure to items from different popularity groups and also suppliers with different levels of popularity.



**Figure 6.18:** Cumulative Frequency of recommendations from different supplier groups versus their popularity in rating data

# Chapter 7

## Conclusion and future work

## 7.1 Summary of Contributions

#### 7.1.1 Multi-stakeholder recommendation

In Chapter 3 I described a new paradigm for recommender systems' design, implementation, and evaluation. I defined what the characteristics of multi-stakeholder recommendations are and I discussed its connection to some of the existing work in the literature such as group recommendation, people recommendation and matching, fairness-aware recommendation and value-aware recommendation.

Generally speaking, multi-stakeholder recommendation refers to new practice of taking into account the needs and interests of several stakeholders in the recommendation process. The reason this is an important consideration is that, in many real-world recommendation platforms a simple focus on the users may not address all the complexities of the recommender systems. For instance, in an online dating application, both parties (e.g. men vs women) should be satisfied with a given recommendations. In addition, the platform owner may also have other objectives that is beyond the preferences of users such as fairness concerns, premium users vs free subscriptions. As another example, we can think of a music recommendation system as a multi-sided platform where on one side we have the listeners and on the side we have the artists. It is clear that only focusing on the listeners may not be ideal for the overall long-term health of the business as it may lead to the unsatisfaction of the artists and, hence, losing the artists on the platform altering the balance of the marketplace.

I stated that typically three main stakeholders can be observed in many recommendation platform: 1) Users or consumers, 2) Suppliers of the recommendations and 3) the system owner. The ideal situation would be to have a recommender system where every stakeholder is satisfied with the outcome of the recommendations. However, this might not always happen in reality. Depending on the situation, time, and other considerations, the system owner may decide to tune the recommender system so it improves the outcome for a particular stakeholder or group of stakeholders.

#### 7.1.2 Multi-stakeholder Impact of Popularity Bias

In Chapter 5, I have taken a multi-stakeholder view of the problem of popularity bias. I showed that the popularity bias is not just an overall system-level problem which alters the balance of the distribution of the recommendations towards popular items but rather one that impacts the entire eco-system including the users and the suppliers of the recommendations.

Regarding the users, I showed that large segments of the user population in a typical recommender system have a strong interest in items outside of the "short head" of the distribution. Consistent with prior work, I showed that because these long-tail items are recommended less frequently, these groups of users are not well-served and their recommendations are heavily concentrated on popular items despite their interest towards less popular ones.

I also showed that the popularity bias has a ripple-out effect on the other recommendation stakeholders. For example, the less popular suppliers are extremely being impacted by this bias and are not recommended to the users at which they have shown interest at. For example a less popular artist such as a local artist may have significant number of people interested in her/his music but because the recommendations are dominated by the songs from popular artists, the algorithmic popularity bias does not let her/his songs to be recommended to the desired audience.

This multi-stakeholder evaluation of popularity bias has revealed certain aspects of the algorithms' behavior that cannot be captured using standard evaluation methods and metrics that are focused on overall outcomes, including differences across results returned to different user, supplier, and item groups. For example, in Chapter 6 we saw that some algorithms perform better than others for certain user /item /supplier groups and worse on some other groups. That indicates that, when we take an average perspective over all users or items, we cannot see the differences of different algorithms in their performance for different stakeholders.

### 7.1.3 Algorithmic solutions to tackle popularity bias

In Chapter 6, I presented three different recommendation algorithms to control popularity bias. As I discuss in that chapter, generally two types of solutions can be found to address the popularity bias in recommendation: 1) model-based and 2) post-processing re-ranking. I proposed one model-based approach and two re-ranking methods.

The first approach I presented in this dissertation was a model-based technique that works on top of a particular learning-to-rank algorithm (*RankALS*) and changes its objective function so it incorporates the popularity of the recommendations in each given recommendation list. The algorithm aims to diversify the recommendations in each list according to whether they belong to popular items or the long-tail ones. In other words, it penalizes the lists that only contain popular items or only long-tail items. Experimental results showed that our proposed method was able to improve the long-tail properties of the recommendations such as catalog coverage and long-tail coverage.

The second method I proposed to control popularity bias was in the post-processing reranking category. Generally speaking, in a re-ranking technique, the algorithm first generates a larger list of recommendations and the re-ranking acts as a post-processing step to extract the best top recommendations that have both high relevance to the user and also desired characteristics of long-tail. The algorithm was inspired by a well-known diversification method in information retrieval called xQuAD which aims at covering different aspects related to a given query in the returned search results. In my case, I did not have aspects associated with items but rather a category to which each item belongs. I defined two item categories short-head and long-tail and, therefore, the diversification task was to make sure items from different item categories are fairly represented in the recommendations. The experimental results indicated a significant improvement in long-tail properties of the recommendations with a small loss in accuracy.

The third algorithm I proposed in this dissertation was also a post-processing technique but with a different objective. In this method which I called *Calibrated Popularity* the goal was to make sure the recommendation lists for each user are calibrated in terms of the range of popularity of the items in the list compared to the range of item popularity the user had originally shown interest at. For instance, if a user has 20% popular items, 50% items with medium popularity and 30% less popular items in his profile, the recommender system should try to keep this ratio as much as possible while keeping the accuracy loss as small as possible. Experimental results showed that, when evaluated from the multi-stakeholder paradigm discussed in Chapter 4, this method outperforms our two other proposed methods and also two other state-of-the-art methods proposed by other researchers. Compared to the other algorithms, the Calibrated Popularity algorithm matches the distribution of the recommendations in terms of item popularity with that of rating data much better and hence gives a more calibrated lists to the users. I also showed that this approach improves supplier fairness, as measured by the exposure of items from different supplies groups in recommendation lists.

## 7.2 Future Work

The contributions presented in this thesis extend the state-of-the-art of long-tail recommendation, however they are not exhaustive and a number of questions remain for future work. In this section, I outline some of the envisioned lines of work that could follow the contributions of this thesis.

## 7.2.1 User Studies

Throughout the thesis I have been performing evaluation of the proposed algorithms for long-tail recommendation. This has been done in an offline fashion, where we use the historical interaction data of the users with recommendations as a ground truth to measure the effectiveness of the proposed methods. Such evaluation is based on a number of assumptions about how users interact with recommendations and how they perceive these. Although the adapted evaluation protocol follows some of the common approaches taken in recommender systems, conducting user studies may be particularly important when evaluating the long-tail promoting algorithms. The reason for that is the complexity of user needs and interests that are commonly not well reflected in the recorded datasets. Long-tail promotion and diversification is meant to bring (sometimes) content that is not well explored by a user, and its evaluation is not very conclusive in an offline setting since only items that are already in the user's profile are considered good recommendations. In other words, in offline evaluation we have no knowledge about whether a recommended item that was not in the user's profile was a good recommendation or not and, by the nature of this type of evaluation, we consider them as bad recommendations.

Metrics such as long-tail promotion, diversity, and novelty are frequently placed together with accuracy — as I did — and a trade-off of these is analysed in order to develop a better method, a method which offers smaller loss in accuracy and more prevalent diversity or long-tail improvements. However prior research such as the one by Ekastrand et al. [58] has suggested that non-accuracy metrics actually could lead to accuracy improvements. In fact, this is the ultimate goal of making recommendations more diverse and calibrating the recommendations with the range of users' interests — improving user's satisfaction, which means producing recommendations that users feel address their needs. We can say that, all the algorithms I proposed in this dissertation are evaluated in a rather pessimistic way meaning the real trade-off between accuracy and long-tail diversification would probably be better if they were evaluated on a live system and the users had interacted with those recommendations.

#### 7.2.2 Directly Optimizing for Supplier Fairness

The beneficial side-effect of supplier fairness was not directly optimized as our algorithm focused on the user calibration and no information about the suppliers was used in the calibration

algorithm. However, because of the positive correlation between item popularity and the average popularity of the corresponding suppliers, improving popularity calibration has this indirect effect of improving supplier fairness. An interesting extension of this work would be to incorporate the tendency of each user towards different groups of suppliers as well as different popularity categories in the calibration algorithm. I will leave this for future work.

For each user we can define a vector of popularity interest which can encompass both item categories and supplier groups. For instance, sticking with the triple grouping I presented in this thesis, one could define a vector of size 6 that has the interest of the user towards different item categories and supplier groups and the recommendation algorithm should try to calibrate the recommendation list according this vector. How this extended method would perform for different stakeholders and also how it matches the distributional properties of the item popularity of the recommendations with that of rating data will remain as an open question.

#### 7.2.3 Connection Between Popularity Bias and other Types of Fairness

In this dissertation, our main concentration was on measuring the fairness of the recommendations with respect to popularity. In other words, I intended to evaluate the unfairness of a given recommender system for users with different levels of interest in item popularity and also for suppliers with different degrees of popularity. For example, our grouping strategy I described in Chapter 4 was solely based on the information about the popularity of the items, suppliers, or the users' interest towards popular items. However, in many domains, certain aspects associated with users or suppliers might be important for the system designer to consider them as a group. These could be gender, age, race, nationality, or any other types of pre-defined labels for determining a group of people. The connection between popularity bias and the unfairness in those scenarios is an interesting open question that needs further attention. Can controlling popularity bias help making the recommendation to become fairer with respect to those pre-defined labels? What is the connection of what is referred to "protected group" or under-represented group and the group with lesser popularity? In what situations can popularity bias mitigation can be also helpful in alleviating other types of unfairness? Our preliminary results show that popularity bias is indeed connected to certain types of unfairness in recommendation such as miscalibration of the recommendations for certain groups of users [6].

### 7.3 Limitations

This dissertation, like many other scientific works, is not perfect and there are questions that are not addressed. Two main limitations are as follows:

- Not Investigating Pre-processing Techniques: In Chapter 6 three approaches for mitigating popularity bias were mentioned: pre-processing, in-processing, and post-processing. My model-based technique (*RG*) is an in-processing method since it modifies the existing recommendation algorithm. Tow other methods *XQ* and *CP* are both post-processing methods since they work on top of an existing recommendation algorithm. I have not proposed any pre-processing method for two reasons: these methods require sampling the data and this usually leads to increased data sparsity. In addition, the focus of this dissertation was not on comparing different processing methods for mitigating popularity bias but rather to study the impact of popularity bias on different stakeholders and the extent to which different algorithms can mitigate this bias for these stakeholders. Nevertheless, trying different pre-processing techniques to remove the bias in data would be an interesting thread of research.
- Not taking into account the dynamics of item popularity: In real world, items become popular over time and might lose popularity when new items arrive. In fact, if this was not the case, music in 50's would have been still the most popular music now. My analysis does not take into account this dynamic as I was mainly interested in how popularity bias impacts the performance of the algorithms for different stakeholders and my analysis has been under the fact that certain items are popular and others are not. In other words, my analysis sees a snapshot of the current situation of a recommendation algorithm. How-

ever, the dynamics of item popularity could be incorporated as an interesting future work to analyze the popularity bias of different algorithms on different stakeholders over time.

# **Bibliography**

- Himan Abdollahpouri, Gediminas Adomavicius, Robin Burke, Ido Guy, Dietmar Jannach, Toshihiro Kamishima, Jan Krasnodebski, and Luiz Pizzato. Multistakeholder recommendation: Survey and research directions. <u>User Modeling and User-Adapted Interaction</u>, 30:127– 158, 2020.
- [2] Himan Abdollahpouri, Robin Burke, and Bamshad Mobasher. Controlling popularity bias in learning-to-rank recommendation. In Proceedings of the Eleventh ACM Conference on Recommender Systems, pages 42–46. ACM, 2017.
- [3] Himan Abdollahpouri, Robin Burke, and Bamshad Mobasher. Recommender Systems as Multistakeholder Environments. In Proceedings of the 25th Conference on User Modeling, Adaptation and Personalization (UMAP'17), page To appear, Bratislava, Slovakia, 2017.
- [4] Himan Abdollahpouri, Robin Burke, and Bamshad Mobasher. Managing popularity bias in recommender systems with personalized re-ranking. In <u>The Thirty-Second International</u> Flairs Conference, 2019.
- [5] Himan Abdollahpouri and Masoud Mansoury. Multi-sided exposure bias in recommendation. arXiv preprint arXiv:2006.15772, 2020.
- [6] Himan Abdollahpouri, Masoud Mansoury, Robin Burke, and Bamshad Mobasher. The impact of popularity bias on fairness and calibration in recommendation. <u>arXiv preprint</u> arXiv:1910.05755, 2019.
- [7] Himan Abdollahpouri, Masoud Mansoury, Robin Burke, and Bamshad Mobasher. The unfairness of popularity bias in recommendation. In <u>RecSys workshop on Recommendation</u> in Multi-Stakeholder Environments, 2019.
- [8] Panagiotis Adamopoulos and Alexander Tuzhilin. On over-specialization and concentration bias of recommendations: Probabilistic neighborhood selection in collaborative filtering systems. In <u>Proceedings of the 8th ACM Conference on Recommender systems</u>, pages 153–160, 2014.
- [9] Panagiotis Adamopoulos and Alexander Tuzhilin. The business value of recommendations: A privacy-preserving econometric analysis. In Proc. ICIS '15, 2015.
- [10] G. Adomavicius and A. Tuzhilin. Toward the next generation of recommender systems: A survey of the state-of-the-art and possible extensions. <u>IEEE transactions on knowledge and</u> data engineering, pages 734–749, 2005.

- [11] Gediminas Adomavicius and YoungOk Kwon. Toward more diverse recommendations: Item re-ranking methods for recommender systems. In <u>Workshop on Information</u> Technologies and Systems. Citeseer, 2009.
- [12] Gediminas Adomavicius and YoungOk Kwon. Improving aggregate recommendation diversity using ranking-based techniques. <u>IEEE Transactions on Knowledge and Data</u> Engineering, 24(5):896–911, 2011.
- [13] Gediminas Adomavicius and YoungOk Kwon. Maximizing aggregate recommendation diversity: A graph-theoretic approach. In Proc. of the 1st International Workshop on Novelty and Diversity in Recommender Systems (DiveRS 2011), pages 3–10. Citeseer, 2011.
- [14] Gediminas Adomavicius and YoungOk Kwon. Optimization-based approaches for maximizing aggregate recommendation diversity. <u>INFORMS Journal on Computing</u>, 26(2):351– 369, 2014.
- [15] Deepak Agarwal, Bee-Chung Chen, Pradheep Elango, and Xuanhui Wang. Click Shaping to Optimize Multiple Objectives. In Proceedings of the 17th ACM SIGKDD International <u>Conference on Knowledge Discovery and Data Mining</u>, KDD '11, pages 132–140, New York, NY, USA, 2011. ACM.
- [16] Deepak Agarwal, Bee-Chung Chen, Pradheep Elango, and Xuanhui Wang. Personalized Click Shaping Through Lagrangian Duality for Online Recommendation. In Proceedings of the 35th International ACM SIGIR Conference on Research and Development in Information Retrieval, SIGIR '12, pages 485–494, New York, NY, USA, 2012. ACM.
- [17] Charu C Aggarwal. Neighborhood-based collaborative filtering. In <u>Recommender systems</u>, pages 29–70. Springer, 2016.
- [18] Fabio Aiolli. Efficient top-n recommendation for very large scale binary rated datasets. In Proceedings of the 7th ACM conference on Recommender systems, pages 273–280, 2013.
- [19] L. Akoglu and C. Faloutsos. Valuepick: Towards a value-oriented dual-goal recommender system. In ICDM '10 Workshops, pages 1151–1158, 2010.
- [20] Ethem Alpaydin. Introduction to machine learning. MIT press, 2020.
- [21] Sindy N Escobar Alvarez, Reshma Jagsi, Stephanie B Abbuhl, Carole J Lee, and Elizabeth R Myers. Promoting gender equity in grant making: what can a funder do? <u>The Lancet</u>, 393(10171):e9–e11, 2019.
- [22] C. Anderson. The long tail: Why the future of business is selling less of more. Hyperion Books, 2008.
- [23] Chris Anderson. The long tail: Why the future of business is selling more for less. Hachette Books, 2006.
- [24] Eugene W Anderson. Customer satisfaction and word of mouth. Journal of service research, 1(1):5–17, 1998.
- [25] Arda Antikacioglu and R Ravi. Post processing recommender systems for diversity. In Proceedings of the 23rd ACM SIGKDD International Conference on Knowledge Discovery and Data Mining, pages 707–716, 2017.

- [26] Arda Antikacioglu and R. Ravi. Post processing recommender systems for diversity. In <u>In</u> <u>Proceedings of the 23rd ACM SIGKDD International Conference on Knowledge Discovery</u> and Data Mining, pages 707–716, 2017.
- [27] Kenneth J Arrow, Amartya Sen, and Kotaro Suzumura. <u>Handbook of social choice and</u> welfare, volume 2. Elsevier, 2010.
- [28] Amos Azaria, Avinatan Hassidim, Sarit Kraus, Adi Eshkol, Ofer Weintraub, and Irit Netanely. Movie recommender system for profit maximization. In <u>Proc. RecSys '13</u>, pages 121–128, 2013.
- [29] Lynn A Barnett and Sandra Wolf Klitzing. Boredom in free time: Relationships with personality, affect, and motivation for different gender, racial and ethnic student groups. <u>Leisure</u> sciences, 28(3):223–244, 2006.
- [30] Mohammad Hossein Bateni, Yiwei Chen, Dragos F Ciocan, and Vahab Mirrokni. Fair resource allocation in a volatile marketplace. 2016.
- [31] Alejandro BellogíN, Iván Cantador, and Pablo Castells. A comparative study of heterogeneous item recommendations in social systems. Information Sciences, 221:142–169, 2013.
- [32] James Bennett, Stan Lanning, et al. The netflix prize. In Proceedings of KDD cup and workshop, volume 2007, page 35. Citeseer, 2007.
- [33] R Alexander Bentley, Mark E Madsen, and Paul Ormerod. Physical space and long-tail markets. Physica A: Statistical Mechanics and its Applications, 388(5):691–696, 2009.
- [34] David M Blei, Andrew Y Ng, and Michael I Jordan. Latent dirichlet allocation. Journal of machine Learning research, 3(Jan):993–1022, 2003.
- [35] J.S. Breese, D. Heckerman, C. Kadie, et al. Empirical analysis of predictive algorithms for collaborative filtering. In <u>Proceedings of the 14th conference on Uncertainty in Artificial</u> Intelligence, pages 43–52, 1998.
- [36] Erik Brynjolfsson, Yu Jeffrey Hu, and Michael D Smith. From niches to riches: Anatomy of the long tail. Sloan Management Review, 47(4):67–71, 2006.
- [37] R. Burke. Hybrid recommender systems: Survey and experiments. User Modeling and User-Adapted Interaction, 12(4):331–370, 2002.
- [38] R. Burke. Hybrid web recommender systems. In <u>The adaptive web</u>, pages 377–408. Springer-Verlag, 2007.
- [39] R. Burke and M. Ramezani. Matching recommendation technologies and domains. Recommender Systems Handbook, pages 367–386, 2011.
- [40] R.D. Burke, K.J. Hammond, and BC Yound. The FindMe approach to assisted browsing. IEEE Expert, 12(4):32–40, 1997.
- [41] Robin Burke. Multisided Fairness for Recommendation. In <u>Workshop on Fairness</u>, <u>Accountability and Transparency in Machine Learning (FATML)</u>, page 5 pages, Halifax, Nova Scotia, 2017.

- [43] Robin Burke, Jackson Kontny, and Nasim Sonboli. Synthetic attribute data for evaluating consumer-side fairness. arXiv preprint arXiv:1809.04199, 2018.
- [44] Pedro G. Campos, Fernando Díez, and Iván Cantador. Time-aware recommender systems: a comprehensive survey and analysis of existing evaluation protocols. <u>User Modeling and</u> User-Adapted Interaction, 24(1):67–119, 2014.
- [45] Rocío Cañamares and Pablo Castells. A probabilistic reformulation of memory-based collaborative filtering: Implications on popularity biases. In Proceedings of the 40th International <u>ACM SIGIR Conference on Research and Development in Information Retrieval</u>, pages 215– 224, 2017.
- [46] Pablo Castells, Saúl Vargas, and Jun Wang. Novelty and diversity metrics for recommender systems: choice, discovery and relevance. In <u>Proceedings of International Workshop on</u> Diversity in Document Retrieval (DDR), pages 29–37. ACM Press, 2011.
- [47] Oscar Celma and Pedro Cano. From hits to niches?: or how popular artists can bias music recommendation and discovery. In Proceedings of the 2nd KDD Workshop on Large-Scale Recommender Systems and the Netflix Prize Competition, page 5. ACM, 2008.
- [48] Jun Chen, Chaokun Wang, Jianmin Wang, and S Yu Philip. Recommendation for repeat consumption from user implicit feedback. <u>IEEE Transactions on Knowledge and Data</u> Engineering, 28(11):3083–3097, 2016.
- [49] Pei-Yu Sharon Chen, Shin-yi Wu, and Jungsun Yoon. The impact of online recommendations and consumer feedback on sales. In Proc. ICIS '04, pages 711–724, 2004.
- [50] Ludovik Coba, Panagiotis Symeonidis, and Markus Zanker. Novelty-aware matrix factorization based on items' popularity. In International Conference of the Italian Association for Artificial Intelligence, pages 516–527. Springer, 2018.
- [51] Paolo Cremonesi, Yehuda Koren, and Roberto Turrin. Performance of recommender algorithms on top-n recommendation tasks. In <u>Proceedings of the fourth ACM conference on</u> Recommender systems, pages 39–46, 2010.
- [52] M. Deshpande and G. Karypis. Item-Based Top-N Recommendation Algorithms. <u>ACM</u> Transactions on Information Systems, 22(1):143–177, 2004.
- [53] C. Desrosiers and G. Karypis. A comprehensive survey of neighborhood-based recommendation methods. Recommender Systems Handbook, pages 107–144, 2011.
- [54] T. Dietterich. Ensemble methods in machine learning. <u>Multiple classifier systems</u>, pages 1–15, 2000.
- [55] Kowald Dominik, Schedl Markus, and Lex Elisabeth. The unfairness of popularity bias in music recommendation: A reproducibility study. arXiv preprint arXiv:1912.04696, 2019.

- [56] Gideon Dror, Noam Koenigstein, Yehuda Koren, and Markus Weimer. The yahoo! music dataset and kdd-cup'11. In <u>Proceedings of the 2011 International Conference on KDD Cup</u> 2011-Volume 18, pages 3–18, 2011.
- [57] Trevor Egli, Helen W Bland, Bridget F Melton, and Daniel R Czech. Influence of age, sex, and race on college students' exercise motivation of physical activity. Journal of American college health, 59(5):399–406, 2011.
- [58] Michael D Ekstrand, F Maxwell Harper, Martijn C Willemsen, and Joseph A Konstan. User perception of differences in recommender algorithms. In <u>Proceedings of the 8th ACM</u> Conference on Recommender systems, pages 161–168, 2014.
- [59] Michael D Ekstrand, Mucun Tian, Ion Madrazo Azpiazu, Jennifer D Ekstrand, Oghenemaro Anuyah, David McNeill, and Maria Soledad Pera. All the cool kids, how do they fit in?: Popularity and demographic biases in recommender evaluation and effectiveness. In Conference on Fairness, Accountability and Transparency, pages 172–186, 2018.
- [60] Michael D Ekstrand, Mucun Tian, Mohammed R Imran Kazi, Hoda Mehrpouyan, and Daniel Kluver. Exploring author gender in book rating and recommendation. In Proceedings of the 12th ACM Conference on Recommender Systems, pages 242–250. ACM, 2018.
- [61] David Evans, Richard Schmalensee, Michael Noel, Howard Chang, and Daniel Garcia-Swartz. <u>Platform economics: Essays on multi-sided businesses</u>. Competition Policy International, 2011.
- [62] David S. Evans and Richard Schmalensee. <u>Matchmakers: The New Economics of Multisided</u> Platforms. Harvard Business Review Press, 2016.
- [63] Yi-Wen Fan and Yi-Feng Miao. Effect of electronic word-of-mouth on consumer purchase intention: The perspective of gender differences. <u>International Journal of Electronic Business</u> Management, 10(3):175, 2012.
- [64] R Edward Freeman. <u>Strategic management: A stakeholder approach</u>. Cambridge university press, 2010.
- [65] Florent Garcin, Christos Dimitrakakis, and Boi Faltings. Personalized news recommendation with context trees. In <u>Proceedings of the 7th ACM conference on Recommender</u> systems, pages 105–112, 2013.
- [66] Mouzhi Ge, Carla Delgado-Battenfeld, and Dietmar Jannach. Beyond accuracy: evaluating recommender systems by coverage and serendipity. In <u>Proceedings of the fourth ACM</u> conference on Recommender systems, pages 257–260. ACM, 2010.
- [67] Mustansar Ali Ghazanfar and Adam Prügel-Bennett. Leveraging clustering approaches to solve the gray-sheep users problem in recommender systems. <u>Expert Systems with</u> Applications, 41(7):3261–3275, 2014.
- [68] D. Goldberg, D. Nichols, B.M. Oki, and D. Terry. Using collaborative filtering to weave an information tapestry. Communications of the ACM, 35(12):70, 1992.

- [69] Kenneth E Goodpaster. Business ethics and stakeholder analysis. <u>Business ethics quarterly</u>, pages 53–73, 1991.
- [70] Philipp Hacker. Teaching fairness to artificial intelligence: Existing and novel strategies against algorithmic discrimination under eu law. 2018.
- [71] Moritz Hardt, Eric Price, and Nati Srebro. Equality of opportunity in supervised learning. In Advances in neural information processing systems, pages 3315–3323, 2016.
- [72] F Maxwell Harper and Joseph A Konstan. The movielens datasets: History and context. Acm transactions on interactive intelligent systems (tiis), 5(4):1–19, 2015.
- [73] F Maxwell Harper and Joseph A Konstan. The movielens datasets: History and context. ACM Transactions on Interactive Intelligent Systems (TiiS), 5(4):19, 2015.
- [74] J.L. Herlocker, J.A. Konstan, L.G. Terveen, and J.T. Riedl. Evaluating collaborative filtering recommender systems. ACM Transactions on Information Systems (TOIS), 22(1):53, 2004.
- [75] W. Hill, L. Stead, M. Rosenstein, and G. Furnas. Recommending and evaluating choices in a virtual community of use. In <u>Proceedings of the SIGCHI conference on Human factors in</u> computing systems, pages 194–201. ACM Press/Addison-Wesley Publishing Co., 1995.
- [76] Thomas Hofmann. Unsupervised learning by probabilistic latent semantic analysis. Machine learning, 42(1-2):177–196, 2001.
- [77] Ganesh Iyer, David Soberman, and J Miguel Villas-Boas. The targeting of advertising. Marketing Science, 24(3):461–476, 2005.
- [78] Michael Jahrer and Andreas Töscher. Collaborative filtering ensemble for ranking. In Proceedings of the 2011 International Conference on KDD Cup 2011-Volume 18, pages 153– 167, 2011.
- [79] Michael Jahrer and Andreas Töscher. Collaborative filtering ensemble. In KDD Cup, pages 61–74, 2012.
- [80] Tamas Jambor and Jun Wang. Optimizing Multiple Objectives in Collaborative Filtering. In Proceedings of the Fourth ACM Conference on Recommender Systems, RecSys '10, pages 55–62, New York, NY, USA, 2010. ACM.
- [81] Dietmar Jannach and Gediminas Adomavicius. Recommendations with a purpose. In Proceedings of the 10th ACM Conference on Recommender Systems, pages 7–10. ACM, 2016.
- [82] Dietmar Jannach, Lukas Lerche, and Michael Jugovac. Item familiarity as a possible confounding factor in user-centric recommender systems evaluation. <u>i-com Journal of</u> Interactive Media, 14(1):29–39, 2015.
- [83] Dietmar Jannach, Lukas Lerche, Iman Kamehkhosh, and Michael Jugovac. What recommenders recommend: an analysis of recommendation biases and possible countermeasures. User Modeling and User-Adapted Interaction, 25(5):427–491, 2015.

- [84] Kalervo Järvelin and Jaana Kekäläinen. Ir evaluation methods for retrieving highly relevant documents. In <u>ACM SIGIR Forum</u>, volume 51, pages 243–250. ACM New York, NY, USA, 2017.
- [85] Gawesh Jawaheer, Peter Weller, and Patty Kostkova. Modeling user preferences in recommender systems: A classification framework for explicit and implicit user feedback. <u>ACM</u> Transactions on Interactive Intelligent Systems (TiiS), 4(2):1–26, 2014.
- [86] Ray Jiang, Silvia Chiappa, Tor Lattimore, András György, and Pushmeet Kohli. Degenerate feedback loops in recommender systems. In <u>Proceedings of the 2019 AAAI/ACM</u> Conference on AI, Ethics, and Society, pages 383–390, 2019.
- [87] Yuanchun Jiang and Yezheng Liu. Optimization of online promotion: a profit-maximizing model integrating price discount and product recommendation. <u>International Journal of</u> Information Technology & Decision Making, 11(05):961–982, 2012.
- [88] Marius Kaminskas and Derek Bridge. Diversity, serendipity, novelty, and coverage: a survey and empirical analysis of beyond-accuracy objectives in recommender systems. <u>ACM</u> Transactions on Interactive Intelligent Systems (TiiS), 7(1):1–42, 2016.
- [89] Toshihiro Kamishima and Shotaro Akaho. Considerations on recommendation independence for a find-good-items task. In <u>Workshop on Responsible Recommendation (FATRec)</u>, Como, Italy, 2017.
- [90] Toshihiro Kamishima, Shotaro Akaho, Hideki Asoh, and Jun Sakuma. Correcting popularity bias by enhancing recommendation neutrality. In <u>Poster Proceedings of the 8th ACM</u> <u>Conference on Recommender Systems, RecSys 2014, Foster City, Silicon Valley, CA, USA,</u> October 6-10, 2014, 2014.
- [91] Alexandros Karatzoglou, Linas Baltrunas, and Yue Shi. Learning to rank for recommender systems. In <u>Proceedings of the 7th ACM conference on Recommender systems</u>, pages 493– 494, 2013.
- [92] Mesut Kaya and Derek Bridge. A comparison of calibrated and intent-aware recommendations. In Proceedings of the 13th ACM Conference on Recommender Systems, pages 151–159, 2019.
- [93] Finn Kensing and Jeanette Blomberg. Participatory design: Issues and concerns. <u>Computer</u> Supported Cooperative Work (CSCW), 7(3-4):167–185, 1998.
- [94] Young Ae Kim and Jaideep Srivastava. Impact of social influence in e-commerce decision making. In <u>Proceedings of the ninth international conference on Electronic commerce</u>, pages 293–302, 2007.
- [95] Daniel Kluver and Joseph A Konstan. Evaluating recommender behavior for new users. In Proceedings of the 8th ACM Conference on Recommender Systems, pages 121–128, 2014.
- [96] J.A. Konstan, B.N. Miller, D. Maltz, J.L. Herlocker, L.R. Gordon, and J. Riedl. GroupLens: Applying Collaborative Filtering to Usenet News. <u>Communications of the ACM</u>, 40(3):77–7, 1997.

- [97] Y. Koren. Factorization meets the neighborhood: a multifaceted collaborative filtering model. In <u>Proceeding of the 14th ACM SIGKDD international conference on Knowledge</u> discovery and data mining, pages 426–434. ACM, 2008.
- [98] Yehuda Koren and Robert Bell. Advances in collaborative filtering. In <u>Recommender</u> systems handbook, pages 77–118. Springer, 2015.
- [99] Yehuda Koren, Robert Bell, and Chris Volinsky. Matrix factorization techniques for recommender systems. Computer, 42(8), 2009.
- [100] Kee-Young Kwahk and Xi Ge. The effects of social media on e-commerce: A perspective of social impact theory. In 2012 45th Hawaii international conference on system sciences, pages 1814–1823. IEEE, 2012.
- [101] Neal Leavitt. Recommendation technology: Will it boost e-commerce? <u>Computer</u>, 39(5):13– 16, 2006.
- [102] Eric L Lee, Jing-Kai Lou, Wei-Ming Chen, Yen-Chi Chen, Shou-De Lin, Yen-Sheng Chiang, and Kuan-Ta Chen. Fairness-aware loan recommendation for microfinance services. In Proceedings of the 2014 international conference on social computing, pages 1–4, 2014.
- [103] Jianhua Lin. Divergence measures based on the shannon entropy. <u>IEEE Transactions on</u> Information theory, 37(1):145–151, 1991.
- [104] Yuzhong Lin, Joran Jessurun, Bauke De Vries, and Harry Timmermans. Motivate: Towards context-aware recommendation mobile system for healthy living. In <u>Pervasive Computing</u> <u>Technologies for Healthcare (PervasiveHealth), 2011 5th International Conference on, pages</u> 250–253. IEEE, 2011.
- [105] Roderick JA Little and Donald B Rubin. <u>Statistical analysis with missing data</u>, volume 793. John Wiley & Sons, 2019.
- [106] Jiahui Liu, Peter Dolan, and Elin Rønby Pedersen. Personalized news recommendation based on click behavior. In Proceedings of the 15th international conference on Intelligent user interfaces, pages 31–40, 2010.
- [107] Giseli Rabello Lopes, Mirella M Moro, Leandro Krug Wives, and José Palazzo Moreira De Oliveira. Collaboration recommendation on academic social networks. In <u>International</u> Conference on Conceptual Modeling, pages 190–199. Springer, 2010.
- [108] Pasquale Lops, Marco De Gemmis, and Giovanni Semeraro. Content-based recommender systems: State of the art and trends. In <u>Recommender systems handbook</u>, pages 73–105. Springer, 2011.
- [109] Masoud Mansoury, Himan Abdollahpouri, Mykola Pechenizkiy, Bamshad Mobasher, and Robin Burke. Feedback loop and bias amplification in recommender systems. <u>arXiv preprint</u> arXiv:2007.13019, 2020.
- [110] Sean M McNee, John Riedl, and Joseph A Konstan. Being accurate is not enough: how accuracy metrics have hurt recommender systems. In <u>CHI'06 extended abstracts on Human</u> factors in computing systems, pages 1097–1101, 2006.
- [111] Rishabh Mehrotra, James McInerney, Hugues Bouchard, Mounia Lalmas, and Fernando Diaz. Towards a fair marketplace: Counterfactual evaluation of the trade-off between relevance, fairness & satisfaction in recommendation systems. In <u>Proceedings of the 27th ACM</u> <u>International Conference on Information and Knowledge Management</u>, pages 2243–2251. ACM, 2018.
- [112] Aranyak Mehta, Amin Saberi, Umesh Vazirani, and Vijay Vazirani. Adwords and generalized online matching. Journal of the ACM, 54(5):22, 2007.
- [113] Seiki Miyamoto, Takumi Zamami, and Hayato Yamana. Improving recommendation diversity across users by reducing frequently recommended items. In <u>2018 IEEE International</u> Conference on Big Data (Big Data), pages 5392–5394. IEEE, 2018.
- [114] Hervé Moulin. Fair division and collective welfare. MIT press, 2004.
- [115] Arvind Narayanan. Translation tutorial: 21 fairness definitions and their politics. In <u>Proc.</u> Conf. Fairness Accountability Transp., New York, USA, volume 1170, 2018.
- [116] Phong Nguyen, John Dines, and Jan Krasnodebski. A multi-objective learning to rerank approach to optimize online marketplaces for multiple stakeholders. <u>arXiv preprint</u> arXiv:1708.00651, 2017.
- [117] Tien T Nguyen, F Maxwell Harper, Loren Terveen, and Joseph A Konstan. User personality and user satisfaction with recommender systems. <u>Information Systems Frontiers</u>, 20(6):1173–1189, 2018.
- [118] Tien T Nguyen, Pik-Mai Hui, F Maxwell Harper, Loren Terveen, and Joseph A Konstan. Exploring the filter bubble: the effect of using recommender systems on content diversity. In <u>Proceedings of the 23rd international conference on World wide web</u>, pages 677–686. ACM, 2014.
- [119] Xia Ning, Christian Desrosiers, and George Karypis. A comprehensive survey of neighborhood-based recommendation methods. In <u>Recommender systems handbook</u>, pages 37–76. Springer, 2015.
- [120] Douglas W Oard, Jinmook Kim, et al. Implicit feedback for recommender systems. In Proceedings of the AAAI workshop on recommender systems, volume 83. WoUongong, 1998.
- [121] Gal Oestreicher-Singer and Arun Sundararajan. The visible hand? demand effects of recommendation networks in electronic markets. Manage. Sci., 58(11):1963–1981, 2012.
- [122] Ioannis Paparrizos, B Barla Cambazoglu, and Aristides Gionis. Machine learned job recommendation. In <u>Proceedings of the fifth ACM Conference on Recommender Systems</u>, pages 325–328, 2011.
- [123] Yoon-Joo Park and Alexander Tuzhilin. The long tail of recommender systems and how to leverage it. In <u>Proceedings of the 2008 ACM conference on Recommender systems</u>, pages 11–18, 2008.

- [124] Bhavik Pathak, Robert Garfinkel, Ram D Gopal, Rajkumar Venkatesan, and Fang Yin. Empirical analysis of the impact of recommender systems on sales. <u>Journal of Management</u> Information Systems, 27(2):159–188, 2010.
- [125] M. Pazzani and D. Billsus. Content-based recommendation systems. <u>The adaptive web</u>, pages 325–341, 2007.
- [126] István Pilászy, Dávid Zibriczky, and Domonkos Tikk. Fast als-based matrix factorization for explicit and implicit feedback datasets. In <u>Proceedings of the fourth ACM conference on</u> Recommender systems, pages 71–78, 2010.
- [127] Luiz Pizzato, Tomek Rej, Thomas Chung, Irena Koprinska, and Judy Kay. Recon: a reciprocal recommender for online dating. In Proceedings of the fourth ACM conference on Recommender systems, pages 207–214. ACM, 2010.
- [128] Victor Ponce, Jean-Pierre Deschamps, Louis-Philippe Giroux, Farzad Salehi, and Bessam Abdulrazak. Quefaire: context-aware in-person social activity recommendation system for active aging. In Inclusive Smart Cities and e-Health, pages 64–75. Springer, 2015.
- [129] Pearl Pu, Li Chen, and Rong Hu. A user-centric evaluation framework for recommender systems. In <u>Proceedings of the fifth ACM conference on Recommender systems</u>, pages 157– 164. ACM, 2011.
- [130] Carl Edward Rasmussen. Gaussian processes in machine learning. In <u>Summer School on</u> Machine Learning, pages 63–71. Springer, 2003.
- [131] Steffen Rendle and Christoph Freudenthaler. Improving pairwise learning for item recommendation from implicit feedback. In Proceedings of the 7th ACM international conference on Web search and data mining, pages 273–282, 2014.
- [132] Jasson DM Rennie and Nathan Srebro. Fast maximum margin matrix factorization for collaborative prediction. In <u>Proceedings of the 22nd international conference on Machine</u> learning, pages 713–719, 2005.
- [133] Paul Resnick, R Kelly Garrett, Travis Kriplean, Sean A Munson, and Natalie Jomini Stroud. Bursting your (filter) bubble: strategies for promoting diverse exposure. In <u>Proceedings of</u> the 2013 conference on Computer supported cooperative work companion, pages 95–100. ACM, 2013.
- [134] Francesco Ricci, Lior Rokach, and Bracha Shapira. Introduction to recommender systems handbook. In Recommender systems handbook, pages 1–35. Springer, 2011.
- [135] Jean-Charles Rochet and Jean Tirole. Platform competition in two-sided markets. Journal of the European Economic Association, 1(4):990–1029, 2003.
- [136] Mario Rodriguez, Christian Posse, and Ethan Zhang. Multiple objective optimization in recommender systems. In <u>Proceedings of the sixth ACM conference on Recommender systems</u>, pages 11–18. ACM, 2012.
- [137] Mario Rodriguez, Christian Posse, and Ethan Zhang. Multiple objective optimization in recommender systems. In Proceedings of the sixth ACM conference on Recommender systems, pages 11–18. ACM, 2012.

- [138] Alvin E Roth and Marilda Sotomayor. Two-sided matching. <u>Handbook of game theory with</u> economic applications, 1:485–541, 1992.
- [139] Donald B Rubin. Inference and missing data. Biometrika, 63(3):581–592, 1976.
- [140] Alan Said, Benjamin Kille, Brijnesh J Jain, and Sahin Albayrak. Increasing diversity through furthest neighbor-based recommendation. Proceedings of the WSDM, 12, 2012.
- [141] Robert Sanders. The pareto principle: its use and abuse. Journal of Services Marketing, 1987.
- [142] Rodrygo LT Santos, Craig Macdonald, and Iadh Ounis. Exploiting query reformulations for web search result diversification. In Proceedings of the 19th international conference on World wide web, pages 881–890. ACM, 2010.
- [143] Rodrygo LT Santos, Craig Macdonald, Iadh Ounis, et al. Search result diversification. Foundations and Trends® in Information Retrieval, 9(1):1–90, 2015.
- [144] Badrul Sarwar, George Karypis, Joseph Konstan, and John Riedl. Item-based collaborative filtering recommendation algorithms. In <u>Proceedings of the 10th international conference</u> on World Wide Web, pages 285–295. ACM, 2001.
- [145] Badrul Sarwar, George Karypis, Joseph Konstan, and John Riedl. Item-based collaborative filtering recommendation algorithms. In <u>Proceedings of the 10th international conference</u> on World Wide Web, pages 285–295. ACM, 2001.
- [146] J Ben Schafer, Joseph Konstan, and John Riedl. Recommender systems in e-commerce. In Proceedings of the 1st ACM conference on Electronic commerce, pages 158–166, 1999.
- [147] Markus Schedl. The lfm-1b dataset for music retrieval and recommendation. In Proceedings of the 2016 ACM on International Conference on Multimedia Retrieval, pages 103–110, 2016.
- [148] Oguz Semerci, Alois Gruson, Catherinee Edwards, Ben Lacker, Clay Gibson, and Vladan Radosavljevic. Homepage personalization at spotify. In <u>Proceedings of the 13th ACM</u> Conference on Recommender Systems, pages 527–527. ACM, 2019.
- [149] Guy Shani and Asela Gunawardana. Evaluating recommendation systems. In Recommender systems handbook, pages 257–297. Springer, 2011.
- [150] Upendra Shardanand and Pattie Maes. Social information filtering: algorithms for automating "word of mouth". In Proceedings of the SIGCHI conference on Human factors in computing systems, pages 210–217, 1995.
- [151] Yue Shi, Alexandros Karatzoglou, Linas Baltrunas, Martha Larson, Nuria Oliver, and Alan Hanjalic. Climf: learning to maximize reciprocal rank with collaborative less-is-more filtering. In <u>Proceedings of the sixth ACM conference on Recommender systems</u>, pages 139–146, 2012.
- [152] Yue Shi, Martha Larson, and Alan Hanjalic. List-wise learning to rank with matrix factorization for collaborative filtering. In <u>Proceedings of the fourth ACM conference on</u> Recommender systems, pages 269–272, 2010.

- [153] Barry Smyth and Paul McClave. Similarity vs. diversity. In <u>Case-Based Reasoning Research</u> and Development, pages 347–361. Springer, 2001.
- [154] Harald Steck. Item popularity and recommendation accuracy. In <u>Proceedings of the fifth</u> ACM conference on Recommender systems, pages 125–132, 2011.
- [155] Harald Steck. Evaluation of recommendations: rating-prediction and ranking. In Proceedings of the 7th ACM conference on Recommender systems, pages 213–220, 2013.
- [156] Harald Steck. Calibrated recommendations. In Proceedings of the 12th ACM Conference on Recommender Systems, pages 154–162. ACM, 2018.
- [157] Wenlong Sun, Sami Khenissi, Olfa Nasraoui, and Patrick Shafto. Debiasing the humanrecommender system feedback loop in collaborative filtering. In <u>Companion Proceedings</u> of The 2019 World Wide Web Conference, pages 645–651, 2019.
- [158] Ozge Sürer, Robin Burke, and Edward C Malthouse. Multistakeholder recommendation with provider constraints. In Proceedings of the 12th ACM Conference on Recommender Systems, pages 54–62. ACM, 2018.
- [159] Richard S Sutton, Andrew G Barto, et al. <u>Introduction to reinforcement learning</u>, volume 135. MIT press Cambridge, 1998.
- [160] Krysta M. Svore, Maksims N. Volkovs, and Christopher J.C. Burges. Learning to Rank with Multiple Objective Functions. In Proceedings of the 20th International Conference on World Wide Web, WWW '11, pages 367–376, New York, NY, USA, 2011. ACM.
- [161] Gábor Takács and Domonkos Tikk. Alternating least squares for personalized ranking. In Proceedings of the sixth ACM conference on Recommender systems, pages 83–90. ACM, 2012.
- [162] Jie Tang, Sen Wu, Jimeng Sun, and Hang Su. Cross-domain collaboration recommendation. In Proceedings of the 18th ACM SIGKDD international conference on Knowledge discovery and data mining, pages 1285–1293. ACM, 2012.
- [163] Saúl Vargas and Pablo Castells. Rank and relevance in novelty and diversity metrics for recommender systems. In Proceedings of the fifth ACM conference on Recommender systems, pages 109–116. ACM, 2011.
- [164] Saúl Vargas and Pablo Castells. Rank and relevance in novelty and diversity metrics for recommender systems. In Proceedings of the Fifth ACM Conference on Recommender Systems, RecSys '11, pages 109–116, New York, NY, USA, 2011. ACM.
- [165] Saúl Vargas and Pablo Castells. Exploiting the diversity of user preferences for recommendation. In Proceedings of the 10th conference on open research areas in information retrieval, pages 129–136, 2013.
- [166] Saúl Vargas and Pablo Castells. Improving sales diversity by recommending users to items. In Proceedings of the 8th ACM Conference on Recommender systems, pages 145–152, 2014.
- [167] Jacek Wasilewski and Neil Hurley. Incorporating diversity in a learning to rank recommender system. In FLAIRS Conference, pages 572–578, 2016.

- [168] Jacek Wasilewski and Neil Hurley. Intent-aware item-based collaborative filtering for personalised diversification. In <u>Proceedings of the 26th Conference on User Modeling</u>, Adaptation and Personalization, pages 81–89. ACM, 2018.
- [169] Jacek Wasilewski and Neil Hurley. Bayesian personalized ranking for novelty enhancement. In Proceedings of the 27th ACM Conference on User Modeling, Adaptation and Personalization, pages 144–148, 2019.
- [170] Markus Weimer, Alexandros Karatzoglou, Quoc V Le, and Alex J Smola. Cofi rankmaximum margin matrix factorization for collaborative ranking. In <u>Advances in neural</u> information processing systems, pages 1593–1600, 2008.
- [171] Shengli Wu. The weighted condorcet fusion in information retrieval. <u>Information processing</u> & management, 49(1):108–122, 2013.
- [172] Peng Xia, Benyuan Liu, Yizhou Sun, and Cindy Chen. Reciprocal recommendation system for online dating. In <u>Proceedings of the 2015 IEEE/ACM International Conference on</u> Advances in Social Networks Analysis and Mining 2015, pages 234–241. ACM, 2015.
- [173] Sirui Yao and Bert Huang. Beyond parity: Fairness objectives for collaborative filtering. In Advances in Neural Information Processing Systems, pages 2921–2930, 2017.
- [174] Huang, Bert Yao, Sirui. New fairness metrics for recommendation that embrace differences. In Workshop on Fairness, Accountability and Transparency in Machine Learning (FATML), page 5 pages, Halifax, Nova Scotia, 2017.
- [175] Shuai Yuan, Ahmad Zainal Abidin, Marc Sloan, and Jun Wang. Internet advertising: An interplay among advertisers, online publishers, ad exchanges and web users. <u>arXiv preprint</u> arXiv:1206.1754, 2012.
- [176] Meike Zehlike, Francesco Bonchi, Carlos Castillo, Sara Hajian, Mohamed Megahed, and Ricardo Baeza-Yates. Fa\* ir: A fair top-k ranking algorithm. In <u>In Proceedings of the 2017</u> <u>ACM on Conference on Information and Knowledge Management</u>, pages 1569–1578. ACM, 2017.
- [177] Mi Zhang and Neil Hurley. Avoiding monotony: improving the diversity of recommendation lists. In <u>Proceedings of the 2008 ACM conference on Recommender systems</u>, pages 123–130, 2008.
- [178] Xiangyu Zhao, Zhendong Niu, and Wei Chen. Opinion-based collaborative filtering to solve popularity bias in recommender systems. In <u>International Conference on Database and</u> Expert Systems Applications, pages 426–433. Springer, 2013.
- [179] Ziwei Zhu, Xia Hu, and James Caverlee. Fairness-aware tensor-based recommendation. In Proceedings of the 27th ACM International Conference on Information and Knowledge Management, pages 1153–1162, 2018.
- [180] Cai-Nicolas Ziegler, Sean M McNee, Joseph A Konstan, and Georg Lausen. Improving recommendation lists through topic diversification. In <u>Proceedings of the 14th international</u> conference on World Wide Web, pages 22–32, 2005.

[181] Cai-Nicolas Ziegler, Sean M McNee, Joseph A Konstan, and Georg Lausen. Improving recommendation lists through topic diversification. In <u>Proceedings of the 14th international</u> conference on World Wide Web, pages 22–32. ACM, 2005.